## Hydrogen strategy optimization for a sustainable electricity grid

**EPA2942: EPA Master Thesis** Stijn Knoop





# Hydrogen strategy optimization for a sustainable electricity grid

Research on strategy optimization of a sustainable electricity grid including hydrogen storage using particle swarm optimization

by



Student Number

4608046

Composition of the graduation committeeFirst supervisor:J. De StefaniSecond supervisor:Y. HuangChair:E. ChappinExternal supervisor:B. Willemse - CGIExternal supervisor:P. Badaltjawdharie - CGIProject Duration:December, 2022 - June, 2023Faculty:Faculty of Technology, Policy & Management, Delft



## Preface

It is with great pleasure that I present this thesis as the culmination of my Master of Engineering and Policy Analysis program at Delft University of Technology. Over the course of an extensive research period, I have investigated the optimization of an operating strategy for an electricity grid incorporating hydrogen storage. This endeavor has been driven by a desire to delve deeper into sustainable solutions and make a meaningful contribution to the existing body of knowledge in this field.

I would like to express my sincere appreciation to my supervisors for their invaluable guidance throughout the research process. I extend my gratitude to Jacopo De Stefani for his numerous feedback meetings, which have provided crucial guidance in shaping the structure and content of this master thesis. Additionally, I would like to thank the other members of the graduation committee, Emile Chapping and Yilin Huang. Although our interactions were less frequent, I am confident that your feedback has steered this thesis in the right direction.

Furthermore, I would like to acknowledge CGI Nederland for their support in facilitating this research project. During the period of writing this thesis, I have gained valuable insights into their organizational practices, which will undoubtedly benefit me in my future endeavors. I wish to express special thanks to Bas Willemse and Prashant Badaltjawdharie. Their efforts have been instrumental in enabling this research, and I am sincerely grateful for their guidance and insightful reflections.

Lastly, I would like to express my deepest gratitude to my family, with special appreciation to my parents, for their unwavering support throughout my entire academic journey. Their support and encouragement have been instrumental in my accomplishments and have provided me with the strength and motivation to overcome challenges.

Stijn Knoop Delft, June 2023

## Summary

Global warming is one of the biggest challenges in the world today. The cause of global warming is largely attributed to human activities, particularly the burning of fossil fuels, which releases greenhouse gases into the atmosphere. Electricity production has a significant contribution to the emission of greenhouse gases and other pollutants that adversely impact the environment. Renewable energy sources, such as solar and wind power, have the potential to provide sustainable electricity and thereby significantly reduce greenhouse gas emissions and mitigate climate change. Popular sustainable resources such as solar energy or wind energy, are highly dependent on weather conditions, which causes problems regarding availability of electricity. Hydrogen storage is emerging as a promising sustainable solution for reliability challenges that come with weather dependent sustainable resources. However, hydrogen storage systems also face challenges. The efficiency of hydrogen electricity is lower compared to electricity directly sourced from sustainable resources and costs associated with a hydrogen system are high.

This thesis aimed to determine an optimized operating strategy for an electricity grid including a hydrogen storage, considering three key performance indicators: financial costs, reliability, and sustainability. The objective is to deliver electricity in a sustainable manner while ensuring financial efficiency forming the following research question:

#### How can electricity demand be covered by a system of solar power, conventional power supply, and hydrogen systems, optimized for reliability, sustainability, and cost-efficiency, by adjusting the operating strategy?

The study expressed all key performance indicators in terms of costs, with sustainability measured by the price of carbon credits, quantifying CO2 emissions. To obtain the optimal strategy, this research uses discrete event modelling to simulate an electricity system on an hourly basis including solar energy, a hydrogen system, and non-sustainable energy. The particle swarm optimization algorithm, enhanced with linear decay, was employed to optimize operating strategies within this model and identify the most optimal operating strategies. Four scenarios were set up to analyze what the most optimal operating strategy was in different circumstances.

The study found that, in the scenario in which the price of hydrogen is significantly higher than that of conventional resources, the optimal strategy avoids using hydrogen electricity. The main cost drivers are the high costs associated with electrolysis, followed by the costs of fuel cells. In the scenarios in which hydrogen is competitive with conventional resources, the optimal operating strategy was found and only a slight portion of the total electricity produced came from non-sustainable resources.

In conclusion, this thesis addressed the main research question by optimizing the operating strategy of an electricity grid incorporating hydrogen technology. As of today, the sustainability benefits of hydrogen electricity do not outweigh the financial advantages of non-sustainable electricity. This research finds that to achieve a future scenario in which hydrogen electricity competes with natural gas, substantial cost reductions in the components of the hydrogen system are required, along with an increased price of carbon credits. The findings highlighted the challenges and opportunities in implementing hydrogen systems and provided valuable insights for stakeholders in the energy sector. Further research is needed to enhance the understanding and efficiency of operating hydrogen systems and to explore additional factors and components within the electricity system.

## Contents

Preface i								
Sur	nma	iry ii						
1		InImage: constraint of the systemDeduction1Problem introduction11.1.1Problem1.1.2Renewable energy as a solution11.1.2Renewable energy as a solution11.1.3Integration of hydrogen storage as a solution11.1.3Integration of hydrogen storage as a solution1Knowledge gaps2Research goal3Research goal3Research question and sub-questions51.5.1Research question51.5.2Sub-questions5Research relevance61.6.1Scientific relevance61.6.2Societal relevance71.6.3Masters' relevance71.6.4Stakeholder relevance8						
2	1.7 <b>Bac</b> l 2.1 2.2 2.3 2.4	Thesis structure8kground information10Introduction10Transitioning the electricity grid to sustainability10Study of this research11System properties of components in the electricity system132.4.1Solar panels132.4.2Electrolysers152.4.3Fuel cell162.4.4Inverter172.4.5Hydrogen storage172.4.6Gas turbine18						
	Lite 3.1 3.2 3.3 3.4 3.5	rature review19Introduction19Electricity system including hydrogen storage193.2.1Contribution20Optimizing operating strategy213.3.1Contribution22Positioning22Research choices after reviewing the literature223.5.1Scale223.5.2Sustainable resources233.5.3Energy storage23						
•	4.1 4.2 4.3	earch approach24Introduction24Type of research24Methodology24Tools26						

	4.5 4.6	Methods       4.5.1       Simulation       4.5.2         4.5.2       Optimization: Particle swarm optimization       4.6.1         Datasets       4.6.1       Electricity generation         4.6.2       Electricity demand       4.6.3         Modification of datasets       4.6.3       Modification of datasets	26 26 29 30 34 36							
5	A Si	imulation model for finding the optimal operating strategy	<b>37</b> 37							
	5.1 5.2	Conceptual model								
	5.3 5.4	5.2.1 Assumptions and limitations	38 39 41							
	5.5	Optimization through particle swarm optimization	45							
6	<b>Res</b> 6.1	ults Introduction	<b>47</b> 47							
	6.2	Scenario setup	47							
	6.3	6.2.1 Introduction	47 51							
	0.0	6.3.1 Introduction	51							
		<ul><li>6.3.2 Scenario 1: Current state</li></ul>	51 52							
		6.3.4 Scenario 3: Sustainable priority	52							
		6.3.5 Scenario 4: Competitive hydrogen	62							
	6.4	Overview of scenarios and its results	67 <b>69</b>							
7	Discussion         7.1       Introduction									
	7.2	Result interpretation	69 69							
		7.2.2 Question 2: Explanation of expensive hydrogen in scenario 2, Current state								
		7.2.3 Question 3: Transition from scenario 2, Current state, towards Scenario 3 and 4								
		<ul><li>7.2.4 Question 4: Model limitations</li></ul>	71 72							
	7.3	Limitations of the research	73							
	7.4	Validity of the research	73							
8		nclusion	74							
	8.1 8.2	Introduction	74 74							
	0.2	8.2.1 Sub-question 1	74							
		8.2.2 Sub-question 2	75							
		8.2.3 Sub-question 3:	75 75							
		8.2.5 Sub-question 5:	76							
		8.2.6 Main conclusion	76							
	8.3	8.2.7 Sub-question 6:	77 77							
Re	References 78									
	Dataset transformation coding									
в	Model coding									
С		Particle swarm optimization coding								

## Introduction

#### 1.1. Problem introduction

#### 1.1.1. Problem

Global warming is one of the biggest challenges in the world today. The effects of climate change are already being felt, with rising sea levels, increased frequency and intensity of extreme weather events, and threats to biodiversity and ecosystems [19]. The cause of global warming is largely attributed to human activities, particularly the burning of fossil fuels, which releases greenhouse gases into the atmosphere [23]. The urgency of addressing this issue cannot be overstated, and the transition to renewable energy sources is a crucial step toward mitigating the impact of climate change. The Paris Agreement of 2015, which aims to limit global warming to well below 2°C above pre-industrial levels, highlighted the need to transition away from fossil fuels and move towards renewable energy sources [79].

Electricity production has a significant contribution to the emission of greenhouse gases and other pollutants that adversely impact the environment and human health. According to [2], electricity production accounted for 25% of total greenhouse gas emissions in 2021, making it the second-largest source of emissions. This highlights the need for urgent action to mitigate the environmental impact of electricity generation.

#### 1.1.2. Renewable energy as a solution

Renewable energy sources, such as solar and wind power, have the potential to provide sustainable electricity and thereby significantly reduce greenhouse gas emissions and mitigate climate change [104]. However, the adoption of these sources of energy is not without its challenges. One such challenge is the transformation of electricity infrastructures. A well-functioning sustainable infrastructure can not be constructed overnight, and a lot of investment and development is required to create clean, efficient, and reliable infrastructures that supply societies with sustainable energy [112]. Another challenge is the issue of electricity security. Electricity security refers to the reliability and availability of electricity supply, which is essential for modern societies to function [45]. Popular sustainable resources such as solar energy or wind energy, are highly dependent on weather conditions, which causes problems regarding availability of electricity. When a society is completely dependent on sustainable methods of generating energy, coming directly from natural resources such as wind or solar power, it will not be able to fulfill the electricity demand at all times, which will result in moments of power shortages and all the costs that comes with it. Building a sustainable energy infrastructure only consisting of solar and wind energy alone is therefore not sufficient because it does not suffice the electricity demand and therefore does not comply with all the aspects of electricity security [20].

#### 1.1.3. Integration of hydrogen storage as a solution

Hydrogen storage is emerging as a promising solution for reliability challenges that come with weatherdependent sustainable resources such as solar panels and wind turbines. This is due to the long-term storage capabilities of hydrogen storage [4] The integration of renewable energy sources such as wind and solar power into electricity grids has led to an increasing need for energy storage systems to ensure a reliable and stable electricity supply [11], which can be offered by including a hydrogen system.

The use hydrogen, two different chemical operations are required: electrolysis and electrochemical reactions in a fuel cell. Electrolysis is a method that converts water into hydrogen and oxygen. This process splits the water molecules into hydrogen and oxygen molecules using electricity, visualized in Formula 1.1 [95]. The hydrogen is then compressed and stored in tanks or underground reservoirs. When needed, this hydrogen can be used to generate electricity again. This is done in a fuel cell in which the hydrogen reacts with oxygen to create water and electricity, visualized in Formula 1.2.

$$2H_2O \to O_2 + 2H_2 \tag{1.1}$$

$$O_2 + 2H_2 \to 2H_2O \tag{1.2}$$

In times, when there is abundant sustainable electricity generated, the excess electricity can be used for electrolysis, to create hydrogen. In times when sustainable electricity production through solar and wind energy is scarce, hydrogen storage systems can provide a source of electricity, by transforming hydrogen into electricity using fuel cells [95]. Integrating hydrogen storage into an electricity grid can help to reduce the need for conventional non-sustainable power plants to meet electricity demand, improving the overall sustainability, while maintaining the reliability of the grid.

The storage of hydrogen poses several challenges that must be addressed. One of the challenges is the need for an infrastructure that supports the storage and distribution of hydrogen. In addition, another challenge is in regard to operating an electricity grid including a hydrogen system [116]. The intermittent nature of renewable energy sources such as wind and solar adds complexity to operating the storage and distribution of hydrogen. An efficient operating strategy for a hydrogen system is necessary to ensure a constant supply of electricity while being cost-efficient. Furthermore, the cost of hydrogen as a resource is not competitive with non-sustainable resources such as natural gas, which limits its widespread adoption despite its potential [46]. These challenges underscore the need for continued research and development to improve hydrogen storage technologies and overcome the economic and technical barriers associated with its use in the electricity grid.

#### 1.2. Knowledge gaps

Despite ongoing research and development efforts to address the challenges mentioned in the preceding paragraph, there remain significant knowledge gaps in the literature that need to be explored. This section highlights the need for more research in hydrogen systems. It also delves deeper into the knowledge gaps, before explaining which specific gap this research aims to address and contribute towards closing.

While there is a wide variety of studies executed regarding hydrogen storage in a sustainable electricity grid, the literature states that there are a lot of knowledge gaps concerning hydrogen systems, regarding the challenges mentioned in 1.1.3 Integration of hydrogen storage as a solution. [30] state that there is a need for more research on integrating hydrogen storage with different combinations of energy storage systems, such as batteries, hydrogen storage in tanks, or hydrogen storage in salt caverns. Also, many articles that study hydrogen-related electricity systems state that the development of new technologies in hydrogen storage, fuel cells, and electrolysis require further research since these components are to become more developed in the future [59, 75, 97].

An additional aspect that requires more research in the hydrogen field is research on operating a hydrogen system efficiently [118]. The authors in the article [116] underline this lack of research by claiming that there is a significant knowledge gap in the integration of controls, operation, and maintenance requirements of these systems. The authors of [116] state that there is a deficit in the understanding of how to integrate hydrogen technology into an operational environment and be able to maintain power quality and reliability. Including hydrogen technology in a conventional electricity system will result in the electricity supply becoming more sustainable but at the same time less reliable if not managed correctly, which highlights the need for research on operating hydrogen systems.

What is striking is that most of the literature that is concerned with conducting an electricity system including hydrogen focuses on the ratio between the components in the electricity grid, such as the number of electrolyzers, the capacity of hydrogen storage, etc, and not on operating the system [74, 121, 65]. Studies that have been focusing on finding an optimal strategy are very limited and mostly focus on economic cost reduction [115, 88, 10]. The literature that focuses on balancing cost reduction with carbon emission is even more limited, and each research has a different approach regarding what electricity system it analyzes. For instance, [98] researches an electricity system including a hydrogen storage tank, proton exchange membrane fuel cells, and electrolyzers. There are much more variations possible with different types of components which are yet to be explored, as stated before in the second paragraph of this section by [30].

This research aims to contribute by exploring the knowledge gaps on hydrogen systems, stated above by including the following. This study will focus on including a hydrogen system in a conventional electricity grid using natural gas and solar panels to generate electricity. It will specifically try to optimize an operating strategy that will include balancing financial cost efficiency with carbon emissions. Furthermore, it will focus on a unique combination of components required for a hydrogen system, as well as unique data. Lastly, future scenarios, in which the components within a hydrogen system are more efficient will be analyzed.

#### 1.3. Research goal

The goal of this thesis is to find an optimized operating strategy for an electricity grid that makes use of hydrogen technology taking three key performance indicators into account, which makes it a multiobjective operating strategy. An operating strategy is in this research defined as a set of rules which control the power supply and control which resources in the system are producing the power. The first key performance indicator that should always be honored is reliability. This indicator is focused on providing enough electricity to the electricity system so that the electricity demand is always covered. However this is a crucial indicator in electricity systems it will not be used in this research. Reliability should always be met and the consequences of power shortages, as mentioned earlier, are disastrous and very expensive if translated to financial costs. The second key performance indicator in the operating strategy is **sustainability**. Hydrogen systems are placed in the electricity system to enhance the use of sustainable resources as explained in 1.1.3 Integration of hydrogen storage as a solution. There are times when there is low electricity demand and excess renewable energy. Hydrogen systems are in place to generate and store hydrogen from this excess renewable energy through electrolysis. This stored hydrogen can be used as a sustainable resource to provide the electricity demand in times of peak demand or when renewable energy generation is low. The third and last key performance indicator is **financial costs**. Electricity should always be at accessible prices in order for a country to create prosperity [114].

The key performance indicators have different objectives which can be contradictive. For instance, if there is a scenario in which generating electricity in a conventional way, using natural gas, is cheaper than generating electricity in a sustainable manner with hydrogen, what strategy should be chosen? If the key performance indicator "Sustainability" has a higher priority than the indicator financial "Financial costs", one can state that hydrogen is the best option to choose. But how costly can electricity, generated by hydrogen, become before the tipping point is reached, and electricity produced by conventional methods, such as natural gas, is the preferred method? This trade-off between the key performance indicators sustainability and financial costs is influenced by the acceptability of fossil fuels [53]. In Chapter 5 Model, elaboration will be provided on how the key performance indicators are assessed in this research.



Figure 1.1: Key performance indicators for an optimized strategy (own figure)

#### 1.4. Research scope

To not over-complicate this study and maintain it at an achievable level, this research will have some demarcations.

One demarcation in this research is that the supply of energy will be delimited to solar energy and fossil energy. The reason for this demarcation is because solar energy provide over 10% of the global energy in 2021 and is therefore the largest source of sustainable energy [83]. In the future, this generation method will increase in usage across the world [104]. There are also other sustainable resources that have a significant share in producing electricity supply, such as wind energy[14], but to keep this thesis manageable, other resources will be excluded for this research[14].

This research will use weather data from the Netherlands on an hourly basis. Seasonal changes in weather conditions have a high impact on how much solar irradiance is present to generate solar power. Different parts of the world have different weather conditions and different seasons which is why it is important to demarcate a specific climate. Also, historical hourly data from the Netherlands on electricity demand is used in this research. This data, together with the data on solar irradiance in the Netherlands will be the only datasets used in this research.

Energy storage in this research will be limited to hydrogen storage in salt caverns. The Netherlands has the unique circumstance of having several depleted salt caverns that are already being repurposed for hydrogen storage [111]. The Hystock project in Groningen, for example, is exploring the potential of storing hydrogen in salt caverns [41]. Currently, six of them are used to store natural gas which already proves that this technology is feasible. Batteries for short-term power supply are not included because batteries are difficult to scale which makes them less suitable for large-scale storage [90]. Another reason is that batteries carry an environmental footprint due to factors such as raw material extraction, manufacturing processes, and end-of-life disposal, which make them not really a sustainable resource [13]. Fossil fuel will be included in the research to compensate for any shortage of sustainable energy, or to find a more efficient strategy.

Events such as cyberattacks, or unpredictable and unusual events, such as hurricanes, tsunamis,

geopolitical conflicts, etc, which will have effects on the supply and demand of electricity will not be considered in this research.

#### 1.5. Research question and sub-questions

#### 1.5.1. Research question

The goal of this research is to find the most optimal operating strategy for an electricity system including hydrogen storage. Firstly, an operating strategy, existing out of a set of rules, will be constructed based on the literature review. Further, this strategy will be optimized according to the key performance indicators mentioned earlier in 1.3 Research goal. Lastly, attention will be given to the results of this research to obtain insights and will reflect on what components of the electricity system are having the most influence on the outcome of the strategy. The research question of this research is as follows:

How can electricity demand be covered by a system of solar power, conventional power supply, and hydrogen systems, optimized for reliability, sustainability, and cost-efficiency, by adjusting the operating strategy?

#### 1.5.2. Sub-questions

In order to obtain an answer to the research question, it is important to understand the problem on a deeper level. Hence the following sub-questions have been identified:

1. According to the scientific literature, what are the most common approaches to finding an optimized operating strategy of an electric system including hydrogen technology?

The purpose of this question is to obtain more insight into how electricity is functioning. The literature review performed in Chapter 3 Literature review will be dedicated to answering this sub-question. To further break down this sub-question, three more questions are identified:

- (a) What are the different interpretations of an electricity grid according to the literature that studies an electricity system including hydrogen storage?
- (b) What are the different operating strategies stated in the literature and what methods are being used to optimize them?
- (c) What are the properties of the components used in an electricity grid including hydrogen storage, according to the literature?

By splitting the sub-question into three topics, an elaborate overview will be accomplished that analyzes all the relevant topics required for this research. By answering question 1a an attempt will be made to gain an understanding of how the literature describes an electricity system that includes hydrogen storage. Exploring question 1b will delve deeper into what operating strategies are proposed in the literature and what methods are being used to optimize these. Question 1c describes the properties of the components in the electricity grids.

The answers to these questions will provide the knowledge required to perform the analysis for this study.

2. How should the key performance indicators in the operating strategy, referred to in 1.3 Research goal, be prioritized to conceptualize the optimal operating strategy?

This question will elaborate on how the sustainability, reliability, and financial costs of an operating strategy can be interpreted in an operating strategy. The indicators on which the performance of the strategies are being evaluated have different units which makes it difficult to express a final score for the strategies. Chapter 5 Model will be dedicated to answering this sub-question.

3. According to scientific literature, which research approach, methodology, and methods are appropriate for determining an optimized operating strategy for electricity supply in the considered system?

To find an answer to the research question, the literature will be reviewed and examined how similar studies have approached this problem. By finding an answer to this question a well-structured and grounded research approach will be found. This sub-question will be answered while studying the literature, performed in Chapter 3 Literature review, and discussed in Chapter 4 Research approach.

4. In the considered case, how can an optimal, with respect to the key performance indicators defined in sub-question 2, operating strategy be determined?

This research will elaborate further on the multi-objective operating strategy and will delve deeper into what research is conducted to find the optimized operating strategy. This is done by modeling a discrete-time model, which will be optimized by particle swarm optimization. In Chapter 4 Research approach, and Chapter 5 Model this question will be elaborated on.

5. How can the optimization model, found in sub-question 4 be used to optimize realistic future scenarios?

When the previous sub-question has found a solution, scenario analysis can be carried out. As mentioned previously in 1.2 Knowledge gaps, there is much literature stating that components, such as electrolysis, fuel cells, and hydrogen storage are becoming more developed. [59, 75, 97]. By defining scenarios the strategy can adapt to these more developed components in the future and conceptualize possible future circumstances.

6. What are the recommendations that can be formulated to the problem's stakeholders based on the results of sub-question 5?

This sub-question will discuss the outcomes of the scenarios which are set up in the previous sub-question. The scenarios will be run to obtain insights in a changing environment and will reflect what components of the electricity system are having the most influence on the performance of the strategy. *These results will be interpreted and recommendations will be formulated for the stakeholders.* 

#### 1.6. Research relevance

This research is relevant from multiple perspectives, which are outlined below. Firstly, the scientific relevance and the social relevance of the research are discussed. Thereafter, the relevance of the study with respect to the master's program, Engineering and Policy Analysis (EPA), is described. Finally, the relevance for the stakeholders involved is discussed.

#### 1.6.1. Scientific relevance

As discussed in Section 1.2 on knowledge gaps, this research seeks to address the integration of hydrogen technology into an operational environment by optimizing an operating strategy. It contributes to the existing literature by examining an optimal strategy for a hybrid electricity system that includes solar energy, hydrogen systems with hydrogen storage, and gas energy, which distinguishes it from previous studies. Unlike studies that solely focus on cost reduction as a one-dimensional objective, this research takes into account environmental factors as well. Section 1.3 on the research goal explains the three key performance indicators that define this objective, which sets it apart from most other research. By considering the additional dimension of "Sustainability," this research encompasses a multidimensional research objective, thereby making a valuable contribution to the scientific literature, which is scarce on this topic.

Furthermore, this research aims to contribute to the literature by utilizing different datasets compared to existing studies. As mentioned in Section 1.4 on research scope, this research employs Dutch load data and Dutch climate data on an hourly basis. Conducting research in a specific climate context, this research adds to the scientific literature as different regions experience distinct weather conditions and

seasonal variations. This approach enhances the relevance and applicability of the optimized operating strategy.

Throughout Chapter 3 Literature review, the scientific relevance of this research will be further reinforced, establishing its position in relation to existing literature and emphasizing its valuable contributions.

#### 1.6.2. Societal relevance

This research holds societal relevance by addressing two interconnected topics. The first topic focuses on the significance of this study for electricity grids, followed by a broader societal relevance concerning climate change.

Currently, in many parts of the Netherlands, the electricity grid is overloaded as of 2023. This can be attributed to two main factors. Firstly, the rapid installation of solar panels across the country in recent years has resulted in excess electricity production in certain areas, exceeding the capacity of the electricity grid to accommodate it [69, 76]. Consequently, sustainable energy sources are occasionally disconnected from the grid, hindering the energy transition process [69]. Secondly, the energy sector has primarily focused on efficiency and cost reduction, leading to insufficient long-term investments in supporting the energy transition [84]. Given the need for investments in the electricity grid and the surplus of sustainable electricity production during certain periods, the storage of excess electricity in the form of hydrogen presents a potential solution to alleviate the challenges faced by the Dutch electricity grid. Therefore, a hydrogen-based solution aligns with the current societal situation and challenges.

Furthermore, this thesis addresses the broader societal relevance of combating climate change by promoting a shift toward sustainable electricity production. Countries are urged to adopt low-carbon energy sources instead of relying on fossil fuels. However, these new energy resources, such as wind and solar power, are highly variable due to their dependence on natural conditions [27]. Consequently, ensuring electricity availability from sustainable resources becomes more complex compared to a grid powered by fossil fuels. Conducting research on a more sustainable system that addresses these challenges can contribute to a better future for society as a whole, highlighting its societal relevance.

The United Nations has established Sustainable Development Goals (SDGs) as a blueprint for a more prosperous future [110]. Access to a sustainable, clean, and well-organized energy grid is interconnected with almost all SDGs, as clean electricity plays a crucial role in fostering prosperity. Specifically, the main overlap lies with the following SDGs:

- UN SDG 7: Ensure access to affordable, reliable, sustainable, and modern energy for all
- UN SDG 12: Ensure sustainable consumption and production patterns
- UN SDG 13: Take urgent action to combat climate change and its impacts

In conclusion, the clear connection between the SDGs and the focus of this study underscores the societal relevance of researching an optimal strategy for electricity grids.

#### 1.6.3. Masters' relevance

The relevance of this Master's thesis will be discussed based on three requirements outlined for the Master's thesis Engineering and Policy Analysis (EPA) [22]. The first requirement states that the subject should be related to Grand Challenges, inform decision-makers, and be relevant in the public or policy domain, or at the interface between public and private domains. The second requirement is that the work should demonstrate a systems perspective and a multi-actor perspective. Lastly, the thesis must utilize EPA methods and techniques for problem analysis and exploration systematically, and incorporate (conceptual) modeling and/or simulation techniques. The following paragraphs clarify how this study meets these three requirements.

Regarding the first requirement, the EPA master's program aims to address "Grand Challenges." This research, which analyzes the transition towards more sustainable electricity while maintaining accessibility and affordability, fulfills this requirement, as the energy transition is considered a Grand Challenge

[78]. The study provides a deeper understanding of integrating sustainable energy systems into conventional electricity systems and can serve as a guideline for policymakers. Unlike most literature that focuses solely on cost reduction, this research considers environmental factors, offering more comprehensive insights into energy provision and supporting discussions on effective environmental policies.

Furthermore, the second requirement is met as this thesis adopts both systems and a multi-actor perspective. It explores the integration of a conventional electricity system with a sustainable electricity system, including hydrogen storage. Additionally, the research's goal is multidimensional, encompassing both cost reduction and environmental factors, indicating a transdisciplinary objective. This demonstrates an awareness of the broader context. Transitioning towards a sustainable society inherently involves multiple actors, including the realms of the market, government, science and technology, and civil society, as described by Grin [38]. Similarly, the adaptation of hydrogen systems into electricity grids impacts electricity markets, society's energy resources, technological applications, and electricity provision policy.

Lastly, the third requirement is fulfilled as the thesis systematically employs EPA methods and techniques for problem analysis and exploration. Furthermore, the research utilizes modeling, simulations, and scenario analysis, which will be elaborated upon in subsequent chapters.

#### 1.6.4. Stakeholder relevance

It is important to note that an internship was conducted at CGI consultancy for this study. However, while the thesis provides valuable research for CGI, the company itself will not be considered a stakeholder in this context. Instead, the focus will be on the organizations to which CGI provides advice on electricity systems. The following paragraphs elaborate on the stakeholders involved and how the key performance indicators mentioned in section 1.3 Research goal are linked to these stakeholders.

The first stakeholder to consider is energy companies. These companies have set goals to transition towards more sustainable energy supplies and are actively researching new technologies to achieve their objectives [94, 18]. The visualization presented by [42] shows a consistent increase in globally reported corporate energy research and development (RD) spending on renewable energy, electricity generation, and networks over the past decade. Given the growing need for research on renewables and electricity networks/generation, this study on finding an optimized operating strategy for an electricity system including hydrogen storage contributes directly to the interests of energy companies. The vision of energy companies aligns with all the key performance indicators considered in this study. They have an obligation to provide society with a reliable electricity supply, falling under the category of 'reliability'. Additionally, energy companies have a social responsibility to produce cost-efficient electricity, and addressing the demand for more sustainable energy sources has become increasingly important, to which they must respond.

The second stakeholder is the Dutch government, along with the Dutch society as a whole. A stable electricity grid is crucial for the prosperity of a country [36]. Consequently, the Netherlands benefits from maintaining a reliable electricity infrastructure. As mentioned in section 1.6.2 on societal relevance, investments need to be made to adapt the electricity infrastructure to accommodate the increasing capacity in the grid. Research on hydrogen storage, which could be part of a sustainable solution, has gained significant importance in society and can help release pressure on the overloaded electricity grid. Therefore, this research is relevant to the Dutch government and society, as it addresses both reliability and sustainability indicators that are important to these stakeholders.

#### 1.7. Thesis structure

The research consists of eight chapters in total. Chapter 1 serves as an introduction to the problem and outlines the scope of the study. It provides an overview of the research objectives and sets the context for the subsequent chapters. Chapter 2 on the Theoretical Framework aims to clarify the terminology used in this research and provide a deeper understanding of the concepts and theories relevant to the study. It is not intended to substantiate research choices or discuss the literature extensively. Its primary goal is to provide additional context and background knowledge before delving into the main

research. Chapter 3 focuses on the literature review, where previously conducted research on the topic will be analyzed. This chapter examines the existing body of knowledge to define an electricity grid, position this research within the relevant literature, and identify key insights for constructing the key performance indicators and developing an optimized strategy. Chapter 4 discusses the research approach, methodology, and methods employed in this study. It also introduces the data sources used for analysis and provides an explanation of their relevance. This chapter details the specific steps taken in conducting the research and ensures transparency in the methodology. Chapter 5 presents the conceptual model used in this research. It explains how the modeling objective function and the strategy was developed. Besides it describes the inputs and steps within the model and outlines the model's assumptions and limitations. Chapter 6 focuses on the presentation and interpretation of the results obtained from the model. It begins by clarifying the scenarios that were run in the model, followed by a detailed explanation and analysis of the results. Chapter 7 engages in a comprehensive discussion of the implications of the research findings. Lastly, Chapter 8 draws conclusions by addressing the main research question and sub-questions. It summarizes the key findings, discusses their implications, and offers a concise and coherent conclusion to the study. Also it provides recommendations for future research.

# $\sum$

## Background information

#### 2.1. Introduction

This chapter aims to provide an overview of the research topic, clarify the terminology used in this study, and enhance the understanding of hydrogen systems. Its purpose is to establish a foundation for comprehending the subsequent chapters of this thesis. Importantly, this chapter does not aim to substantiate the research through an extensive literature review or justify research choices, as these aspects will be addressed in the following chapters. The primary objective is to offer contextual information on the subject matter.

The chapter begins with an introduction to electricity supply and how sustainable resources function within the electricity system. It highlights the challenges associated with electricity supply, with a particular emphasis on the potential role of hydrogen systems in addressing these challenges. Furthermore, this chapter provides a detailed explanation of the research problem, giving additional insights into the specific electricity system being examined in this research. Lastly, the components of the electricity system used in this research will be elaborated on by analyzing their properties.

#### 2.2. Transitioning the electricity grid to sustainability

When examining the electricity system from a simplified perspective, it consists of two main components: electricity generators and energy consumers. Traditionally, the grid heavily relied on nonsustainable energy sources like natural gas and coal. These sources offered the advantage of resource storage, allowing flexibility in meeting fluctuating electricity demand. Managing the matching of electricity production with demand was relatively straightforward as grid operators had control over the utilization of these non-sustainable resources.

However, renewable energy operates differently. As mentioned in Chapter 1, renewable energy sources are highly dependent on external factors such as daily weather conditions. For instance, during favorable conditions with abundant sunshine and wind, solar panels and wind turbines can produce high amounts of energy, leading to an excess of electricity that surpasses demand. Conversely, unfavorable weather conditions like cloudy and windless days, coupled with high electricity demand due to factors like low outdoor temperatures, may result in a situation where renewable energy is insufficient to meet the electricity demand. This stands in contrast to the convenience offered by non-sustainable resources.

The ideal scenario is to provide sustainable electricity to the grid in both surplus and deficit scenarios, regardless of weather conditions. This is where hydrogen storage comes into play. As previously mentioned in Chapter 1 Introduction, electrolysis can be used to split water into oxygen and hydrogen particles using electricity. The generated hydrogen can then be stored and utilized at a later time. Essentially, the fuel hydrogen is created that produces no pollutants when used. When the electricity used for electrolysis, and thus the production of hydrogen, comes from sustainable resources, hydrogen generation becomes entirely sustainable. By integrating this system into the electricity grid, it can serve as a solution to the limitations of sustainable electricity generation dependent on weather conditions. In situations where weather conditions are unfavorable and sustainable energy generators cannot produce sufficient electricity, stored hydrogen can be utilized to compensate for the deficit in electricity supply and ensure the electricity demand is met.

However, hydrogen storage systems also face challenges related to efficiency and financial costs. The efficiency of hydrogen electricity is lower compared to electricity directly sourced from sustainable resources. This efficiency loss occurs in the three sequential steps involved: electrolysis, storage, and the utilization of hydrogen to generate electricity. Electrolysis incurs an efficiency loss, as does the storage of hydrogen, and generating electricity from hydrogen also incurs energy losses. Additionally, costs play a significant role. Establishing and maintaining a hydrogen system incurs expenses that are reflected in the price of electricity generated from such a system. If the costs associated with a hydrogen system are high, it may result in significantly higher electricity prices compared to other resources used for electricity generation. Consequently, utilizing electricity from sustainable resources proves to be more efficient.

Despite these challenges, a hydrogen system still offers a solution to the limitations of sustainable resources dependent on wind and solar irradiance. However, it is important to investigate ways to mitigate these challenges and enhance the functionality of hydrogen systems.

#### 2.3. Study of this research

Having established the significance of a hydrogen system in an electricity grid with sustainable resources, this section will now discuss the focus of this research. Subsequently, the components within the electricity grid used in this study will be elucidated. Finally, an explanation will be provided on how the research goal will be approached based on the outlined electricity grid.

This research examines an electricity grid incorporating hydrogen storage. The objective is to deliver electricity in a sustainable manner while ensuring financial efficiency. To achieve this, the research focuses on determining an optimal operating strategy, which involves deciding when to utilize different energy resources such as solar panels, hydrogen, or non-sustainable resources like natural gas. To facilitate the development of an effective operating strategy, a simulation model has been created to replicate an electricity grid. The conceptualization of this model is visualized in Figure 2.1, and the components within this system are listed in the following bullet points. The subsequent paragraphs will provide further clarification on the outlined electricity system and the concept of the operating strategy.

- Energy source (Electricity source + inverter)
- Electrolysis
- Hydrogen storage
- Fuel cells
- Inverter (fuel cell)
- · Electricity demand
- Fossil fuel
- Gas turbine



Figure 2.1: Electricity system including hydrogen storage (own figure)

The electricity system used in this research, as depicted in Figure 2.1, comprises three methods of providing electricity to meet the demand. The first method is electricity generated from solar panels, representing the sustainable source considered in this study. The figure represents a specific number of hectares of solar panels defined for analysis. Prior to the electricity flowing from the solar panels to the demand, an intermediate step is shown in the figure: the inverter. This device converts the electricity into a usable form for meeting the demand.

In addition to the sustainable source, the figure also includes a non-sustainable source depicted as Fossil fuel. The non-sustainable resource used in this research is natural gas, which can also be used to generate electricity. Although this energy source emits emissions, it provides a reliable form of electricity and is therefore included in the research to ensure the reliability of the electricity system.

Lastly, there is the hydrogen system. As seen in Figure 2.1, the only input to the hydrogen system is electricity from the electricity source. This means that the hydrogen system in this research is solely powered by sustainable electricity. The hydrogen system consists of four components. As mentioned earlier, electrolysis splits water into hydrogen and oxygen using sustainably produced electricity. The hydrogen can then be stored in the hydrogen storage until needed. When required, this stored hydrogen can be used to generate electricity through a fuel cell. It is worth noting that continuous on-off cycling of the fuel cell can cause rapid deterioration, which is also a consideration for the smart operating system. Finally, the electricity needs to be converted into a usable form for meeting the demand, which is achieved through the inverter.

With the system outlined, the focus now shifts to the operating strategy. As previously mentioned, the objective of this research is to deliver electricity in the most sustainable and financially efficient way possible. Based on the depicted electricity system, the research aims to optimize an operating strategy that can best achieve this objective. This strategy will determine which energy source (hydrogen, solar, or natural gas), or combination thereof, is used to supply electricity to the grid. Challenges associated with this include the costs of electricity production from each resource and the associated emissions. Additionally, the possibility of depleting the hydrogen storage, rendering hydrogen electricity unavailable, will be investigated. This research aims to identify the best operating strategy for different

scenarios.

#### 2.4. System properties of components in the electricity system

So far, this chapter has elaborated on how a electricity system consists of and what challenges it posses, the potential role of the implementation of hydrogen storage into this electricity system, and what this research will study. This section will discuss what the literature tells about the properties of photovoltaic panels (PVs or solar panels), electrolyzers, fuel cells, hydrogen storage, and inverters, which are referred to in figure 2.1. The properties of the various components will be used in the model, which will be elaborated on in chapter 5 Model. The choices on why these components are chosen for this research will be elaborated on in 3.5 Research choices after reviewing the literature.

#### 2.4.1. Solar panels

#### Hectares

For the modeling part of this research, it is important to have an indication of how many hectares of solar panels are present in the Netherlands which will be explored in this section. As will be elaborated on in 3.5.1 Scale, this research scope will be the electricity coverage of the Netherlands which is why the electricity produced by solar power in the Netherlands will be considered. Solar power will be divided into two groups: solar power retrieved from solar parks and solar power retrieved from installations on roofs.

The demography of solar parks in the Netherlands is visualized in Figure 4.5. According to [70] 3221 hectares of Dutch ground was covered with solar panels at the beginning of 2022. The data obtained by [16] show that the installation of new solar panels is strongly increasing each year which is why can be assumed that there are even more hectares covered as the number stated in Figure 4.5. However, this data is a well-substantiated point of reference as input for the model, discussed in chapter 5 Model.

Besides solar parks, there is also solar power retrieved from solar panel installation on roofs. Data on how much roof is covered in the Netherlands is not available but an estimation can be made. [80] states that the Netherlands has 60,000 hectares of roof surface and 40,000 hectares of roof surface available for solar panels. According to [68] 20% of the Dutch houses have installed solar panels. Under the assumption that the houses that have installed solar panels have covered all of their available roofs with solar panels, 8000 hectares of solar panels are installed on roofs.



Figure 2.2: Solar parks in the Netherlands ([70])

#### **Technical details**

This section will discuss what type of solar panel is most common, what the efficiencies are, and what costs are involved. According to [48] it is likely that the lifetime, efficiency, stability, and manufacturability will increase in the future and the cost will decrease over the years. To not overcomplicate this study, this section will assign static parameters and will not include the technological advancements over the years in the study.

The most used type of solar panels is panels with crystalline silicon cells [21] which are also used in this research. The performance of solar panel modules depends on the temperature and the solar irradiance, as well as on the spectrum of the sunlight [35]. In the literature, there is not a general consensus on the efficiency of these solar panels. For instance, [119] claims that the efficiency is around 15-21 % taking dust and damages into account while [105] argues that the efficiency is already up to 28%. For this research, an efficiency of around 20% will be asserted because this is an efficiency that is compliant with most literature [17, 72, 37]. However, it will be interesting to adjust the efficiency of the solar panels in the modeling part of this research.

For asserting the total costs of solar panels, the purchase and installation costs, maintenance costs, and lifespan are considered. Realistically, the costs of solar parks are different from solar panels on roofs. For this research, the cost properties of solar parks will be explored and will be used for all hectares of solar panels in the model. The reason for this scoping is for the sake of simplicity and for policy reasons. However, one should keep in mind that solar panels on houses are owned and paid for by the house owner. Therefore the total costs assumed in this thesis might vary from other literature. The investment costs (purchase and installation costs) are €650.000 per hectare [99]. These costs include the cost of solar panels, inverters, foundation, construction, wiring, installation, and security. Yearly costs stated in [99] are tenancy, security, maintenance, and unforeseen costs which are adding up to €2.000 per hectare per year. Lastly, the lifetime of solar panels will be discussed so the depre-

ciation over time can be calculated. [99] states that the depreciation period is 15 years while other sources are stating that the life expectancy is between 20 to 35 years [106, 50]. This dissimilarity can be explained by the rapid technological advancements of solar panels which will continue in the future [64]. Based on the literature the life expectancy will be set at 25 years for this research. However, for the scenario analysis later on in this research, it will be insightful to change these parameters.

#### 2.4.2. Electrolysers

In this section a clarification will be provided on what type of electrolyzer is used in this research, why this type is used, what the efficiency is, and what costs are implied. As explained earlier in 1.1.3 Integration of hydrogen storage as a solution, electrolyzers are activated by electricity and generate hydrogen by splitting water into hydrogen and oxygen. This process is visualized in Figure 2.3



Figure 2.3: Electrolysis, [107]

There are several types of performing electrolysis. The literature doesn't elaborate on the details of the electrolyzers used to conduct research, or why certain types of electrolyzers are chosen. Alkaline water electrolysis is the most mature and widely used technology for hydrogen production. [120]. It uses an alkaline electrolyte, such as potassium hydroxide, and operates at relatively low temperatures and pressures. Alkaline water electrolysis can achieve high efficiency and can be scaled up for large-scale hydrogen production [55]. For this research alkaline water electrolyzers will be used in the model due to their maturity, high efficiency, and scalability.

In a typical industrial-scale alkaline water electrolysis system it takes 51.2 kWh to produce 1 kilogram of hydrogen, depending on the operating conditions and efficiency of the system ([47]). According to [8], one alkaline water electrolysis installation can handle up to 6MW of electricity to create hydrogen and oxygen, with an efficiency of around 70%.

To obtain the total cost of an electrolyzer the lifespan, maintenance costs, and investment costs should be determined. As will be presented in the remainder of this paragraph, the data on electrolyzers in the literature is dispersed which is why the following approach is carried out in this research. According to [29] the cost of electrolysis is \$6/kilogram of hydrogen produced. However, the cost of electrolyzers has been decreasing over the years due to technological advancements and economies of scale and may decrease in the future as well([47]). For this research, the cost of \$6/kilogram of hydrogen produced will be used.

As mentioned earlier, the literature is very dispersed on the properties of electrolyzers. According to [55], an alkaline electrolyzer's lifespan is about 90,000 hours. Other sources, such as [47], claim

that electrolyzers for alkaline water electrolysis can have a lifespan of 10-20 years or more, with some systems lasting even longer. According to [66] the cost of electrolyzers for alkaline water electrolysis varies depending on the size and complexity of the system. Large-scale electrolyzers can cost millions of dollars, while smaller systems can be less expensive. Because of this indistinctness, the value presented in [85] is used for this project.

#### 2.4.3. Fuel cell

In this section a clarification will be provided on what type of fuel cell is used in this research, why this type is used, what the efficiency is, and what costs are implied. As explained earlier in 1.1.3 Integration of hydrogen storage as a solution, fuel cells generate electricity by combining hydrogen with oxygen. When these two molecules are combined, water and energy are produced as visualized in Figure 2.4.



Figure 2.4: Hydrogen fuel cell, [67]

#### Choice of fuel cell

There are multiple types of fuel cells for the production of energy from hydrogen. What type of fuel cell is the best fit is very dependent on what scale the study is performed on [3]. [3] use Proton exchange membrane fuel cells (PEMFCs) which typically generate between 1 W and 100 kW of output power. This research focuses on small-scale applications which is why the PEMFCs were chosen. However, the study performed in this thesis is large-scale which requires different properties. [108] has put together an overview of different types of fuel cells which shows that a solid oxide fuel cell (SOFC) is the most suitable fuel cell for the purpose of powering a grid on a large-scale level and thus for this research. [63] claim that the solid oxide fuel cell is the most efficient power-to-power technology but also the most expensive.

#### **Technical details**

As stated in the overview presented in [108] the solid oxide fuel cell has a production range of 1kW up to 2MW, needs a long start-up time of about 3 hours [28], and has a limited number of shutdowns. The reason for the limited number of shutdowns is because of the high temperatures required for this type of fuel cell, and too much thermal stress can cause degradation and breakdowns [108].

The solid oxide fuel cell has an efficiency of 60%. The specific energy of a hydrogen fuel cell (which is the measure of how much energy can be produced per unit mass of hydrogen fuel) typically falls within the range of 40-60 MJ/kg, depending on the type of fuel cell and its operating conditions [7]. This means that to produce 1 MWh of electricity, it would require about 72 kg of hydrogen per hour, assuming a fuel cell with a specific energy of 50 MJ/kg. This efficiency will be adopted in the modeling of this research.

[52] has performed research on solar panel and Solid oxide fuel cell electricity production and concludes that the levelized cost of electricity, or break-even price is 0.11/kWh. The model will include this specific price to perform model runs on.

#### 2.4.4. Inverter

An inverter is an electronic device that converts direct current (DC) to alternating current (AC) and will be modeled in the model in this research. Inverters are commonly used in renewable energy systems such as solar photovoltaic (PV), wind turbine systems, and fuel cells, where the DC electricity generated by the system needs to be converted to AC electricity to be used in the grid or in buildings [61].

#### **Technical details**

For this research, the system loss due to cable loss and inverters will be set at 14%. This is because of the reason that the dataset of solar irradiance, which will be elaborated on in 4.6, also uses this number and there is no reason to adopt another value. Besides, the literature is indifferent to the efficiency of inverters. According to [50] the efficiency of an inverter depends on several factors, such as the duration of the inverter operation, or the temperature. [12] state that high-performance inverters have an efficiency of 98%. Other literature, such as [62] states that modern inverters designed for large-scale electricity supply applications can be highly efficient, with conversion efficiencies typically ranging from 90% to 95% or higher. The costs of the inverter are already included in the costs of solar parks as stated in 2.4.1. For simplicity's sake, the cost of inverters required for fuel cells will be left out of this research. However, the efficiency loss will be adapted.

#### 2.4.5. Hydrogen storage

In 3.5.3 Energy storage a substantiated perspective is provided on why salt caverns are being used in this research. This section will focus on what type of storage is commonly used in the research, the efficiencies, and the cost of hydrogen storage in salt caverns.

#### Common practice in literature

In the literature studied for this research, the common approach to storing hydrogen is in hydrogen tanks [77]. However, as mentioned earlier in this chapter, the scale of the research matters when determining what components to use in the model [77]. Since the scale of this research is larger than the most common research studied in the literature, underground hydrogen storage in salt caverns will be chosen above storage in hydrogen tanks.

The choice for a large-scale underground hydrogen system implementation is influenced by the unique circumstances in the Netherlands, which make it feasible to construct a nationwide hydrogen infrastructure. For instance, the Hystock project in Groningen investigates the possibility of storing hydrogen in salt caverns ([41]). The project website reports that there are 10 salt caverns in Groningen, each capable of holding at least 6000 tons of gas. At present, six caverns are used for natural gas storage, demonstrating the viability of this technology. The remaining four caverns are reserved for hydrogen storage, but the allocation of salt caverns for natural gas and hydrogen storage can be altered in the future, making it worthwhile to explore an enlarged capacity of hydrogen storage in the model.

#### **Technical details**

The efficiency of the storage is, according to [41] 98%. This value can be interpreted in many ways. However, for this research it will be interpreted as, if accessed, only 98% of the hydrogen stored will be accessible, and 2% is going to waste.

The cost of hydrogen storage in salt caverns depends on several factors, including the cost of constructing and maintaining the storage infrastructure, and the cost of extracting and distributing the hydrogen when it is needed [117]. Another key factor that influences the costs, according to [58], is the electricity price and electrolytic activity. According to an article that studied the costs of hydrogen storage in salt caverns, the costs of storage is always under 5% of the overall cost [58]. Another study claims that the cost of hydrogen storage in salt caverns is approximate €0.13/kg of Hydrogen for daily storage [1]. For this research, the costs related to storage will be determined based on the study performed by [58], which means that the hydrogen costs will be calculated as a percentage of the total costs (lower than 5%).

#### 2.4.6. Gas turbine

In this section, the choice of a gas turbine as a non-sustainable resource will be explained, and general information on how much gas the Netherlands consumes as well as the technical details will be discussed.

#### Natural gas in the Netherlands

One of the reasons for choosing gas turbines is the significant contribution of natural gas to total energy production. According to the most recent available data, in the second quarter of 2022, 11 billion kWh of electricity was generated out of a total of 28 billion kWh in the Netherlands [15]. As argued by [14], in 2022, the Netherlands consumed 66.52 billion kWh of electricity, out of which approximately 56.2% was generated from fossil fuels, primarily natural gas.

Natural gas power plants are more efficient than other fossil fuel power plants, such as coal and oil [91]. This means that they can generate more electricity with the same amount of fuel, resulting in lower operating costs and fewer greenhouse gas emissions [25]. Besides, natural gas power plants are highly reliable and can be turned on and off quickly to meet changes in electricity demand. This makes them ideal for providing backup power during times of high demand or in the event of power outages ([102]).

In this research, the gas turbines will only be used when the sustainable energy resources and hydrogen fuel cells do not produce enough electricity for the grid. Therefore, it is valuable to have gas turbines as a backup because it is able to start up fast when there is a peak in electricity demand.

#### **Technical details**

In this section, it is valuable to obtain the efficiency of a gas turbine, the costs of energy produced by this resource, and the fuel used per kWh.

The literature assigns different efficiencies on power production via gas turbines. For this research, only one type of gas turbine will be considered, for simplicity's sake, which will be the Rolls-Royce Trent 60. This gas turbine is known to be a high-performance gas turbine that is commonly used in peaking power plants. The Trent 60 has a capacity of up to 58 MW and is known for its fast start-up and rapid response times [34]. According to [34] the efficiency of a general gas turbine is about 35.8%. Therefore, this efficiency will be used in this research.

Furthermore, the costs of one MWh are according to [73] 90.9\$/MWh. According to [26] it takes 7.36 cubic feet of natural gas to produce one kWh and [33] states that 0.0550 kg CO2 is produced per cubic foot of natural gas. This data provides enough leads to model the emissions and prices of natural gas to find a value for the key performance indicators 'Cost efficiency' and 'Sustainability'.

## Literature review

#### 3.1. Introduction

This chapter delves deeper into the existing literature on optimizing the operating strategy of an electricity system using hydrogen storage. As mentioned in Chapter 1 Introduction, significant research has already been conducted on this topic, although with slightly different approaches. It is crucial to analyze how these studies were conducted and how they addressed the problem, as it provides a foundation for further understanding of the subject and contributes to this research.

The aim of this chapter is to answer sub-question 1: "How can an electricity system including hydrogen technology be defined?". As presented in Chapter 1 Introduction, this sub-question is divided into three specific questions (a), (b), and (c) to obtain a comprehensive answer. Question (a) will be investigated in the section 3.2 Electricity system including hydrogen storage, followed by the answer to question (b) in the same section. The focus will then shift to discussing the position of this thesis in the existing literature and its contribution to it in section 3.4 Positioning. Additionally, in section 3.5 Research choices, the research choices made in this thesis will be elaborated upon, which are based on the findings of the literature studies in this chapter so far.

The method used in this chapter involves gathering academic literature using search engines such as Google Scholar and Scopus. Keywords such as "electricity grid," "hydrogen storage," "strategy optimization," "hybrid electricity systems," and "sustainable electricity grid" will be used to retrieve relevant literature. The selected literature and their citations will provide a comprehensive overview of how electricity grids and the optimization of operating systems are defined in the literature. In addition to academic sources, grey literature will also be collected through Google searches using terms such as 'hydrogen initiatives' combined with keywords like 'electricity grid,' 'renewable energy,' and 'energy supply.' The results and their citations will be reviewed to select relevant literature that provides a comprehensive overview of hydrogen systems, the state of affairs of electricity in the Netherlands, and the combination of both.

#### 3.2. Electricity system including hydrogen storage

To gain a comprehensive understanding of an electricity system that includes hydrogen storage, the literature search in this section focuses on the design of electricity grids in previous studies. The criteria that are looked for are the following: the literature should examine electricity demand and sustainable electricity supply. Furthermore, it should include a hydrogen system, and studies that investigate both sustainable and non-sustainable electricity will be sought. Sustainable electricity is generated from sources that can meet present energy requirements without compromising the needs of future generations, while non-sustainable electricity provides insights into the relationship between "cost efficiency" and "sustainability," as described in section 1.3 on Research Goal. The composition of electricity systems from various studies is presented in the table below.

The literature in the table below, dedicated to obtain a comprehensive understanding of an electricity system that includes hydrogen storage, will be limited to the sustainable resources solar panels, and wind turbines, for simplicity's sake and because these resources are the most used sustainable resources in existing research. Non-sustainable resources in this literature search are defined as any resources that are coming from fossil fuels. Furthermore, the storage component in the literature is split into hydrogen storage and batteries. Again, these two storage methods were the most common practices to store energy which is why these two methods are chosen to present in this table. All methods to store hydrogen are considered in this table as hydrogen storage, as well as all types of batteries, are considered as batteries

Article	Resource		Storage		Scale	
	PV	Wind	Non-sustainable	Hydrogen	Battery	
[57]	X			Х	Х	Microgrid
[88]	X		Х		X	Small scale - Medium scale
[00]						Corsica - Ajaccio
[121]		Х		Х		Eastern Iran - Remote area
[118]	Х	Х		Х		China - Remote industrial park
[51]	X	Х	Х	Х	Х	Canada - Newfoundland
[54]	Х	Х		Х	Х	-
[82]	Х	Х			Х	-
[24]	Х		Х	Х	Х	-
[87]	X	Х		Х		Microgrid
This study	x		X	X		Large scale The Netherlands

Table 3.1: Composition of electricity grids in the li	iterature
---	-----------

Judging from the literature analyzed in the table above the following can be stated. Due to the highly complex nature of electricity systems, simplified representations of reality are commonly used in research. This is reflected in the literature presented in Table 3.1, where models typically consist of a sustainable electricity supply, electricity demand, energy storage systems including hydrogen storage, and an electricity management unit. In an electricity system including hydrogen storage, the studies show that electrolyzers and fuel cells are required to store and generate energy. As explained in 1.1.3 Integration of hydrogen storage as a solution, electrolyzers create hydrogen from electricity and water (see Formula 1.1), and fuel cells generate electricity from hydrogen and oxygen (Formula 1.2). Section 3.5 Research choices after reviewing the literature will delve deeper into the choices made in this research regarding what type of components are being used.

#### 3.2.1. Contribution

As can be observed in the table, the compositions of the electricity systems in previous studies differ from the composition analyzed in this research. As stated before by [30] there is a need for more research on integrating hydrogen storage with different combinations of energy storage systems. Studying the specific composition of the electricity system used in this research does not overlap with the already available literature and will add to the already existing literature. Therefore it will add value to the existing literature and will contribute to the knowledge gap referred to in the introduction.

Additionally, it should be noted that the majority of research on hydrogen storage systems has mainly focused on optimizing the scaling of the different components in the system. This study, on the other hand, will concentrate on the ideal operating strategy for a given ratio between the elements in the system (e.g., a set size of solar parks, a set size of hydrogen storage, etc.) and is therefore not aimed at optimizing the proportions between these components. In 3.3 Optimizing operating strategy an elaboration will be given specifically on literature that aims to optimize the operating strategy.

#### 3.3. Optimizing operating strategy

This section focuses on the various strategies and their optimization methods as described in the literature. During the literature review of this specific section, it was observed that nearly all studies concentrate solely on cost minimization, with very little literature available that incorporates environmental considerations into the operating strategy. The table below presents several research studies found that optimize their electricity operating strategy and the methods they employ. Only studies that focus on optimizing a strategy are included, while research that optimizes the optimal ratio between components within the electricity system, such as fuel cells, electrolyzers, and solar panels, are excluded.

Article	Details article	Optimization method
[88]	<ul> <li>Fuzzy logic controller as operating system</li> <li>24 parameters for Particle swarm optimization</li> <li>Inputs: net power flow, battery state of charge</li> <li>Wind and solar power, battery hydrogen storage</li> <li>Hourly data</li> </ul>	Particle swarm optimization
[115]	<ul> <li>Fuzzy logic controller as operating system</li> <li>Mamdan type of fuzzy controller</li> <li>216 rules for operating system</li> <li>Inputs: solar energy, battery state of charge, electricity load hourly</li> <li>Solar power, battery</li> <li>Hourly data</li> </ul>	Particle swarm optimization
[113]	<ul> <li>Neural net algorithm as operating system</li> <li>Solar power, battery, hydrogen storage</li> <li>Hourly data</li> </ul>	Neural net does not require optimization
[90]	<ul> <li>Markov decision process as operating system</li> <li>Involves electricity market in research and strategy</li> <li>Wind power, hydrogen storage</li> <li>Hourly data</li> </ul>	Backward dynamic programming
[86]	<ul> <li>Distributed energy resource (DER) as operating system</li> <li>Trade-off between economic and environmental performances</li> <li>Solar power, gas</li> <li>Hourly data</li> </ul>	Multi-objective linear programming methodology
[5]	<ul> <li>A rule-based strategy as operating system</li> <li>Wind and solar power, hydrogen storage</li> <li>Solar power not directly to the grid but only used for hydrogen production</li> <li>Wind power directly to the grid but also used for hydrogen production</li> <li>Hourly data</li> </ul>	Grey wolf four objectives optimization technique
[10]	<ul> <li>A rule-based strategy as operating system</li> <li>Rules contain mostly "if" and "then" statements</li> <li>Wind and solar power, battery, diesel generator</li> <li>Hourly data</li> </ul>	Grasshopper optimization algorithm
[24]	<ul> <li>A rule-based strategy as operating system</li> <li>Wind, solar and hydropower, diesel, battery, hydrogen storage</li> <li>Hourly data</li> </ul>	Genetic algorithms
[98]	<ul> <li>The economy and environment as main indicators to evaluate the energy system</li> <li>Optimize two strategies: economic and environmental strategies optimized seperately</li> <li>Solar power, gas, hydrogen storage</li> <li>Hourly data</li> </ul>	Mixed integer linear programming
This study	<ul> <li>A rule-based strategy as operating system</li> <li>Trade-off between economic and environmental performances</li> <li>Solar power, gas, hydrogen storage</li> <li>Hourly data</li> </ul>	Particle swarm optimization

Table 3.2:	Strategies and	optimization	methods in	the literature
10010 0.2.	ouracegies and	opunization	methodo m	the menuture

Based on the literature reviewed in Table 3.2, several observations can be made. Firstly, hourly data is the most commonly used data for optimizing operating strategies. Secondly, the studies differ in their approach, as they utilize different resources. Additionally, it is noteworthy that many studies employ rule-based or fuzzy logic controllers, which are also rule-based, to determine an optimized operating strategy. Finally, it is worth mentioning that artificial intelligence methods are often used in the optimization of these strategies. In fact, of the eight research studies presented in the table above, six have adopted an artificial intelligence approach to optimize their strategies. As noted by the authors of [10], newly-emerged algorithms have been explored to find more suitable optimization techniques, which is why artificial intelligence optimization methods are employed.

#### 3.3.1. Contribution

This research aims to contribute to the existing literature by providing a multi-dimensional approach to optimize an operating strategy. It focuses on the trade-off between cost efficiency, sustainability, and reliability by using key performance indicators. Specifically, it will attempt to blend a sustainable energy supply (in this case, solar energy and hydrogen energy) with a non-sustainable energy supply and determine the best trade-offs in the operating strategy. As mentioned in the first paragraph of this section, the majority of literature focused on optimizing operating strategy is based on cost reduction. This one-dimensional approach only reduces costs and neglects the trade-off explained earlier. Article [86] and article [98] in the above table seems to be the only articles that emphasize both environmental and economic factors when optimizing the operating strategy. However, these studies use a different optimization technique and do formulate their objective function, which will be elaborated on Chapter 5 Model, differently. Therefore, the research in this thesis, which focuses on both environmental and economic factors, includes hydrogen storage and uses an up-to-date optimization method, will add value to the existing literature, and thereby contribute to filling the knowledge gap regarding multi-objective operating strategies.

#### 3.4. Positioning

In an effort to position this research in the literature, this section will decorate the similarities and contradictions in the literature and will justify this research within the literature.

There are many similarities between the literature and the research which are visualized in Table 3.1 and Table 3.2. Table 3.1 shows that there is research that analyses a system with sustainable and/or non-sustainable resources. Also in combination with hydrogen storage has been evaluated. Regarding the operating strategy in Table 3.2, the same can be perceived. Rule-based operating strategies have been optimized using particle swarm optimization and multi-objective strategies have been executed.

However, the combination of the different methods and components in the electricity system is what sets this research apart. To the knowledge of the author, there is yet to be research performed with the same energy resource composition, the same energy storage system, the same research objective, and the same methods used in this research. The research that optimizes their operating strategy with an artificial intelligence method, does not optimize over a multi-dimensional objective function, as this research does. The literature that performs analyses on a multi-dimensional objective, does not use an artificial intelligence approach. Also, the data regarding electricity supply and electricity demand, elaborated on in 4.6 Datasets are unique and have not been studied in the literature so far.

This new combination of the multiple facets of this research is justifiable because of the ample other research that has been written on this topic that has similar research on this topic. This research differentiates itself from, as explained in this chapter, but at the same time is very in line with the existing literature. Therefore it will complement the research on this topic with its findings and will thereby contribute to the academic literature.

#### 3.5. Research choices after reviewing the literature

I intend to start this section with the following: this thesis has not yet provided an explanation for the chosen composition of the electricity grid to be analyzed, the strategy, and the optimization method. After analyzing the literature in the previous sections of this literature review, an attempt will be made to clarify the rationale behind the choices of this study and how it relates to the existing literature. This is required to substantiate the components used in this research elaborated on in section 2.4 System properties of components in the electricity system since this section delves deeper into the properties chosen for this research. Firstly, the scale of the research will be discussed followed by the composition of the electricity grid, the operating system, and the optimization method.

#### 3.5.1. Scale

The scope of this study encompasses the electricity grid of the Netherlands. This choice was motivated by the fact that most research has been focused on microgrids and small to medium-scale grids and not a lot of studies have investigated a nationwide approach to hydrogen storage. Subsequently, datasets

of solar irradiance and electricity demand in the Netherlands will be utilized in this study. These will be discussed in detail in Chapter 4: Research Approach.

#### 3.5.2. Sustainable resources

The sustainable energy source in this research is solar energy. To keep this research manageable other sustainable resources such as wind energy or biomass are excluded. Solar energy refers to the energy derived from the sun's radiation, which is captured and converted into usable electricity using solar panels. This renewable energy source has become increasingly popular in recent years and is considered a good alternative for non-sustainable energy resources such as gas or coil [9]. As the world continues to move towards a more sustainable future, solar energy is likely to play an increasingly important role in meeting our energy needs which is why it is being used in the research [40].

#### 3.5.3. Energy storage

The choice for a large-scale hydrogen system implementation is influenced by the unique circumstances in the Netherlands, which make it feasible to construct a nationwide hydrogen infrastructure. For instance, the Hystock project in Groningen investigates the possibility of storing hydrogen in salt caverns ([41]). The project website reports that there are 10 salt caverns in Groningen, each capable of holding at least 6000 tons of gas. At present, six caverns are used for natural gas storage, demonstrating the viability of this technology. The remaining four caverns are reserved for hydrogen storage, but the allocation of salt caverns for natural gas and hydrogen storage can be altered in the future, making it worthwhile to explore an enlarged capacity of hydrogen storage in the model. The choices regarding the type of electrolyzers, fuel cells, and other components are also based on a large-scale (national) hydrogen system implementation and will be detailed in section 2.4.

Some studies have included battery systems as a short-term storage option. However, this research purposely excludes batteries since their production involves polluting processes, as noted in [10]. Thus, only hydrogen is considered as a storage option in this study.

4

## Research approach

#### 4.1. Introduction

This chapter will provide an elaboration on research question 3: "Which research approach, methodology and methods are appropriate for finding the objective, which is an optimized operating strategy for electricity supply?. By answering this question, this chapter will elaborate on when and how the other research questions designed in 1.5 Research question and sub-questions will be answered.

This chapter is organized into several sections. Initially, the type of research to be conducted will be discussed. This will be followed by an explanation of the chosen methodology and the tools that will be used in the study. The chapter will then clarify the methods used and delve into an extensive elaboration of particle swarm optimization. Subsequently, research biases will be discussed, and strategies for minimizing their effects will be proposed. Finally, the datasets used in the study will be described.

#### 4.2. Type of research

It can be inferred that the thesis is an applied research project rather than a basic research project. This is because the aim of the research is to develop techniques or strategies to solve a practical problem, which is in the field of hydrogen storage. Additionally, the research is explorative in nature, as the field of hydrogen storage is under-researched [116], and the research is focused on exploring the main aspects of the problem. The research also employs a flexible design, as the timescale and location were not set before data collection began. The main focus of the research is on developing an optimized strategy given a certain dataset, rather than establishing generalizable facts or testing hypotheses. Furthermore, the research involves computational modeling and simulation runs, which makes it a quantitative research approach. Besides a quantitative approach, this research also involves a qualitative by reviewing the literature on the subject. The simulations will manipulate the control variables, indicating that this research also involves experimental research.

#### 4.3. Methodology

The methodology utilized in this research project will be based on the simulation modeling design proposed by Law and Kelton in their book "Simulation Modeling and Analysis" [56]. This approach provides a systematic framework for carrying out modeling research projects that involve simulations. The simulation modeling design consists of ten stages, which include problem definition, conceptual model development, data collection, validation, computer modeling, verification through pilot runs, validation again, experimentation, output analysis, and reporting. The validation process is conducted twice during the design to ensure the accuracy and reliability of the model. This approach is particularly relevant to our research as it provides a structured and rigorous methodology for conducting simulation-based research. By following this approach, we can ensure that all necessary steps are taken to develop a comprehensive and robust model that can effectively address our research question. The visualization of the design in Figure 4.1 serves as a helpful guide to the various stages involved in the simulation modeling process. Overall, this methodology will enable us to create a reliable and accurate simulation model that can effectively support our research objectives. The remainder of this section will discuss each step individually and clarify what tools and methods will be used, and what research question will be answered.



Figure 4.1: modeling research design (own figure)

The first step in the simulation modeling design proposed by Law and Kelton is already settled in Chapter 1 Introduction and this chapter, which clarifies how the study will be executed. The next step, data collection, and conceptual model development are partially executed already. In Chapter 3 Literature review qualitative data is gathered by exploring how the literature designs the electricity system including hydrogen storage and how the strategy is defined in common literature. Quantitative data is gathered by retrieving the properties of components in the electricity system discussed in the literature review, and by retrieving the datasets from open data sources. The datasets will be elaborated on in 4.6 Datasets. By examining the literature this step will provide an answer to sub-question 1. The conceptual model will be discussed in Chapter 5 Model. In this chapter sub-question 2, regarding the key performance indicators and how they are influencing the strategy, will be clarified Afterward, a validation is planned to evaluate whether the collection of data and the conceptual modeling are performed appropriately.

The subsequent step in the research process involves the creation of the computer model using the Python programming language, which will be discussed in more detail in the tools section of this chapter. The model's accuracy will be assessed through pilot runs to determine if it aligns with the conceptual model, produces appropriate interactions, and provides logical output data. If any discrepancies are detected, the model will be adjusted during the validation stage, which will revisit previous stages in the research design. In addition to the validation stage, another evaluation is scheduled to track the research's progress.

After creating a simulation model, the research will focus on the experiments (or scenarios) to be run and will create output data. Sub-question three will be answered by defining the settings of the experiments. These settings will be determined by the data retrieved from the extensive literature research and interviews. Once, the experiments are executed, the output data will be analyzed to eventually answer the sub-question four, sub-question five, and the research question.

#### 4.4. Tools

In this research, several tools have been employed to aid in data collection, analysis, and modeling. Google Scholar and Scopus were utilized to conduct a comprehensive literature review, which helped to identify existing knowledge and research gaps in the field of operating strategy optimization of electricity systems including hydrogen storage. Python, a programming language, was used to create the computer model and perform simulations. For ease of use and accessibility, Google Colaboratory was utilized as the coding platform for the simulations. Additionally, namely two Python packages, Pandas and NumPy, were used to manage the data and perform mathematical operations in the model. The use of these tools allowed for efficient and effective data analysis, model creation, and simulation runs.

#### 4.5. Methods

#### 4.5.1. Simulation

Discrete event simulation is a method used in this research, which involves modeling an organizational system as a set of entities evolving over time based on the availability of resources and triggering of events. [100] states the following: "A good DES model can replicate the performance of an existing system very closely and provide a decision-maker insights into how that system might perform if modified, or how a completely new system might perform". This is exactly what is performed in analyzing the subject of this research. This study aims to optimize the operating strategy of an electricity grid which provide insights in the performance of the system, and is therefore useful for decision-making. Scenario simulation is another method used in this research, which involves testing the model's sensitivity to different input parameters and scenarios. This method is used to explore the system behavior under different conditions, such as changes in resource availability or demand, and to identify critical factors that affect system performance. Together, these methods provide a powerful tool for understanding and analyzing complex organizational systems and for evaluating the effectiveness of different strategies and therefore policies.

#### 4.5.2. Optimization: Particle swarm optimization

Particle swarm optimization is a heuristic-based optimization algorithm and a method used in this research to optimize the operating strategies proposed. This section will elaborate on why this method is chosen. It will not yet elaborate on the details of how particle swarm optimization is implemented in this research since model specifications are needed which are not discussed yet. Therefore, an elaborate introduction to particle swarm optimization will be given in the remainder of this section. In Chapter 5 Model the implementation of the objective function and what parameters are being mutated by the optimization algorithm will be discussed.

#### Why Particle swarm optimization?

There are multiple algorithms that can be implemented for finding an optimal solution. The exhaustive search, for instance, searches the solutions for all parameter combinations. One can imagine that this is a very thorough way of finding the best solution but is not very efficient, since all combinations need to be explored. As mentioned earlier artificial intelligence techniques to optimize operating strategies are not uncommon which is also visualized in Table 3.2. To efficiently find the best parameter setup to find the most optimized solution PSO has proven in previous research that it is a qualified method to use. For this research, the algorithm will also implement linear decay which is explained below.

#### What is it and how does it work?

Particle swarm optimization (PSO) is a computational heuristic-based optimization algorithm that mimics the social behavior of bird flocks and fish schools. Kennedy and Eberhart introduced PSO in 1995 [49]. The algorithm consists of a swarm of particles, each representing a candidate solution to an optimization problem. The particles all have a different, random setup of the parameters, which move through the search space by adjusting their parameters, based on their own best-known position, as well as the best-known position of their neighboring particles. The objective of the algorithm is to find the optimal solution to the optimization problem by iteratively updating the position of each particle. A visualization of this principle is given in Figure 4.2 in which a parameter has different outcomes on the objective, based on the different setups of the parameter.



Figure 4.2: PSO - one parameter ([103])

In the same way, multiple parameter dimensions can be added. In Figure 4.4 the search space is defined by the X and Y axis, and the objective is given in the third axis. The particles, visualized as black dots, move over the search space to look for new objective scores. This is also visualized by the black arrows/vectors. If a particle recognizes that a neighboring particle has a better best-known position, the particle's direction will be adjusted in the next iteration. The direction and speed will be determined by its own best-known position, the global best-known position, and stochastic factors. By moving around across the search space the particles may find other best-known positions on the way, which will trigger other particles to move to this position. The logic behind each particle is visualized in the flowchart presented in Figure 4.3. This algorithm is therefore exploring new positions while also converging toward the best-known position in the search space.



Figure 4.3: Flowchart PSO particle ([109])

Random initialization of N = 30 particles with velocity



Figure 4.4: PSO - two parameters ([101])

The elements required to perform PSO to find an optimal solution are:

#### · Variables and a defined objective

The defined objective of this research is to secure electricity stability, limit the use of fossil fuels by using sustainable energy resources, and minimize cost. These objectives are expressed in different units which creates room for interpretation. While constructing the model the output needs to represent all the objectives in an overall score. To get to this score the person running the model should estimate how much costs are acceptable to prevent a certain amount of pollution.

The variables in the model are found in the previous chapter in the description of the elements of the conceptual model. The variables will be related to the objective and define the search space of the problem

#### • (Number of) particles

The number of particles is an important setting of PSO. It determines, among other things, how

many places in the search space are investigated in one iteration. If the number is too low, the optimal solution might not be found and will settle for a local best (see Figure 4.2). On the one hand, it is not efficient if the number of particles is set too high, and for each iteration, a lot of places on the search spaces are discovered.

#### Particle hyperparameters

The particle has three parameters which are its cognitive ratio, its social ratio, and its inertia weight. The cognitive ratio is the place of the particle in the search space. The particle can translate the location and the corresponding score to the rest of the particles. The second parameter is the cognitive ratio which is the ability to receive the best location found by any particle. Lastly, the personal velocity of the particle, or the inertia weight, is considered. This parameter defines the size of the step it takes closer to the global best, found by other particles.

#### Balanced stochastic parameters

The balanced stochastic parameters are implemented to find positions in the search space that are different than the best-known solution. Once there is a best solution in the search space, a neighboring solution might be better, and should also be investigated. The stochastic parameter allows the particles to move around a bit and not keep still in the "perfect spot". It is important that the stochastic parameter are balanced among the variables in the model. All particles should move in all directions with the same randomness.

#### Linear decay

In this research, we utilized a technique called particle swarm optimization, specifically a modified version known as linear decay proposed by [96]. The idea behind it is to create a group of particles that can make the most out of their numbers to explore a wide range of possibilities in the search space. Initially, the particles have a more adventurous nature, thanks to the linear decay approach. They are in the beginning free to roam around and discover various solutions. However, as the simulation progresses and the number of iterations increase, the particles gradually start to pay more attention to the global minimum. They become more influenced by it and converge towards that promising point. This way, a balance between exploration and exploitation is obtained throughout the simulation.

#### Pros and cons

Even though PSO is an efficient way to optimize a model it also has some downsides. One downside is that there are quite some parameters that need to be set up [92]. Among other things, what inertia the particles have, how many particles should be initiated, and when to stop running, all need to be determined. When using an advanced version of PSO parameters also need to be assigned to the weights of the particle's inertia. Another disadvantage of PSO is the high computational cost [92]. For each iteration of the algorithm, all particles need to be run, with the assigned variables. If there are 50 iterations of 50 particles, the model will be run 2500 times in total. Lastly, it requires a high level of programming skills which could be seen as a restricting factor. However, PSO is simple to implement in code. It is also flexible and can be used with other optimization algorithms [92]. Furthermore, the research discussed in the chapter Literature Study has demonstrated that PSO is a very suitable approach to optimize a model on.

In conclusion, PSO is a powerful optimization algorithm that has been successfully applied to various optimization problems in different fields. Its ability to mimic the social behavior of bird flocks and fish schools makes it an attractive approach for solving complex optimization problems, which is why its implementation can be found in this study.

#### 4.6. Datasets

Two open-source datasets will be used as input to generate the outcomes of the simulations. One dataset will be used to define how much sustainable electricity can be generated from sustainable resources. The dataset will contain hourly data regarding solar irradiation. The second dataset will contain hourly data about the electricity demand in the Netherlands. Together these datasets are input for the model to create the most optimal strategy. This section will individually discuss both datasets.

#### 4.6.1. Electricity generation

The data on solar irradiance used in this research is retrieved from the Science Hub of the European Commission and is provided with hourly electricity output. Furthermore, the dataset contains data from the beginning of 2005 until the end of 2020. The dataset already includes the conversion from solar irradiance to electricity. This is different from the studies previously discussed in this thesis. For instance, [60] use hourly solar irradiance and wind speed to find the electricity generated per time step using mathematical formulas. The data used in this research does not require formulas for conversion. The method used in the literature, in which the electricity output is calculated based on wind speed or solar irradiance and finds formulas to determine the electricity output [121, 60] is in my opinion, more prone to errors than directly using the data from the Science Hub of the European Commission, assuming that this entity is a reliable source.

The remainder of this section will elaborate on the choices made while retrieving the data from the EU Science Hub to obtain a deeper understanding of the data used in this research. In addition, visualizations of the data will be shown to check the validity of the data.

#### Setting up the data

The first choice that must be made is the location of the solar park. Since we want to generate electricity for all of the Netherlands multiple locations will need to be occupied. For the model to not become too complex, this research will consider one big solar park. In future studies, this could be split up and multiple smaller solar parks could be used in the model to obtain more detailed data on electricity generation from solar energy. The reference point of the one solar park being used in the model will be the city of "Utrecht". This decision is made based on that this city is considered the middle of the Netherlands and since there are solar parks all over the Netherlands, shown in Figure 4.5, it is a convincing choice to choose Utrecht as the reference point.

The second choice that must be made in the settings of the solar panels (PVs) used in the solar parks. Factors that play a role in the amount of electricity that can be generated are: the mounting type, slope, azimuth, PV technology, installed peak power, and system loss. For this research, the mounting type 'fixed' is chosen, which means that the solar panels do not rotate or move during the day. According to the Science Hub of the European Commission, this is the most common type of PV system which is why it is chosen in this research [35]. The solar panels will be installed with the optimal slope and azimuth so that the electricity generation is as efficient as possible. The slope is the angle of the PV modules from the horizontal plane and the azimuth (or orientation) is the angle of the PV modules relative to the direction due South, according to [35]. On the website of [35] there is the possibility to set the position of the PVs in the most optimal setting. For this research, the slope and azimuth are set at the optimal setting which is a slope of 39 degrees, and an azimuth of -1 degrees.



Figure 4.5: Slope (left) and azimuth (right), ([35])

Installed peak PV power is also a variable in the dataset that can be modified. Installed peak PV power is the power that the PV array can produce under standard test conditions, which are a constant 1000W of solar irradiance per square meter, at a temperature of 25°C. Because this research assumes that
the efficiency of solar panels is only 20% as stated in 2.4, it means that 5m2 is needed to produce 1 kWp.

The last variable that can be modified to set up the database of electricity generation from solar panels is system loss. [35] have included estimated losses such as losses in cables, power inverters, dirt (sometimes snow) on the modules, and so on, to create a dataset that considers real-world inefficiencies. They also include the degradation of the solar panels in a linear manner: "Over the years the modules also tend to lose a bit of their power, so the average yearly output over the lifetime of the system will be a few percent lower than the output in the first years" [35].

One last important note is that [35] states that a standard overall loss of 14% is incorporated. This includes, among other things, the loss of electricity which is created by the inverter, which means that the dataset already includes the inefficiencies of an inverter ([35]). Therefore, the computational model does not require to include an inverter.

#### Visualisation

The data will be delivered in a table consisting of seven columns, which is visualized in Figure 4.6. The column names are as follows: time, P, G(i), H\_sun, T2m, WS10m, Int, which meanings can be found in Table 4.1. For this research, the most important columns in this dataset are P and Time which will be used as input for the model. The other columns indicate variables to determine how much power is being generated from the solar panels, but will not be taken into consideration in the model. In Figure 4.7 the energy production is shown of a random day in the dataset (with 35,000 hectares of solar park). It shows that the electricity generation is mostly during the day, which makes sense because of the absence of sunlight in the night. In Figure 4.8 the electricity production of two years is visualized. The figure shows a pattern that indicates that more electricity is produced during the summer and less during the winter. Needless to say, this is also a very logical phenomenon since the summer has an increased presence of sunlight.

Elevation (m):-5						
Radiation database	PVGIS-SARA	H2				
Slope: 39 deg. (opti	mum)					
Azimuth: 1 deg. (op	timum)					
Nominal power of t	he PV systen	n (c-Si) (kWp)	):1.0			
System losses (%):1	.4.0					
time	▼ P	G(i)	H_sun	T2m	WS10m	Int
20050101:0011	0.0	0.0	0.0	6.4	2.9	0.0
20050101:0111	0.0	0.0	0.0	6.19	3.17	0.0
20050101:0211	0.0	0.0	0.0	6.22	3.24	0.0
20050101:0311	0.0	0.0	0.0	6.14	2.97	0.0
20050101:0411	0.0	0.0	0.0	5.77	2.62	0.0
20050101:0511	0.0	0.0	0.0	5.47	2.76	0.0
20050101:0611	0.0	0.0	0.0	4.6	3.03	0.0
20050101:0711	0.0	0.0	0.0	4.72	3.45	0.0
20050101:0811	0.0	0.0	0.0	4.61	3.66	0.0
20050101:0911	22.64	39.6	7.65	5.3	4.0	0.0
20050101:1011	94.92	123.18	12.06	6.13	4.21	0.0
20050101:1111	30.52	49.94	14.44	6.9	4.97	0.0
20050101:1211	591.4	678.65	14.58	7.32	6.28	0.0
20050101:1311	226.89	268.24	12.47	7.41	5.31	0.0
20050101:1411	6.17	17.25	8.3	7.41	5.79	0.0
20050101:1511	0.0	3.63	2.4	7.27	5.79	0.0

Figure 4.6: Data regarding solar power per hour, ([35])

Column name	Explanation
Time	Time period per hour
Р	PV system power (W)
G(i)	Global irradiance on the inclined plane (plane of the array) (W/m2)
H_sun	Sun height (degree)
T2m	2-m air temperature (degree Celsius)
WS10m	10-m total wind speed (m/s)
Int	1 means solar radiation values are reconstructed

Table 4.1: Strategies and optimization methods in the literature









#### 4.6.2. Electricity demand

This section will elaborate on the dataset regarding electricity demand. It will first address how it is retrieved before explaining the data and presenting figures about the data.

The data regarding electricity demand is retrieved from the European Network of Transmission System Operators for Electricity (ENTSOE) [32]. ENTSOE is a non-profit organization that represents 43 electricity transmission system operators from 36 countries across Europe and its main mission is to ensure the reliable, secure and efficient operation of the European electricity transmission system, as well as to promote the integration of renewable energy sources and the development of a pan-European electricity market [31].

The dataset contains data about multiple countries but can be filtered down in the column 'Country' to only data about the Netherlands. Furthermore, the columns in the dataset contain the year, month, day, coverage, and amount of electricity consumed for each hour in the day, measured in MW. The data ranges from the beginning of 2006 until the end of 2015. In this research the most interesting columns are the columns that indicate the date (Year, month, day), and the columns that present the load values each hour. The country is not important since the data is filtered on the Netherlands already. Also, the column coverage is insignificant because one of the key performance indicators in this research is reliability. The data about coverage provides insight that the Netherlands have 100% coverage of all hours between 2006 and 2015, which will be interpreted as the benchmark for this research. A visualization of the dataset is presented in Figure 4.9. Figure 4.10 and Figure 4.11 show the behavior of electricity demand throughout a day and two years. In the behavior during the day one can state that the demand increases during the day and decreases during the night, which is logical behavior. Additionally, we see that during the winter the electricity demand is higher than during the summer, which is also coherent with a common rational.

Hourly load values 2006-2015

Values in MW



To get the	o get the real value, you need to scale them to 100% for coverage ratio less than 100.																											
Country	🝸 Year \Xi	1ont \Xi	þay ⇒	Cov \Xi	0 =	1 =	2 \Xi	3 =	4 \Xi	5 \Xi	6 \Xi	7 \Xi	8 =	9 =	10 \Xi	11 \Xi	12 \Xi	13 \Xi	14 \Xi	15 \Xi	16 \Xi	17 \Xi	18 \Xi	19 \Xi	20 \Xi	21 \Xi	22 \Xi	23 \Xi
				erag																								
NL	2006	1	1	100	10215	9979	9460	8833	8525	8458	8526	8760	9171	9600	10317	10865	11217	11322	11305	11202	11553	12691	13134	13094	12719	12130	11464	10482
NL	2006	1	2	100	9590	9017	8758	8580	8621	9062	10274	12121	13946	14814	15343	15689	15696	15653	15464	15202	15292	16301	15926	15583	14725	13936	12905	12165
NL	2006	1	3	100	11064	10145	9717	9573	9634	9972	11120	12937	14824	15543	15732	15821	15634	15677	15600	15435	15708	16592	15995	15563	14732	13958	12922	12184
NL	2006	1	4	100	11088	10221	9859	9700	9730	10091	11240	13170	15109	15798	16020	16077	15897	15888	15896	15799	15973	16581	15907	15465	14568	13837	12773	12076
NL	2006	1	5	100	10925	10029	9612	9386	9501	9950	11144	13091	15080	15891	16207	16333	16154	16161	16042	16003	16233	16514	15921	15569	14851	13926	12833	12113
NL	2006	1	6	100	10955	10090	9697	9559	9566	9952	11072	12980	14956	15805	16155	16221	15982	15818	15704	15508	15630	16364	15742	15191	14361	13435	12417	11947
NL	2006	1	7	100	10942	10030	9472	9214	9129	9271	9676	10546	11857	13066	13794	13901	13563	13317	13094	12942	13242	14274	14233	13906	13182	12357	11747	11054
NL	2006	1	8	100	10264	9467	9000	8776	8689	8787	8991	9412	10164	10847	11592	12038	12218	12097	11990	11814	12075	13573	13869	13795	13362	12678	11838	10846
NL	2006	1	9	100	9930	9316	9109	9066	9242	9763	11237	13889	15813	16127	16237	16320	15996	16070	15909	15782	15808	16621	16071	15796	14914	14101	12966	12049
NL	2006	1	10	100	10903	10133	9790	9664	9675	10105	11424	14102	16068	16448	16323	16323	16062	16017	15840	15667	15780	16519	15831	15376	14499	13685	12535	11713
NL	2006	1	11	100	10544	9656	9241	9088	9078	9513	10918	13486	15428	15991	16115	16201	16112	16306	16207	16177	16343	16784	16119	15716	14792	14009	12904	12119

Figure 4.9: Data regarding solar power per hour, [32]







Figure 4.11: Electricity demand of two years, (own figure)

# 4.6.3. Modification of datasets

As discussed in the preceding sections, the two datasets used in this thesis are in different formats. To make the datasets usable, they were modified using the Python programming language and the Pandas package. The modified datasets were transformed and now only contain columns for date, time, and electricity demand/supply. The code used to accomplish this can be found in Appendix. A

# 5

# A Simulation model for finding the optimal operating strategy

# 5.1. Introduction

This chapter aims to introduce the model used in this study. Chapter 3 Literature review, will be consulted to determine the models in this chapter. Firstly, a conceptual model will be presented to describe a delimited system, including the boundaries and assumptions made. Subsequently, a computational model will be developed in Python based on the conceptual model, in which the mathematical relationships between the elements in the system will be applied. Section 5.4 will first explain the model, followed by an explanation of the mathematical relationships within the model, including the objective function. Limitations of the model will then be discussed before moving on to the implementation of Particle Swarm Optimization.

# 5.2. Conceptual model

The conceptual model for this research is based upon the conceptual models of other research and can be considered straightforward due to its clear structure. Other studies that have done similar studies on finding an optimal strategy have defined similar conceptual models with similar components in the system [121, 60, 93]. These components are:

- Energy source (Electricity source + inverter)
- Electrolysis
- Hydrogen storage
- Fuel cells
- · Inverter (fuel cell)
- · Electricity demand
- Fossil fuel
- Gas turbine

This section will first describe the electricity demand, followed by an explanation of the electricity inflows from solar energy and non-sustainable energy sources. Finally, the hydrogen system will be described. The interaction between these components is illustrated in the figure below.

Figure 5.1 depicts the electricity demand, which in this study represents the electricity needs of the Netherlands. As shown, there are three arrows connected to this component, all providing electricity. Two of these arrows generate electricity from resources that come from outside of this model: 'Electricity source' and 'Fossil fuel'. In this model, 'Electricity source' defines the electricity coming from solar panels. The solar electricity dataset in the Netherlands discussed in the previous chapter will simulate the behavior of solar energy and will be used as input for this component. To make this solar energy usable, the electricity must be transformed by an inverter before it is delivered to the electricity demand. When the inflow of solar energy is not sufficient and the operating strategy indicates the need to switch to non-sustainable energy, the inflow of fossil fuel will increase. In this model, 'Fossil fuel' represents natural gas, which will be converted into electricity using gas turbines. The last inflow which will be discussed is the hydrogen system consisting of four components. This hydrogen inflow starts with electricity that is not being used by the electricity demand and is instead used to generate hydrogen through electrolysis. The produced hydrogen is then stored in hydrogen storage. In case solar energy does not provide sufficient electricity. This generated electricity still needs to be converted into usable electricity through an inverter and is then ready to be supplied to the electricity grid.



Figure 5.1: Conceptual model (own figure)

#### 5.2.1. Assumptions and limitations

This model only concerns the electricity supply and leaves out other factors, which will be discussed further in this section. This was done to avoid making the model too complex, given the time frame of this project. Additionally, the research focuses on an operating strategy regarding hydrogen and is not intended to incorporate various factors that are not of significant value to the research goal. The remainder of this section will explain the assumptions and demarcations made in the model.

The assumption is made that there is always exactly enough electricity being produced: the model does not incorporate any predictive value to estimate how much electricity will be demanded in the coming hours. This study does not aim to investigate the prediction of energy demand for the upcoming hours but only focuses on how to optimize a hydrogen strategy. However, such a predictive model could be included in the future for expansion to examine the problem in a more complex and detailed manner.

Furthermore, the price of electricity varies throughout the day. Electricity is traded on electricity markets where providers aim to sell electricity at a high price while consumers aim to buy electricity at a low price. Hydrogen storage can optimize strategies so that energy companies can time when to sell their electricity. Developing a strategy that maximizes profit for energy companies is interesting, but it is not the focus of this research and therefore not included in the model. In this model, the price of electricity is defined as the break-even cost. Profit or complex strategies that take into account electricity market

trends are not considered in the model.

In addition, the costs in the model are evenly distributed across time steps, meaning that each hour carries an equal cost. This is a limitation since investments that span multiple years, which is applicable in this model, have a net present value, and costs have a different value in the future. Additionally, the degradation of solar panels, as determined by [35], is linear, which is not realistic. The final limitation that will be discussed is the efficiency of the fuel cells, electrolyzers, gas turbines, and solar panels. One can imagine that different inputs to these components correspond to different efficiencies. For example, at a certain hydrogen supply, the fuel cell will achieve its maximum efficiency, while it may not at a higher or lower hydrogen supply. However, this model has chosen to keep all efficiencies static and not allow them to vary. This was done to make the project manageable and not overly complex. However, it is of great interest to investigate this further in future studies to improve simulation runs.

Finally, this model only considers two states for the fuel cells: either all fuel cells are on, or all fuel cells are off. If future research considers different efficiencies, it may be possible to switch on or off a certain percentage of fuel cells. However, for the scope of this research, this is too complex as it would also affect the costs and lifespan of the fuel cells. Therefore, in this study, the fuel cells are represented with less complexity.

# 5.3. Strategy

This section will discuss the rules and the variables with reference to the operating strategy. By delving deeper into the strategy, a deeper understanding will be obtained of the reasoning behind the methods to obtain the optimal strategy. The operating strategy considered for this research aims to find the most optimal solution to minimize the objective function. The objective functions contain three key performance indicators that relate with each other as explained before in 1.3 Research goal.

As mentioned in Chapter 3 Literature review this research will aim to find an optimal solution by practicing a rule-based strategy as the operating system. It considers six rules and five flexible variables to execute this study. This configuration of the strategy is based on the literature studied in Chapter 3 Literature review. The rules and variables are designed to create different behaviour of the strategies that will be tested. For further research these could be refined or be made more complex to obtain an advanced strategy.

The variables will be labeled with letters A through E, and their discussion will be presented in the bullet points below. The variables will be optimized with respect to the objective function using particle swarm optimization, which will be discussed later in this chapter. Also, the rules will be presented in the bullet points. The rules applicable are as follows:

• **Rule 1:** If the hydrogen storage is below threshold A, the solar energy does not suffice the electricity demand, and the fuel cells are not turned on, the remaining electricity will be produced by a gas turbine that uses non-sustainable energy

To ensure a steady supply of electricity to the grid, it is imperative that the demand for electricity is met. In the event that solar energy is insufficient to meet the demand, the hydrogen system or gas turbine can be employed. It has been established in the literature, as discussed in section 2.4 System properties of components in the electricity system, that frequent start and stop cycles of the fuel cell are inefficient and may lead to thermal stress, thereby reducing the lifespan of the fuel cell. Therefore, the rule presented in this bullet point takes into account the state of the hydrogen storage to determine a strategy that rejects scenarios in which the fuel cells operate for only a short duration. It is illogical to activate the fuel cells when there is insufficient hydrogen in the hydrogen tank, as they require this resource to generate electricity. Thus, this rule favors the use of non-sustainable resources such as natural gas.

• **Rule 2:** If the hydrogen storage is above threshold B, the solar electricity does not suffice the electricity demand, and the fuel cells are not turned on, the fuel cells are turned on

In accordance with rule 1, there should be a moment when the fuel cells are turned on. Rule two is examined when the solar panels do not supply enough energy to the electricity grid and the hydrogen storage in the model is above threshold B. The rationale behind this rule aligns with Rule 1, namely, to minimize the frequency of starting and stopping the fuel cells in order to prevent a shortened lifespan of the fuel cells. If solar energy and hydrogen energy combined, do not supply enough electricity to fulfill the electricity demand, the gas turbine will be activated to generate the remaining demand. If the activation of fuel cells results in an excess of electricity, the remaining electricity will be used to generate hydrogen again.

• **Rule 3:** Considering the fuel cells are on, the solar energy combined with the hydrogen energy does not suffice the electricity demand, and the hydrogen in the tank is used up during the timestep, the fuel cells will be turned off and the gas turbines will produce the remaining electricity to fulfill the electricity demand

In this scenario, the fuel cells are operating and producing electricity through the use of hydrogen. However, in this particular time step, there is not enough energy available from renewable resources and the hydrogen storage has been depleted. Consequently, the fuel cell cannot continue to function and is turned off. As previously mentioned, this is not a scenario that should occur frequently as a high frequency of starting and stopping the fuel cell is detrimental to its lifespan. To still meet the demand for electricity, the gas turbine is activated and will provide the remaining electricity needed.

• **Rule 4**: Considering the fuel cells are on, the solar energy combined with the hydrogen energy does not suffice the electricity demand, and the hydrogen in the tank is above threshold C, the fuel cells will produce B Watt and the gas turbines will produce the remaining electricity to fulfill the electricity demand

This scenario is a variation of the situation described in rule 3. However, in this scenario, the hydrogen storage is above the threshold C while the hydrogen storage in rule 3 is empty or nearly empty. Since the hydrogen storage is above the threshold and the fuel cells are already operating, the fuel cells will attempt to meet the energy demand by producing B Watt. However, if this is insufficient, the gas turbines will provide additional support and meet the remaining electricity demand.

• **Rule 5**: Considering the fuel cells are on, the solar energy combined with the hydrogen energy does not suffice the electricity demand, and the hydrogen in the tank is below threshold C, the fuel cells will produce B Watt and the gas turbines will produce the remaining electricity to fulfill the electricity demand

This scenario essentially describes the same situation outlined in rule 4 but with a different state of hydrogen storage. In this scenario, the storage is below threshold C, triggering a different strategy for managing the hydrogen supply. The fuel cells will produce D watts of electricity in this situation, and the remaining electricity will be generated by gas turbines. The rationale behind this rule is that it is more advantageous to run the fuel cells less intensively in order to slow down the depletion of hydrogen storage. This approach can help prevent the hydrogen storage from becoming depleted, which would result in the fuel cells being turned off.

• **Rule 6:** Considering the solar panels are providing enough electricity in the given time step and the fuel cells are turned off, the excess electricity will be used to start electrolysis and produce hydrogen, if there is an excess of electricity.

Rule 6 is a straightforward rule. When the solar panels generate sufficient electricity, and the fuel cells are turned off, it is logical that the electricity supplied by solar panels supplies the electricity grid. If there is an excess of electricity, the redundant electricity is utilized to initiate electrolysis, producing hydrogen. This ensures that the surplus electricity is still utilized effectively.

• **Rule 7:** Considering the solar panels are providing enough electricity in the given time step and the fuel cells are turned on, the fuel cells will stay on and produce E Watt. Additionally, the excess electricity will be used to start electrolysis and produce hydrogen, if there is an excess of electricity.

This rule, as well as the final rule, is in line with rule 6, where the solar panels produce enough electricity to serve the electricity demand. However, in the scenario described in this rule, the fuel cells are on. As it is not optimal to turn the fuel cells on and off frequently, as described earlier in this thesis, the fuel cells will not be turned off but will produce electricity at a different intensity, namely E Watt. If there is excess electricity produced, it will be used to initiate electrolysis and produce hydrogen. An important note to make is that there may be situations where hydrogen storage runs out when producing E Watt in this scenario. In that unlikely case, the fuel cells will be turned off and the solar panels will produce all the electricity.

The previous bullet points have provided detailed explanations of the rules that constitute the operating strategy in this research. Table 5.1 summarizes these rules in a structured and concise manner. One notable aspect is that at the beginning of each time step, the gas turbine is always considered to be off. This, however, does not imply that the gas turbine is always off and it can certainly be turned on for multiple iterations in a row. If there are multiple time steps where the fuel cells and solar panels are insufficient to meet the electricity demand, this means that the gas turbines are also turned on for multiple time steps consequently. However, the strategy is aimed at generating as much electricity as possible from sustainable resources, which is why the focus is on using fuel cells and solar energy. Gas turbines are only used as a backup source, which makes them well-suited for their quick start-up time, as described in Section 2.4 System properties of components in the electricity system. Another observation is that solar panels do not differentiate between not producing electricity or not producing enough electricity. The reason for this is that the strategy always wants to supply available solar energy directly to the grid before turning to other resources. This is because the direct supply of solar energy is much more efficient than electricity via hydrogen, which involves multiple conversions that result in energy losses. Finally, it is important to mention that A is always greater than C. Otherwise, scenarios may arise where both Rule 2 and Rule 5 meet the criteria simultaneously, resulting in the production of both B Watt and D Watt. This cannot happen, and therefore, the consideration was made to always prioritize A above C.

Rules	Solar panels	Fuel cell	Hydrogen storage	Gas turbine	Energy demand	Strategy		
Rule 1	Off/Not sufficient	Off	Below A	Off	Not met	Gas turbines: turn on		
						Fuel cell: turn on produce B Watt		
Rule 2	Off/Not sufficient	Off	Above A	Off	Not met	Shortage electricity: turn on gas turbine		
						Excess electricity: produce hydrogen		
Rule 3	Off/Not sufficient	On	Empty	Off	Not met	Fuel cell: turn off		
Rule 5		Oli	Linpty		Notmet	Gas turbine: turn on		
Rule 4	Off/Not sufficient	On	Above C	Off	Not met	Fuel cell : keep on produce B Watt		
Kule 4		Oli	ADOVE C		Notmet	Gas turbine: on produce remainder		
Rule 5	Off/Not sufficient	On	Below C	Off	Not met	Fuel cell : keep on produce D Watt		
Rule 5	On/Not sumclent	On	Delow C		Notmet	Gas turbine: on produce remainder		
Rule 6	Sufficient	Off	-	-	Met	Excess electricity: produce hydrogen		
Rule 7	Bula 7 Sufficient		Sufficient On		On -		Met	Excess electricity: produce hydrogen
Rule /	Sunclefit	Oli	-	-	wet	Fuel cell: Produce E Watt		

Table	5.1:	Rules	for	operating	strategy
Table	0.1.	i tuico	101	operating	Juacyy

# 5.4. Computational model

The aim of this paragraph is to provide an overview of what is to be discussed in this section. Firstly, a brief summary will be given to indicate the progress made prior to constructing the computational model. Subsequently, an explanation will be provided regarding the tools employed for creating the model before delving into how the model interprets the data in the dataset. This section will then address the implementation of the strategy defined in the previous section. Furthermore, attention will be

given to how the key performance indicators, referred to in the main research question, are interpreted in the model.

Sufficient input has been collected to develop a computational model. In this paragraph, a brief summary will be provided of the work done so far and the input for the computational model. Firstly, the problem to be investigated was identified in Chapter 1 Introduction. Then, in Chapter 3 Literature review, the literature was examined to understand how the problem is perceived and approached in the field. This led to a better understanding of how an electricity system with hydrogen is conceptualized, as well as the techniques and methods employed to optimize the operating system. Additionally, research was conducted into the properties of important components in the electricity system. In Chapter 4 Research approach, the methods, tools, and datasets that will be employed in this study were identified. Based on this information, a conceptual model with an associated strategy was established in this chapter. All this information will be used to construct the computational model, which will be described in the following paragraphs.

#### Tools for model

As mentioned earlier in Section 4.4, Python, a programming language, was chosen for modeling purposes. For ease of accessibility and convenience, the coding platform Google Colaboratory was utilized. Moreover, two Python packages, namely Pandas and NumPy, specialized for data analysis and manipulation were employed to manage the data and to perform mathematical operations in the model. The use of these tools facilitated efficient and effective data analysis and model creation. To visualize the outcomes in graphs, the package matplotlib was employed.

#### Data interpretation

As previously mentioned in section 4.6, this research will utilize two datasets: one containing hourly electricity production from solar power, and the other containing hourly electricity demand based on usage in the Netherlands. The computational model will iterate through each hour, using the energy resources within the model (solar panels, hydrogen system, and gas turbine) to fulfill the electricity demand according to the strategy. At every iteration, new data is used as input, and the model must respond accordingly based on the chosen strategy.

The solar energy dataset presents the electricity production of one square meter of solar panels. Therefore, it needs to be converted into the number of hectares of solar panels in the Netherlands, as illustrated in Formula 5.1. The rationale for determining the total surface area of solar panels in the Netherlands can be found in Section 2.4: System properties of components in the electricity system.

The formula 5.1 will determine whether the solar electricity produced at a given time is sufficient to meet the electricity demand. If the outcome of the formula is positive, it indicates that the solar panels are producing sufficient electricity, and the model will determine whether rule 6 or rule 7 should be applied. If the outcome of Formula 5.1 is negative, it means that the solar panels are not providing enough electricity, and the other rules will be examined. Once the appropriate rule is applied, the model will move on to the next iteration, as illustrated in Formula 5.2, until the end of the dataset is reached.

$$Solar_t * hectares - Demand_t \rightarrow Electricity_{shortage/surplus}$$
 (5.1)

$$Solar_{t+1} * hectares - Demand_{t+1} \rightarrow Electricity_{shortage/surplus}$$
 (5.2)

Solar	Dataset on solar irradiance [35]
Demand	Dataset on electricity demand of the Netherlands [32]
t	Specific hour in the dataset being studied
hectares	The number of hectares of solar panels in the Netherlands
Electricity	Outcome, defining if there is a shortage or surplus in solar electricity on timestep x

#### Strategy implementation

This paragraph will discuss how the model is programmed to determine which rule to use for a specific time step in a structured manner, as illustrated in Figure 5.2. The following paragraph will attempt to

explain this visualization. Finally, there will be a brief explanation of how the steps in the visualization are implemented in the code. A detailed explanation can be found in Appendix B.



Figure 5.2: Flowdiagram for computer model (own figure)

The aim of the flow diagram is to assign a rule to each hour of the dataset. Starting at the green "starting point," each new entry with hourly data on solar electricity and electricity demand is processed. The first question asked is whether there is sufficient solar electricity to meet the electricity demand. The formula stated in Formula 5.1 will be used in this step. If the answer is yes, the status of the fuel cells is examined. If the fuel cells are on, Rule six is applied, while Rule seven is applied when the fuel cells are considered off. If there is not enough solar electricity, the next step is to assess the fuel cell status. In this case, if the fuel cells are not on, Rule one is applied if the hydrogen storage is above threshold A, while Rule two is applied if it is below threshold A. If the fuel cells are on, the next step is to check the status of the hydrogen storage. If the hydrogen storage is not below threshold C, Rule four is applied. If the hydrogen storage is below threshold C, the question is asked whether the storage is empty. If the storage is empty or becomes empty after supplying electricity in this step, Rule three is applied. If

there is still sufficient hydrogen in the tank, Rule five is activated.

In order to implement these steps in a structured manner within the code, functions have been utilized. A function is a block of code that is executed when called by the main code. In this strategy, four functions have been identified and are listed below.

- surplus\_off()
- surplus\_on()
- shortage\_off()
- shortage\_on()

The functions are self-explanatory when called in the main code. When solar energy is sufficient to meet electricity demand, functions starting with 'surplus' are utilized. If the fuel cells are already activated at the start of the time step, surplus\_on() is executed, whereas surplus\_off() is executed if the fuel cells are not activated yet. Similarly, functions beginning with 'shortage' are used when solar electricity is unable to meet the demand. When the fuel cells are already activated at the start of the time step, shortage\_on() is executed if the fuel cells are already activated.

For more elaboration, the code used to navigate through the model, and the code in the functions, are provided in appendix B

#### **Objective function**

In this section, the objective function in the computational model will be discussed. Firstly, the objective function will be elaborated upon, before it is divided into the three key performance indicators. These indicators will then be individually addressed, and an attempt will be made to demonstrate how they are incorporated into the code.

The objective function, as described in 1.3 Research goal, consists of the key performance indicators reliability, sustainability, and financial efficiency. One of the challenges mentioned is that these objectives can be contradictory. For example, a scenario may perform very well in terms of cost efficiency but poorly in terms of reliability and sustainability, rendering it an inadequate strategy for supplying the electricity grid. The equation presented in Formula 5.3 takes all these indicators into account by minimizing the sum of the indicators, expressed in Euros. By doing this, all the indicators will be matched and a scenario can be sought acknowledging all key performance indicators. In order to minimize the sum of the indicators, they must be expressed in the same unit, which will be explained in the following paragraphs.

$$Objective \ function(\mathbf{E}) = Sustainability(\mathbf{E}) + Reliability(\mathbf{E}) + Financial \ efficiency(\mathbf{E})$$
(5.3)

The key performance indicator that is the most straightforward to follow is cost efficiency, which encompasses the cost of depreciation of the components used to generate sustainable electricity, as well as the costs of resources such as natural gas. This is represented by Formula 5.4. However, a different approach is used in this model. In this research, the levelized cost of electricity production methods is considered. The levelized cost of electricity represents the revenue that an investor in a production facility would need to obtain to break even, and therefore, it reflects the minimum costs. The values of these levelized costs used during the runs are explained in the section 2.4 System properties of components in the electricity system. The equation used in this study for the cost efficiency key performance indicator is presented in Formula 5.5.

$$Cost \ efficiency(\mathbf{E}) = Cost \ of \ depreciation(\mathbf{E}) + Cost \ of \ resources(\mathbf{E}) + Cost \ of \ ...(\mathbf{E}) +$$
(5.4)

$$Costs \ efficiency(\mathbf{E}) = levelized \ cost \ of \ solar \ electricity(\mathbf{E}) + levelized \ cost \ of \ hydrogen \ electricity(\mathbf{E}) + levelized \ cost \ of \ solar \ electricity(\mathbf{E})$$

$$(5.5)$$

The next key performance indicator to be discussed is sustainability. As seen in the objective function presented in Formula 5.3, sustainability is defined as a variable measured in Euros. This may seem counter-intuitive since sustainability is concerned with how much reduction there is in CO2 emissions or other harmful pollutants. However, nowadays, there is already a measure that couples emissions to financial costs, which are carbon credits. carbon credits are tradeable tokens that give the right to the owner to emit CO2 into the atmosphere. One carbon credit equals one tonne of carbon dioxide or the equivalent amount of a different greenhouse gas [67]. Governments hand these carbon credits out to companies, and some carbon credits are sold on the market. The market is in place so when a polluter has exceeded its credits, it can buy credits from other polluters. Because carbon credit sare tradeable the price will fluctuate over time. As of May 14th, 2023, the price of one carbon credit is €88.93.

In this model, the amount of natural gas used during a simulation run is tracked. A certain amount of natural gas produces a certain amount of CO2, which is already determined in section 2.4 System properties of components in the electricity system. To obtain the costs of sustainability, as input for the objective function, the equation in Formula 5.6 is presented. The outcome will be in Euros to match the other key performance indicators in the objective function.

$$Sustainability(\mathbf{E}) = CO_2(kg) \times Carbon \ credit(\mathbf{E}/1000 \ kg)$$
(5.6)

The last key performance indicator to be conceptualized is reliability. In this thesis, reliability should always be honored, since the consequences of a power blackout are devastating. According to [81] a power blackout will lead to disrupted communications and transportation, closure of business and services, and water contamination among other things. This is such a disruptive event that this model will not even consider a scenario in which the reliability is low, resulting in a power shortage. The importance of electricity coverage is also underlined in the dataset of electricity demand [32]. This dataset states that the electricity coverage in the Netherlands is always 100% throughout the whole dataset.

The equation used to conceptualize reliability is presented in Formula 5.7. However, as mentioned earlier, reliability is so important that it will not be used in the model because the consequences of failure are just too disruptive and the costs are therefore too high.

$$Reliability(\mathbf{E}) = Cost \text{ of failure per hour}(\mathbf{E}/hour) \times Hour(hour)$$
(5.7)

# 5.5. Optimization through particle swarm optimization

This section will discuss how the particle swarm optimization algorithm aims to optimize the operating strategy. This is done by minimizing the objective function presented in Formula 5.3. The upcoming paragraphs will elaborate on what the algorithm will be optimizing before delving deeper into how this is done. For this research, Python was used to code the algorithm. The coding on how the particle swarm optimization algorithm is constructed is presented in appendix C

The particle swarm optimization algorithm used in this study aims to optimize the objective function, which measures the combined costs of the different key performance indicators. It will be considered well-performing when these costs are minimized. As described in section 4.5.2 Optimization: Particle swarm optimization, the algorithm works by changing parameters. In this study, these parameters are as follows:

- A: threshold hydrogen storage
- B: electricity production fuel cell
- C: threshold hydrogen storage
- D: electricity production fuel cell
- · E: electricity production fuel cell

Together, these parameters form the search space to be explored by the algorithm. The algorithm aims to find the optimal combination of parameters that minimizes the objective function.

Lastly, the parameter ranges will be discussed. Parameter A and parameter C indicate the fill level of the hydrogen storage tank, expressed in percentages. Therefore, these values cannot be lower than 0 or higher than 100. Additionally, parameter C cannot exceed parameter A, as discussed in Section 5.3. Parameters B, D, and E are adjustments for the amount of Wattage a fuel cell can produce. These adjustments are linked to the rules discussed in Section 5.3. The range of the chosen fuel cell for this research is between 1 kW and 2 MW, as discussed in Section 2.4.3. Therefore, the parameters will fall within this range.

# Results

# 6.1. Introduction

This chapter will discuss the results obtained by running the model and optimizing the operating strategy discussed in the previous chapter. It will first elaborate on the different scenarios set ups and will elaborate on why these set ups have been chosen to perform research on. Afterward, the results of each scenario will be discussed individually. Lastly, the results will be interpreted by comparing the results with each other. Consequently, the nuances of the model will come forward in this section and will be discussed as well.

# 6.2. Scenario setup

# 6.2.1. Introduction

In this chapter, four different scenarios will be presented. This section will explain the rationale behind the selection of these scenarios and the corresponding parameter set up. The set up will be visualized in tables that display the configuration of the components in the electricity system under analysis. These components include:

- Hydrogen storage
- Fuel cells
- Electrolysers
- Gas turbines
- Solar panels

#### Scenario 1: Current state

The first scenario 'Current state' will focus on the current situation without hydrogen systems implemented in the electricity system. This scenario will provide insight into the model outcomes if hydrogen is not implemented, and provides a baseline measurement of the system, to compare the other scenario setups with. The setup can be found in the table below. On the left side of Table 6.1 the hydrogen system is visualized, which is non-existing in this scenario, and on the right side of the table the electricity system including solar energy and non-sustainable energy is visualized. The properties of the solar panels and gas turbines used in this scenario are elaborated on in 2.4 System properties of components in the electricity system

#### Table 6.1: Setup scenario 1: Current state

#### Hydrogen system

Hydrogen storage	
Hydrogen storage capacity (kg)	0
Initital hydrogen in storage	-
Efficiency hydrogen storage	-
Costs hydrogen storage	-

#### Fuel cells

# fuel cells	-
Start-up time fuel cells	-
Range electricity produced	-
kWh per kg hydrogen	-
depreciation/ costs fuel cell	-
Depreciation fuel cell switch	-

#### Electrolyzers

# electrolyzers	-
Maximum W per electrolyser	-
kWh needed for 1 kg hydrogen	-
Depreciation costs electrolyzers	-

#### Solar panels and gas turbine

Gas turbine								
Cost per MW	\$90.9/MWh							
CO2/cubic foot	0.0550 kg CO2/cubic foot							
Gas used per kWh	7.36 cubic feet/kWh							
Costs carbon credit	\$93,07/1 tonne of CO2							
Solar panels								
Hectares solar park	11000							
Life expectancy	25 years							
Efficiency	20%							
Investment costs	€650000/ha							
Yearly costs	€2000/ha							
Depreciation costs/ha	€3,19/hour - \$3.51/hour							

#### Scenario 2: Hydrogen implementation

The second scenario 'Hydrogen implementation' will be the same as the first scenario but with the implementation of a hydrogen system. The rules created in 5.3 Strategy will be carried out to find the optimized objective function described in Formula 5.3 by finding the most efficient setup of the parameters A until E. Just like the other tables described in the other scenarios, on the left side of Table 6.2 the hydrogen system is visualized and on the right side of the table the electricity system including solar energy and non-sustainable energy is visualized.

The properties of the solar panels and gas turbines used in this scenario are elaborated on in 2.4 System properties of components in the electricity system. However, some values need some more elaboration because they are not clear when reading the table. The cost of hydrogen storage is 5% of the total price of hydrogen energy. For instance, if the price of electricity, produced from hydrogen is 20 cents, 1 cent will be added to include the price of hydrogen storage. Also, the "Depreciation fuel cell switch" needs more elaboration. The literature state that shutting down a solid oxide fuel cell can cause degradation and breakdowns due to the high temperatures required for this type of fuel cell. However, after thorough research in the literature, I wasn't able to find literature on how much degradation occurs with a single switch-off. Because of this unknown information, degradation of 5% is assigned to the fuel cells. This means that every time a fuel cell is turned off, the efficiency of the fuel cell will decrease by 5%. For instance, if a fuel cell is turned off three times, it will produce 33 \* (0.95)\*\*3 = 28.3 kWh from one kilogram of hydrogen. Lastly, the number of electrolysis and fuel cells will be explained. This number has been set by running the model and thus has been obtained through empirical practice. The goal was that the model would be able to produce enough electricity coming from hydrogen. The number does not affect the costs because the costs are based on the relationship of kWH and kg, and not how many electrolyzers or fuel cells are present in the model.

Hydrogen system

#### Table 6.2: Setup scenario 2: Hydrogen implementation

Solar panels and gas turbine

Hydrogen storage		Gas turbine				
Hydrogen storage capacity (kg)	2400000	Cost per MW	\$90.9/MWh			
Initital hydrogen in storage	50 %	CO2/cubic foot	0.0550 kg CO2/cubic foot			
Efficiency hydrogen storage	2 %	Gas used per kWh	7.36 cubic feet/kWh			
Costs hydrogen storage	5 %*	Costs carbon credit	\$93,07/1 tonne of CO2			
Fuel cells		Sola	r panels			
# fuel cells	5000	Hectares solar park	11000			
Start-up time fuel cells (h)	4	Life expectancy	25 years			
Range electricity produced	[1kW-2MW]	Efficiency	20%			
kWh per kg hydrogen	33kWh	Investment costs	€650000/ha			
depreciation/ costs fuel cell	\$0.11/kWh	Yearly costs	€2000/ha			
Depreciation fuel cell switch	5%**	Depreciation costs/ha	€3,19/hour - \$3.51/hour			
Electrolyzers						
# electrolyzers	5000					
Maximum W per electrolyser	6MW					
kWh needed for 1 kg hydrogen	51kWh					
Depreciation costs electrolyzers	\$6/kg					

#### Scenario 3: Sustainable priority

Scenario 3 'Sustainable priority' will evaluate the optimal strategy when only the key performance indicators 'sustainability' and 'reliability' are included and 'financial costs' is excluded. It will provide more insight into how the strategy would interact with hydrogen storage when its aim is to provide as much sustainable energy as possible. This scenario is relevant as it demonstrates how the strategy can be optimized when the costs of sustainable energy resources are significantly lower than those of non-sustainable resources. A world where sustainable resources are significantly cheaper than non-sustainable resources is an important step towards reducing the ecological footprint of electricity production and combating climate change. Therefore, this scenario deserves attention. Just like the other tables described in the other scenarios, on the left side of Table 6.3 the hydrogen system is visualized in this scenario, and on the right side of the table the electricity system including solar energy and non-sustainable energy is visualized.

The properties of the solar panels and gas turbines used in this scenario are elaborated on in 2.4 System properties of components in the electricity system. From reading the table "Depreciation fuel cell switch" it is not clear what the 5 % means. In 6.2.1 Scenario 2: Hydrogen implementation an explanation for this value is provided. Logically, the costs of sustainable resources, such as solar panels and hydrogen systems, are removed.

#### Table 6.3: Setup scenario 3: Sustainability priority

Hydrogen storage	•	Gas turbine				
Hydrogen storage capacity (kg)	2400000	Cost per MW	\$90.9/MWh			
Initital hydrogen in storage	50 %	CO2/cubic foot	0.0550 kg CO2/cubic foot			
Efficiency hydrogen storage	2 %	Gas used per kWh	7.36 cubic feet/kWh			
Costs hydrogen storage	-	Costs carbon credit	\$93,07/1 tonne of CO2			
Fuel cells		Solar panels				
# fuel cells	5000	Hectares solar park	11000			
Start-up time fuel cells (h)	4	Life expectancy	25 years			
Range electricity produced	[1kW-2MW]	Efficiency	20%			
kWh per kg hydrogen	33kWh	Investment costs	-			
depreciation/ costs fuel cell	-	Yearly costs	-			
Depreciation fuel cell switch	5%**	Depreciation costs/ha	-			
Electrolyzers						
# electrolyzers	5000					
Maximum W per electrolyser	6MW					
kWh needed for 1 kg hydrogen	51kWh					

#### Hydrogen system

Solar panels and gas turbine

Solar panels and gas turbine

#### Scenario 4: Competitive hydrogen

Hydrogen system

Depreciation costs electrolyzers

The fourth scenario, named "Competitive hydrogen," is included to provide insights into what needs to be done to make electricity, produced with hydrogen as a resource, competitive with natural gas. This scenario assumes that hydrogen electricity is not currently competitive and aims to determine what changes would be necessary for it to be competitive, according to the model. The scenario is particularly important because if hydrogen electricity becomes competitive, it could be integrated into the electricity grid, reducing the reliance on non-sustainable energy sources. For this scenario, drastic changes have been made making electricity coming from hydrogen 10 times cheaper whilst the cost of a carbon credit is increased 10 times. Table 6.4 illustrates the setup for this scenario, and the electricity system consists of both sustainable and non-sustainable energy sources. The table's left side shows the hydrogen system and the right side shows the components of the electricity system.

Table 6.4: Setup scenario 4: Competitive hydrogen

, · · · · · · · · · · · · · · · · · · ·		F	3	
Hydrogen storage	)	Gas turbine		
Hydrogen storage capacity (kg)	2400000	Cost per MW	\$90.9/MWh	
Initital hydrogen in storage	50 %	CO2/cubic foot	0.0550 kg CO2/cubic foot	
Efficiency hydrogen storage	2 %	Gas used per kWh	7.36 cubic feet/kWh	
Costs hydrogen storage	5 %*	Costs carbon credit	\$9307/1 tonne of CO2	
Fuel cells		Sola	r panels	
# fuel cells	5000	Hectares solar park	11000	
Start-up time fuel cells (h)	4	Life expectancy	25 years	
Range electricity produced	[1kW-2MW]	Efficiency	20%	
kWh per kg hydrogen	33kWh	Investment costs	€650000/ha	
depreciation/ costs fuel cell	\$0.0011/kWh	Yearly costs	€2000/ha	
Depreciation fuel cell switch	5%**	Depreciation costs/ha	€3,19/hour - \$3.51/hour	
Electrolyzers		-		
# electrolyzers	5000			
Maximum W per electrolyser	6MW			
kWh needed for 1 kg hydrogen	51kWh			
Depreciation costs electrolyzers	\$0.06/kg			

# 6.3. Scenario results

# 6.3.1. Introduction

Now that the setups of the scenarios have been created and the rationale behind the setups has been explained, it is time to execute the scenarios in the computational model and optimize them using particle swarm optimization. This section will provide individual explanations of the outcomes of each scenario. Afterward, the results will be interpreted by examining the scenario results combined.

# 6.3.2. Scenario 1: Current state

The setup of this scenario does not incorporate the integration of a hydrogen system into the electricity grid. As a result, it serves as a baseline measurement for comparing the other scenario setups that involve the implementation of a hydrogen system. Figure 6.1 depicts the visualization of electricity production per resource over accumulated time, revealing that hydrogen does not contribute to electricity production. This observation is further illustrated in Table 6.5, where it is evident that the fuel cell either did not operate or did not generate any electricity. Consequently, the associated costs are also zero.



Figure 6.1: Electricity production per resource visualized in accumulated over time (own figure)

	Hours running	MW produced	costs \$
Solar panel	41394	411042345.1	3384.05
Hydrogen system	0	0	0
Gas turbine	64604	696772994.9	89587.4

Table 6.5: Results of Scenario 1: Current state

#### Interpretation of result

As the information presented in the table and figure above, natural gas is responsible for most of the electricity production, followed by solar power. In the model natural gas is producing 63% of the electricity and the solar panels are producing 37%. According to [16], solar energy accounted for approximately 14 % of the total electricity generated in 2020 in the Netherlands. Therefore, it can be stated that the assumptions made in 2.4 System properties of components in the electricity system were off and that too many hectares of solar panels were assigned to the model. In this section, the assumption was made that the roofs that have solar panels have covered the maximum area possible.

However this setup does not represent the exact same outcomes of solar energy in the current situation of the Netherlands, this setup is still a valuable representation which will be explained in the remainder of this paragraph. According to [14], the percentage of sustainable production, including other resources such as wind energy, in 2022 in the Netherlands is around 40 %. It also states that the share of sustainable resources in electricity production is increasing. This research aims to optimize the operating strategy of a hydrogen system within an electricity grid. Even though wind energy has different behavior compared to the behavior of solar energy, which can affect the strategy, it is useful to have a model that produces more or less the same amount of sustainable energy as in the real world. The reason for this is that the hydrogen system, which will be implemented in the other scenarios will have to manage approximately the same amount of sustainable energy, as it would have to manage in the real world.

## 6.3.3. Scenario 2: Hydrogen implementation

As stated in 6.2 Scenario setup, The second scenario 'Hydrogen implementation' will be the same as the first scenario but with the implementation of a hydrogen system. The rules created in 5.3 Strategy will be carried out to find the optimized objective function described in Formula 5.3 by finding the most efficient setup of the parameters A until E. In Figure 6.2, in which the electricity production per resource is visualized of one week, we see that hydrogen is still not really participating in the electricity production. This is also visualized in Table 6.6, in which we see that the fuel cell has only produced 438164.6 MW, which accounts for approximately 0.04% of the total electricity production. However, regarding the costs, hydrogen energy is responsible for approximately 35.3% of the costs. These results will be interpreted in the following paragraph.

										99 T
										1991 1931
										09 T
										2ST 7ST
										TST
										J 48
										142 145
										56 T
										73C
										133 EET 130 E
										_ ZZT
										154
										- ΤΖΤ - 8ΤΤ
										STT
										777
										60 T 90 T
										SOT
										91 100 94 94 97 94 91 100
										Vatura 94
										IG Z
										66 88
										Hydrogen
										62 <b>2</b> 5
										Hydrogen
										εζ Ι –
										<u>ک</u> 29
										29 29 20 64 01
								_		- 85
							_			SS
										25
										97
										43
										40 24
										34
										τε
										58 - 52 -
										52
										6T
										19 5 5
						_				οτ
										ζ τ
										τ
100%	%06	80%	70%	60%	50%	40%	30%	20%	10%	%0
1(	01			9		4	(*)	14	-	

Figure 6.2: Electricity production of 1 week in % (own figure)

	Hours running	MW produced	costs (\$)
Solar panel	41394	410927175.1	3384.05
Hydrogen system	87634	438164.6	50608.01
Gas turbine	64601	696450000.3	89545.87

Table 6.6: Results of Scenario 2: Hydrogen implementation

#### Interpretation of result

Based on the information presented in the table and figure above (see Table 6.6 and Figure **??**), it can be observed that the electricity is predominantly generated by the non-sustainable resource natural gas and the sustainable resource solar panels. Despite the intended strategy of utilizing hydrogen, the model demonstrates a tendency to minimize the utilization of hydrogen for electricity generation. From this, it can be concluded that the sustainability aspect of electricity derived from hydrogen does not outweigh the associated costs. This observation aligns with the findings presented in the previous paragraph, indicating that hydrogen accounted for approximately 0.04% of the electricity produced while constituting approximately 35.3% of the total costs. The behavior of the optimized operating strategy will be further elucidated in the subsequent paragraph, utilizing Figure 6.3 and Table 6.7.

Firstly, the operating strategy will be analyzed based on the behavior in the hydrogen storage. Figure 6.3 shows that, in the model with optimized parameters, the hydrogen storage remains mostly full and there is little effort made to utilize this stored hydrogen. This indicates that the parameters have been adjusted in such a way that the optimized strategy avoids using hydrogen in order to maximize the objective function value presented in Formula 5.3. This reasoning is supported by the visualized optimized parameter values presented in Table 6.7, which will now be discussed. It is notable that the fuel cells operate at their minimum value, namely 1 kWh. Additionally, parameter D deviates from this pattern and produces 922,177.2 W per fuel cell. This setting is activated when Rule 5, stated in Section 5.3, is applicable. Rule 5 is triggered when the hydrogen storage needs to be below parameter value C and the solar energy is insufficient to cover the entire electricity demand. However, upon analyzing Figure 6.3, it can be observed that the hydrogen storage never falls below parameter value C, which in this case has a value of 7.16% as indicated in Table 6.7. As a result, Rule 5 is never applied in this scenario, rendering the value of parameter D irrelevant. Therefore, in this scenario, the fuel cells will always produce 1000 Watts, operating at their minimum capacity.



Figure 6.3: State of hydrogen storage over time (own figure)

Table 6.7: Parameter values of scenario 2 after optimization

Parameter	Value
A	32.162 %
В	1000 Watt
С	7.16 %
D	922177.2 Watt
E	1000 Watt

Particle swarm optimization

This section will provide further details regarding the process of obtaining the optimized parameter values, which is performed through the utilization of the particle swarm optimization algorithm, as explained in Section 4.5.2. The progress of optimization in Scenario 2 is visualized in Figure 6.4. This figure illustrates a graph depicting the global best solution of the objective function found at each iteration. In each iteration, the parameters A through E are adjusted to obtain a solution with a different configuration.

As observed, there is a slight improvement in the second iteration compared to the first iteration. The most significant improvement is seen in the third iteration, resulting in a substantial decrease in the graph. In the fourth iteration, a further improvement is achieved, which marks the last improvement obtained. No further enhancements to the objective function are found thereafter. This outcome is expected since the lower bounds of parameter B and parameter E have been reached, and this scenario aimed to minimize the production of hydrogen energy, as explained in the section on the interpretation of results.

Figure 6.5 illustrates the movement of particles throughout the iterations. In each iteration, the particles adjust their positions by modifying the parameter values. Since it is not feasible to visualize all parameters, we have chosen to focus on parameter A and parameter B separately and examine how three particles move within the search space. In this case, it can be observed that the particles move directly toward their respective targets with little directional fluctuation. This suggests that there are not many global minima, in terms of the objective function, found in other regions of the search space.



Figure 6.4: Particle swarm optimization: global best through out the iterations (own figure)



Figure 6.5: Particle swarm optimization: the path of three particles on the search space of parameter A and parameter B (own figure)

# 6.3.4. Scenario 3: Sustainable priority

This scenario aims to investigate the optimal strategy when the key performance indicator of cost efficiency is not considered. In this context, the focus is on maximizing electricity production from sustainable resources, as mentioned earlier in Section 6.2. Figure 6.6 presents the visualization of electricity production per resource of one week in this scenario. It can be observed that hydrogen emerges as the preferred method of electricity production, contributing to 57.4% of the total electricity production. Solar energy follows with a production share of 33%, while natural gas accounts for 9.2% of the total electricity production. These statistics are derived from the data presented in Table 6.8. The subsequent paragraph will provide a more detailed interpretation of these results.



1 week: Distribution of electricity

Figure 6.6: Electricity production of 1 week in % (own figure)

	Hours running	MW produced	costs (\$)
Solar panel	41394	370569413	3384.05
Hydrogen system	83181	635438469.25	0
Gas turbine	38531	101807457.8	304993.41

Table 6.8: Results of Scenario 3: Hydrogen implementation

#### Interpretation of result

The logical expectation in this scenario would be that hydrogen energy is indeed utilized, considering that in Scenario 2, it was not used due to the high associated financial costs. In this scenario, however, the financial costs of hydrogen have been reduced to zero, eliminating this obstacle. The obtained results align with this expectation, as hydrogen is extensively utilized in this scenario. The following paragraphs in this section will explain how the parameter values can be interpreted, as visualized in Table 6.9. Furthermore, the behavior of hydrogen consumption will be elucidated by examining the state of the hydrogen storage over time, as visualized in Figure 6.7.

First, we will examine the hydrogen storage depicted in Figure 6.7. It can be observed that hydrogen storage is frequently at maximum capacity but also experiences significant declines during the winter months. These downward spikes may give the impression that the hydrogen storage is rapidly depleting once it is tapped into, but this is not the case. The dataset used for this research spans from early 2006 to the end of 2015, indicating that Figure 6.7 is highly compressed. However, these abrupt downward fluctuations appear impulsive. Additionally, noteworthy is the consistent frequency of the downward movements in the graph. This pattern arises because hydrogen storage primarily declines during the winter season. During winter, solar energy generation is reduced due to decreased sunlight, as visualized in Figure 4.8. Furthermore, Figure 4.11 demonstrates the inverse pattern concerning electricity demand: it increases during winter compared to summer. These two factors influence hydrogen storage. In winter, there is less solar energy available, leading to a greater reliance on hydrogen production. Moreover, the lack of solar energy intake means that hydrogen storage receives fewer replenishments from newly produced hydrogen. From 2006 to 2015, the hydrogen storage depleted completely six times, resulting in a 5% decrease in the efficiency of electricity production by fuel cells, each time the fuel cells are turned off, affecting the remaining iterations through the scenario. This aspect has been discussed in more detail in Section 6.2.1: Scenario 2: Hydrogen Implementation.

This paragraph focuses on the optimized parameters in this scenario and their influence on the strategy's behavior. Similar to scenario 2, it is evident that the parameters have been pushed to the boundaries of the search space. Parameter A has been set to 99%, while parameter C is set to 1%. Additionally, both parameter B and parameter D are assigned a value of 2000000 W, also reaching the search space limits. The only parameter that does not lie at the extreme end of the search space is parameter E, set at 535206 W. This particular combination of parameters results in the following behavior. When the fuel cells are active, they will always produce 2000000 W during electricity shortages. However, when the hydrogen storage is full and there is an excess of solar energy production, the fuel cell will only generate 535206 W to maintain operation. Furthermore, this strategy is characterized by the fact that when the fuel cells are inactive, they will activate only when the hydrogen storage surpasses parameter A. This is when the hydrogen storage is practically full of hydrogen. This explains why the hydrogen storage, as visualized in Figure 6.7, quickly reaches maximum capacity again after being depleted.



Figure 6.7: State of hydrogen storage over time (own figure)

Table 6.9: Parameter values of scenario 3 after optimization

Parameter	Value
A	99 %
В	2000000 Watt
С	1 %
D	2000000 Watt
E	535206 Watt

#### Particle swarm optimization

Similar to scenario 2, attention will be given to the optimization process executed by the particle swarm optimization algorithm, as explained in 4.5.2. The progress of optimization in scenario 3 is depicted in Figure 6.8. This figure shows a graph of the global best solution of the objective function found per iteration. In each iteration, parameters A to E are adjusted to obtain a solution from a different setting.

As observed, there is a slight improvement in the second iteration compared to the first one. Subsequently, between iteration 2 and iteration 6, the line consistently finds better values for the global minimum and the objective function at an almost constant speed. Further improvements are found in subsequent iterations, including the last and tenth iterations. However, the rate of improvement significantly decreases, indicating that the minimum has been nearly reached. Consequently, the results obtained from the optimized parameters also provide a good indication of the location of the global minimum within the search space.

Finally, the movement of particles throughout the iterations has been captured. Figure 6.9 illustrates how the particles shift by adjusting their parameter values in each iteration. A more detailed explanation of this process can be found in Section 6.3.3. In this case, we observe that the selected particles change direction multiple times. This indicates that there have been several global minima in different locations within the search space before the ultimate global minimum is determined.



Figure 6.8: State of hydrogen storage over time (own figure)



Figure 6.9: State of hydrogen storage over time (own figure)

### 6.3.5. Scenario 4: Competitive hydrogen

This is the final scenario among the four, aiming to investigate the optimal parameter settings to achieve the best strategy when hydrogen is competitive. The results from scenario 2 revealed that the sustainability key performance indicator does not outweigh the financial costs associated with it. Subsequently, scenario 3 examined how the strategy is optimized when electricity production using hydrogen as a resource incurs no financial costs. Therefore, it is interesting to analyze how the strategy behaves when hydrogen can be produced competitively. For this purpose, a specific scenario setup, as described in Section 6.2, has been chosen. In Figure 6.10, which visualizes the electricity production per resource of one week, it can be observed that hydrogen, in this scenario, contributes alongside other resources to electricity generation and produces approximately the same amount as solar energy. Electricity supply, while solar energy contributed 37.1%. Natural gas was responsible for only 13.5% of electricity production. These statistics are derived from the data presented in Table 6.10. The following paragraph will further elaborate on the interpretation of these results.



Figure 6.10: Electricity production of 1 week in % (own figure)

	Hours running	MW produced	costs (\$)
Solar panel	41394	410938940	3384.05
Hydrogen system	83034	546559121	437955376
Gas turbine	38571	150317279	311599.15

Table 6.10: Results of Scenario 4: Hydrogen implementation

#### Interpretation of result

In this scenario, the scenario setup has been made more favorable compared to scenario 2, where the current situation of the electricity grid is tested with the implementation of a hydrogen system. The financial costs associated with hydrogen electricity have significantly decreased in this scenario, while the costs of emitting CO2 have considerably increased. The aim is to create a scenario in which hydrogen becomes a competitive resource for electricity generation, and this is reflected in the results. The following paragraphs in this section will explain how the parameter values, visualized in Table 6.11, can be interpreted and will elucidate the behavior of hydrogen consumption based on the state of the hydrogen storage over time, as visualized in Figure 6.11.

First, let's examine the hydrogen storage in Figure 6.11. It can be observed that the hydrogen storage is frequently full, but also experiences significant drops during the winter months. An explanation for this phenomenon has already been provided in scenario 3, as discussed in Section 6.3.4, and will not be reiterated here. We observe that the hydrogen storage becomes depleted six times, resulting in the fuel cells producing 5% less efficient electricity each time the fuel cell is turned off. This aspect has also been elaborated on earlier in Section 6.2.1 of Scenario 2: Hydrogen. It is worth noting that the pattern observed in the visualization roughly aligns with the visualization of the hydrogen storage of scenario 3 in Figure 6.7.

Despite the similarity in the behavior of the hydrogen storage between scenario 3 and scenario 4, the optimization of the parameters in this scenario is significantly different. This paragraph will focus on the optimized parameters in this scenario and how they influence the behavior of the strategy. Similar to scenario 2 and scenario 3, we observe that the parameters have explored the boundaries of the search space. Parameter A is set at 99%, parameter C at 2,000,000 W, and parameter E at 1000 W. Only parameter C and parameter D do not reside at the extreme ends of the search space, with parameter values of 29.9% and 1,868,058, respectively. This parameter combination yields the following results: when the fuel cells are active, they will produce 2,000,000 W until the hydrogen storage drops below 29.9%. Once the hydrogen storage falls below this threshold, the strategy dictates that the fuel cells should produce 1000 W per fuel cell until the hydrogen storage exceeds parameter A again, which is set at 99%. This explains why the hydrogen storage, as visualized in Figure 6.11, quickly replenishes after being depleted. Lastly, it is noteworthy that when solar energy generates enough electricity to meet the demand of the grid, and the fuel cells are active, they will produce 1,868,058 W.



Figure 6.11: State of hydrogen storage over time (own figure)

Table 6.11: Parameter values of scenario 4 after optimization

Parameter	Value
A	99 %
В	2000000 Watt
С	29.9 %
D	1868058 Watt
E	1000 Watt

Particle swarm optimization

Similar to the previous scenarios, attention will be given to the optimization process executed by the particle swarm optimization algorithm. The progress of optimization in scenario 4 is visualized in Figure 6.12. This figure depicts a graph where the global best solution of the objective function is found per iteration. In each iteration, parameters A to E adapt to obtain a solution with a different configuration.

As can be observed, there is a slight improvement in the second iteration compared to the first. Then, between iteration 2 and iteration 6, the line steadily finds a better value for the global minimum and the objective function at an almost constant rate. Subsequently, further improvements are still found until the final and tenth iterations. However, the rate of improvement has significantly decreased, indicating that the minimum has been nearly reached. Consequently, the results regarding the optimized parameters also provide a good indication of the location of the global minimum within the search space.

Finally, the movement of particles throughout the iterations is documented. Figure 6.13 illustrates how the particles shift in each iteration by adjusting the parameter values. For a more detailed explanation, please refer to section 6.3.3. In this case, it can be observed that the selected particles do not change direction frequently. This indicates that for the two parameters visualized, the algorithm did not encounter many other global minima within the search space.



Figure 6.12: State of hydrogen storage over time (own figure)



Figure 6.13: State of hydrogen storage over time (own figure)
### 6.4. Overview of scenarios and its results

In summary, each of the four examined scenarios yields distinct outcomes. This section will provide individual explanations for each scenario based on Table 6.12. In the subsequent chapter, namely, the Discussion (Chapter 7), these results will be further analyzed and interpreted.

In the first scenario, "Current state," the behavior of the model was examined without the implementation of a hydrogen system. This provides a baseline understanding of the changes observed in the other three scenarios where a hydrogen system is implemented. In this scenario, approximately 37.1% of the electricity is generated from solar energy, while approximately 62.9% is derived from natural gas. In terms of costs, the majority (96.4%) is attributed to the energy resource of natural gas, with the remaining costs associated with operating solar parks.

The second scenario, named "Hydrogen implementation," analyzes the behavior of the model with the implementation of a hydrogen system and an optimized operating strategy. It is notable that the optimized strategy prefers to generate minimal electricity from hydrogen. This is evident from the optimization of parameters B and E, which are set to their lowest values, specifically 1000 Watts. Parameter D deviates from this trend and produces 922177.2 Watts but does not contribute to the strategy, as explained earlier in 6.3.3. Electricity generated from hydrogen accounts for only 0.04% of the total electricity production, while the distribution of electricity from solar parks and natural gas remains largely unchanged from Scenario 1. However, there is a clear difference in costs compared to Scenario 1. It becomes apparent that the costs associated with solar parks represent only 2.4% of the total costs, while hydrogen accounts for 35.3% of the total costs. The remaining 62% of costs are attributed to natural gas. It is noteworthy that hydrogen contributes significantly less to the overall electricity generation but incurs a substantial portion of the costs. This observation will be further addressed in the subsequent chapter, the Discussion (Chapter 7).

Next, in Scenario 3, "Sustainable priority," the focus was on examining the behavior of the optimized strategy when the cost of hydrogen was significantly lower than that of natural gas. This investigation was motivated by the findings of Scenario 2, "Hydrogen implementation," which revealed that hydrogen did not contribute significantly due to its high costs in the simulation runs. In this scenario, the costs of electricity generated from hydrogen were reduced to zero. As a result, the optimized values of parameters A, B, C, and D approached their respective boundary values. Parameter A had a value of 99%, and parameter C had a value of 1%, providing information on the fill level of the hydrogen storage. Additionally, parameter B and parameter D had a maximum value of 2000000 Watts, while parameter E had a value of 535206 Watts. Parameter E indicates the amount of electricity produced when Rule 7 is executed, as explained in 5.1, and it is the only parameter that does not reach the boundary of its search field. The impact of these parameter values is reflected in the model's output. Solar energy now accounts for only 33% of the total electricity production, while hydrogen shows a significantly decreased, representing only 9.6% of the total electricity production.

The last scenario, "Competitive hydrogen," aims to explore the conditions under which hydrogen becomes a significant contributor to electricity generation in the model, considering the costs of the hydrogen system and natural gas. In this scenario, the costs of hydrogen were reduced by a factor of 10, while the costs of carbon credits were increased by the same factor. The optimized results are as follows: parameter A is set to 99%, parameter B to 2000000 Watts, and parameter E to 1000 Watts, all of which have been optimized to the boundaries of their search spaces. Additionally, parameter C has a value of 29.9%, and parameter D is set to 1868058 Watts. As a result, hydrogen accounts for 52.8% of the total electricity production, followed by solar energy at 37.1%, and natural gas at 10.1%. The costs are predominantly driven by the hydrogen system. This aspect will be further addressed in the subsequent chapter, the Discussion (Chapter 7).

	Output	Scenario 1: Current state	1: Current state Scenario 2: Hydrogen implementation	Scenario 3: 3	Scenario 4: Competitive hydrogen
	Parameter A	Т	31.2%	%66	
	Parameter B	I	1000 W	200000 W	
Parameters	Parameter C	1	7.2%	1%	
	Parameter D	1	922177.2 W	200000 W	
	Parameter E	1	1000 W	535206 W	
	MW produced Solar (%)	37.1%	37.1%	33.5%%	
	MW produced Hydrogen (%)	%0	0.04%	57.4%	
Borformance variablee	MW produced Natural gas (%)	62,90%	62.9%	9.2%	
	Financial costs Solar (%)	3.6%	2.4%	1.1%	
	Financial costs Hydrogen (%)	0	35.3%	%0	
	Financial costs Natural gas (%)	96.4%	62.4%	98.8%	

Table 6.12: Overview of the scenario's results

# Discussion

## 7.1. Introduction

In this chapter, we will reflect on the research conducted. Firstly, we will interpret the results of the scenarios discussed in the previous chapter (Chapter 6). Additionally, we will compare these results with findings from related studies in the existing literature. Furthermore, we will reflect on the model itself and discuss its limitations. We will also review relevant literature to determine if similar observations were made regarding their models. Subsequently, we will address the limitations of the research and assess the validity of the study.

## 7.2. Result interpretation

In this section, the results will be further interpreted. Due to the multiple aspects analyzed in the interpretation of the results, it has been chosen to address them through the following five questions. The subsequent paragraphs dedicated to this section will aim to answer the questions formulated below.

- 1. Has finding the most optimized strategy for each scenario been performed correctly?
- 2. Why is hydrogen so expensive compared to natural gas in scenario 2, Current state? What are the cost components of hydrogen in this scenario?
- 3. What needs to be done to transition from scenario 2, Current state, to scenarios 3 and 4? How long will this transition take?
- 4. What are the limitations of the model, and how have they influenced the results (including parameter settings)?
- 5. What are the findings of literature that conduct similar research, and what are the similarities and differences?

#### 7.2.1. Question 1: Optimization of operating strategy

When addressing the first question, the focus will not be on the results themselves, but rather on assessing whether the optimization process was successful for each scenario. The particle swarm optimization (PSO) technique, explained in Section 4.5.2, will be examined to determine if the obtained results are truly the optimized outcomes.

In the first scenario, Current state, no optimization was performed, so we will start by analyzing scenario 2, Hydrogen implementation, using Figure 6.4, which illustrates the optimization process. It can be observed that a saturated point is reached quite early, specifically in the fourth iteration. As discussed earlier in the previous chapter, there is a logical explanation for this, namely that the strategy performs best when the boundaries of the search space are approached.

Moving on to the third scenario, Sustainable priority, visualized in Figure 6.8, we see that the optimization process is not as rapid as in scenario 2. Until the final iteration, we observe ongoing optimization of the operating strategy. On one hand, it could be argued that the obtained operating strategy has not yet reached its optimal configuration, or that we are not certain of it at this point. To determine this, more iterations would need to be conducted until no further improvement is found for a few iterations. On the other hand, the figure shows that improvements have significantly diminished since the sixth iteration, suggesting that further optimization is unlikely to yield substantial enhancements. This claim is supported by the parameter settings provided in Table 6.8, which, similar to scenario 2, tend to approach the extremes of the search space. However, to obtain even more precise values for the optimal operating strategy in this scenario, additional iterations would be necessary.

Regarding the optimization of the last scenario, Competitive hydrogen, visualized in Figure 6.12, we observe a similar behavior to scenario 3. In this scenario as well, it takes time for the improvements to taper off, and during the last two iterations, no further enhancements are achieved. Looking at the variables in Table 6.10, we see that the parameters are positioned near the boundaries of the search space, except for parameter D, which reaches 1868058 Watt but does not quite reach 2000000 Watt. It remains unclear whether this is the optimal setting or if this parameter was converging towards 2000000 Watt. To obtain even more precise values for the optimal operating strategy in this scenario, additional iterations would be necessary.

#### 7.2.2. Question 2: Explanation of expensive hydrogen in scenario 2, Current state

To provide a thorough answer to this question, a cost breakdown of hydrogen energy will be conducted, specifically focusing on the cost per kWh and the contribution of hydrogen storage, fuel cells, and electrolyzers. The cost distribution per kWh is presented in Table 7.1.

	1 kWh Hydrogen energy costs	%
Total costs	\$0.305	100%
Hydrogen storage	\$0.015	5%
Fuel cells	\$0.11	36%
Electrolysers	\$0.18	59%

Table 7.1: Cost breakdown of hydrogen elecrticity

According to the table provided, the total cost amounts to \$0.305 per 1 kWh for hydrogen energy. In comparison, 1 MW generated by natural gas costs approximately \$90.9 in this model. Together with the associated carbon credit costs, as explained in 5.3, this translates to approximately \$0.039 per kWh, which is significantly cheaper than the cost of electricity produced from hydrogen. The costs of hydrogen are primarily attributed to fuel cells and electrolyzers. Approximately 59% of the total costs to produce 1 kWh are incurred by electrolyzers. These costs are calculated based on the assumption that it costs 6% to produce one kilogram of hydrogen. The amount of hydrogen required for 1 kWh is then determined, and the cost is calculated accordingly. Fuel cells account for 36% of the costs, followed by hydrogen storage, which contributes 5% of the costs. It should be noted that the efficiency of fuel cells may decrease over the course of the simulation run and may deviate from the values in the table above. As discussed earlier in Scenario 2: Hydrogen Implementation, the fuel cells in this study are modeled to become less efficient as they are switched on and off more frequently. In the case where a fuel cell is turned off, the costs to produce 1 kWh will increase. These costs will be reflected in the electrolyzer costs, as the electrolyzer will need to produce more hydrogen to generate 1 kWh.

This topic will be revisited in the answer to question 6, where a comparison with other literature will be made. It will be investigated whether the cost distribution, particularly the share of electrolysis, aligns with findings from other studies.

#### 7.2.3. Question 3: Transition from scenario 2, Current state, towards Scenario 3 and 4

Now that the costs of hydrogen have been defined in the model, the question arises: What actions need to be taken to transition to a situation where hydrogen electricity is utilized? To address this question, we will examine the pathways from scenario 2, "Current State," to scenario 4, "Competitive Hydrogen," where hydrogen systems are effectively employed, based on the results from this research.

Based on the model used in this research, two approaches can be considered to make hydrogen competitive with natural gas. The first approach involves increasing the cost of carbon credits. Governments can implement policies to reduce the availability of carbon credits, creating scarcity and driving up prices. Consequently, the costs associated with natural gas electricity production would also rise. However, a drawback of this strategy is the potential increase in overall costs, which may result in electricity becoming unaffordable. Although this policy measure can incentivize a reduction in gas-based electricity generation, it also poses challenges in terms of affordability.

The second approach, albeit easier said than done, is to enhance the efficiency and cost-effectiveness of hydrogen systems. As previously discussed, the total cost of 1 kWh from hydrogen is \$0.305, while that of natural gas is \$0.039. This paragraph further explores the prospects of achieving cost parity between hydrogen electricity and natural gas. According to [43], it can be assumed that electrolyzers experience an annual cost reduction of approximately 3%. Although the outlook for fuel cells, as noted in [6], is less optimistic than for electrolyzers, for simplicity, we assume that fuel cells also achieve a 3% cost reduction annually. Assuming that the price of natural gas remains constant and the costs of fuel cells and electrolyzers decrease by 3% each year, it would take approximately 67 years to reach cost parity. To expedite this process, a policy measure could involve increased investment in research and development of these components required in a hydrogen system.

Considering the slow pace of development, a combination of both policies—raising carbon credit prices and intensifying research and development—seems most feasible for expediting the competitiveness of hydrogen electricity. It should be noted that optimizing the operating strategy becomes relevant only when hydrogen electricity can be produced at a lower cost and competes effectively with other resources. Until then, the primary focus should be on improving the components of the hydrogen system, such as fuel cells and electrolyzers.

Nonetheless, numerous other measures can be undertaken to promote sustainable energy or make fossil energy less appealing. For instance, consumers have the potential to decrease their electricity consumption during periods of high CO2 emissions[44]. Nowadays, a significant number of consumers lack awareness regarding the specific hours of the day when CO2 emissions are most intensive and when electricity usage is the least harmful[44]. An effective measure could be to spread awareness to the consumers about what times are the best to consume electricity, which will help hydrogen energy to be more competitive with electricity produced from natural gas.

Implementing the measure described in the paragraph above needs comprehensive research itself and is beyond the scope of this research. The two measures outlined in this study, namely, increasing price of carbon credits and more investments in research and development, have demonstrated their efficacy based on the results of scenario analyses. Consequently, this research exclusively focuses on the examination of these measures.

#### 7.2.4. Question 4: Model limitations

The model employed in this research possesses certain limitations that have influenced the results. These limitations, along with notable findings, will be discussed in the following paragraphs to provide further context to the results.

One significant limitation of the model lies in the static values assigned to electrolyzers and fuel cells. As documented in [39] and [71], each input exhibits its own efficiency. To maintain the manageability of the study, a specific efficiency value was chosen. However, incorporating input-dependent efficiencies would undoubtedly impact the results.

Moreover, the costs associated with hydrogen storage are expressed as a percentage of the total costs, following the rationale presented in [58]. This approach implies that the costs incurred by fuel cells, electrolyzers, and inverters influence the costs of hydrogen storage. In reality, these costs cannot be determined in the same manner and are independent of the total costs. Additionally, the costs associated with turning off the fuel cell remain unclear. Despite extensive efforts to find relevant literature

on this topic, no definitive value was found. Consequently, a degradation value for the fuel cell was defined (as explained in Section 6.2.1).

Furthermore, the simulation runs in the model were conducted in a controlled environment, assuming that all components function as described in the rest of the research. It remains uncertain whether these results would translate directly to the real world or if components may exhibit different behaviors than those assumed in this study.

The penultimate limitation pertains to the rules of the operating strategy. The rules dictate the actions to be taken in various situations. For this research, a set of seven rules was chosen, but they could be expanded and made more detailed. However, to ensure the feasibility of the study, only seven rules were employed. Naturally, this limitation also affects the results.

Building upon the previous paragraph, a notable result in scenario 2, "Hydrogen implementation," warrants further attention. As previously mentioned, the costs of hydrogen electricity play a significant role in this scenario, making the parameter adjustment of parameter D to 922177.2 Watts noteworthy. Intuitively, one would aim to minimize hydrogen production, as reflected in the other parameters. However, when the hydrogen storage is full, and Rule 5 is applied, this is not the case. To analyze why this discrepancy arises, the next paragraph will delve into the consequences of setting parameter D to its minimum value.

When this value assumes the minimum value, similar to parameters B and E, it implies that the fuel cells produce 1000 Watts, while the remaining solar energy is utilized for hydrogen production. In cases where a substantial amount of solar energy is generated and the hydrogen storage is already full, a considerable amount of solar energy is wasted in the model. It is possible that, in such situations, it becomes more cost-effective to utilize more hydrogen, thereby maximizing the utilization of solar energy for hydrogen production and minimizing wastage. Thus, the parameter adjustment of D is optimized to reduce costs.

#### 7.2.5. Question 5: Comparing results with literature

The response to the last question will reflect on the model results and compare them with existing literature. As mentioned earlier in Section 1.2 Knowledge gap, there is limited research available that conducts similar studies. This scarcity is also evident in Table 3.2. Nonetheless, it is worthwhile to explore the similarities and differences with existing literature.

One study that has some resemblance to my research, based on my literature review, is the study performed by [98]. This study also utilizes a solar park, gas turbine, and hydrogen storage system to generate electricity. It also seeks a multi-objective operating strategy optimized for both economic costs and carbon emissions. However, despite these similarities, this research still differs significantly from the current study in terms of methodology, optimization strategies, and types of results. Consequently, a direct comparison of results is not feasible due to substantial disparities between the two studies, despite both aiming to find a strategy optimized based on a multi-objective research goal.

However, valuable insights can still be obtained from this research. In this study, it is observed that similar to this thesis, electrolyzers constitute the largest cost component. Considering the cost distribution of the hydrogen system in this research, electrolyzers account for approximately 41% and 75% of the total costs in the two examined strategies concerning electricity generation from hydrogen. Hence, despite the inherent differences between the studies regarding the costs of electrolysis, it can be asserted that it constitutes a substantial portion of the total electricity production costs from hydrogen. This finding aligns with the identified share of electrolysis in this thesis, as presented in Table 7.1. Another noteworthy observation made in [98] is that the hydrogen storage system rapidly depletes. This behavior is also evident in this research, where the hydrogen storage can quickly fill up and deplete. Finally, it is worth mentioning that many pieces of literature have employed static efficiencies for electrolyzers and fuel cells. This assumption of static efficiencies was also addressed as a limitation in Question 4. However, it is observed that this assumption of static efficiencies is also utilized in [98].

## 7.3. Limitations of the research

The limitations of the model are also limitations of the research itself, as discussed earlier in the section 7.2 Results Interpretation. Additionally, this thesis focuses on a fraction of the entire energy sector. Other factors such as the energy market, regulations, and societal acceptance all have an impact on the electricity market. The decision to solely focus on optimizing the operating strategy for hydrogen storage while excluding other aspects can be seen as a limitation.

Furthermore, certain boundaries were set in this research, which can be considered as limitations. For instance, only solar panels were chosen as the sustainable resource, while a large-scale electricity system typically utilizes other sustainable resources such as wind turbines, biomass, or hydro energy.

Moreover, the system described in this research and the corresponding simulation runs are conducted in a controlled environment. This means that the results of the study, in addition to the aforementioned demarcations, can also be influenced by external factors that are not present in the described system. For example, the performance of fuel cells in terms of energy production may differ on a larger scale.

## 7.4. Validity of the research

There are two ways to assess the validity of the research. One approach is to examine the existing literature and compare the findings of this study with other research. The other method of validation is to replicate the system in the real world and compare the results with those obtained in this research. However, the latter approach is not feasible because there is no existing system in the world with the exact properties and scale as the one described in this study. Moreover, constructing such a system would require significant costs, making this research primarily reliant on validation based on existing literature.

The model results and their comparison with the existing literature have already been addressed in Question 5 in the Results Interpretation section. As highlighted in that section and also in section 1.2 Knowledge gaps, the literature on optimizing an operating strategy in an electricity system that includes hydrogen is limited. Additionally, the studies conducted in this field have explored various electricity systems, such as systems with or without batteries or specific wind parks used for the research. Given that this research focuses on a specific electricity system (see Figure 5.1), it becomes challenging to validate the outcomes against the existing literature, which is already scarce.

# Conclusion

## 8.1. Introduction

This chapter will consist of the following. First, the main research question will be introduced, followed by the answers to the sub-questions. Once these answers are provided, the main research question will be addressed. Additionally, the scientific and societal significance of the research findings will be discussed. Finally, potential avenues for future research will be suggested.

## 8.2. Answering research questions

The goal of this thesis is to determine an optimized operating strategy for an electricity grid incorporating hydrogen technology while considering three key performance indicators: financial costs, reliability, and sustainability. The aim is to minimize the overall cost while ensuring reliability and sustainability. To achieve this, the particle swarm optimization algorithm is employed as the chosen method to identify the most optimal operating strategies. This approach helps address the research question:

#### How can electricity demand be covered by a system of solar power, conventional power supply, and hydrogen systems, optimized for reliability, sustainability, and cost-efficiency, by adjusting the operating strategy?

The sub-questions, as outlined in 1.5 Research question and sub-questions, have been formulated to provide a comprehensive answer to the main research question. In the subsequent paragraphs, each sub-question will be addressed, leading to the ultimate answer to the main research question. However, the answer to sub-question 6 will be presented after the main conclusion, as it entails recommendations to stakeholders, which can be determined based on the main findings.

### 8.2.1. Sub-question 1

# According to the scientific literature, what are the most common approaches to finding an optimized operating strategy of an electric system including hydrogen technology?

A comprehensive answer to this question provides a clear understanding of how the literature addresses this problem and offers more context on research conducted in this area. Firstly, it is evident that studies focusing on finding an optimal strategy are limited, with most studies primarily concentrating on economic cost reduction without incorporating environmental aspects. Secondly, the use of hourly data is prevalent in optimizing operating strategies. Thirdly, there are variations in approaches among studies, as they utilize different resources and storage methods, including solar panels, wind turbines, batteries, and others. It is worth noting that many studies employ rule-based or fuzzy logic controllers, which are rule-based approaches, to determine an optimized operating strategy. Additionally, artificial intelligence methods are often utilized for optimization purposes.

Furthermore, it is notable that a significant portion of the literature concerning electricity systems incorporating hydrogen primarily focuses on the component ratio within the electricity grid, such as the number of electrolyzers, hydrogen storage capacity, and so on, rather than finding an efficient operating strategy.

#### 8.2.2. Sub-question 2

## How should the key performance indicators in the operating strategy, referred to in 1.3 Research goal, be prioritized to conceptualize the optimal operating strategy?

In this study, the decision was made to express all key performance indicators in terms of costs, measured in euros, as explained in Chapter 5. The key performance indicator "Cost efficiency" represents the aggregate levelized cost of the components within the electricity system. "Sustainability" is calculated by quantifying CO2 emissions in terms of euros, considering the costs of carbon credits. Due to the significant disruptive consequences and associated high costs, "Reliability" is not incorporated into the model. The objective function is presented in Formula 8.1.

$$Objective \ function(\mathbf{\epsilon}) = Sustainability(\mathbf{\epsilon}) + Reliability(\mathbf{\epsilon}) + Financial \ efficiency(\mathbf{\epsilon})$$
(8.1)

By formulating this objective function, various strategies can be tested and compared. Furthermore, the strategies can be evaluated by examining the contribution of each key performance indicator to the total costs.

In the literature, different approaches have been taken in formulating multi-objective strategies that incorporate both economic and environmental costs in their analysis. For instance, [86] does not modify the parameters within the strategy itself but rather explores the trade-off between energy costs and CO2 emissions when sustainable resources have a greater role in the electricity system. On the other hand, [98] examines two distinct strategies that prioritize financial costs and sustainability as separate outputs of the study.

#### 8.2.3. Sub-question 3:

#### According to scientific literature, which research approach, methodology, and methods are appropriate for determining an optimized operating strategy for electricity supply in the considered system?

This research utilizes discrete event simulation, where datasets on solar electricity generation and electricity demand are used, given data on an hourly basis. Similar studies in the literature, as discussed in Chapter 3, have also employed this approach, justifying its application in this study. The work by Law and Kelton (2007) offers a systematic framework that guided the chosen modeling research design [56]. Additionally, particle swarm optimization, a heuristic-based algorithm, was employed to optimize the operating strategies. This algorithm involves a swarm of particles, each representing a candidate solution to the optimization problem, which strategically adjusts its parameters to obtain the global best solution. To further enhance its efficiency, linear decay was incorporated into the algorithm. Particle swarm optimization is commonly used in similar studies, as evidenced in Table 3.2. By employing this optimization algorithm, an efficient search for the minimal objective function, and subsequently the optimal operating strategy described in the answer to sub-question 2, can be conducted.

#### 8.2.4. Sub-question 4:

## In the considered case, how can an optimal, with respect to the key performance indicators defined in sub-question 2, operating strategy be determined?

To address this sub-question, a model was developed using the Python programming language. The purpose of this model is to simulate an electricity system where electricity is generated and consumed on an hourly basis. The energy sources modeled in the system include solar electricity, energy from natural gas, and a hydrogen storage system capable of storing electricity.

The operating strategy aims to determine which energy source supplies electricity to the grid for each hour. This decision is guided by seven rules presented in Table 5.1. These rules incorporate parameters A through E, which can assume different values to define different strategies. The performance of the operating strategy is measured using the objective function mentioned in the answer to sub-question 2, and the goal is to minimize it. As mentioned in the answer to sub-question 3, the particle swarm optimization algorithm is employed to efficiently search for this optimal solution.

#### 8.2.5. Sub-question 5:

## How can the optimization model, found in sub-question 4 be used to optimize realistic future scenarios?

Chapter 6 presents the findings of the study, which includes four different scenarios designed to address the research question. Each scenario explores a different future circumstance, leading to distinct operating strategies.

The first scenario, "Current state," examines the scenario where no changes occur regarding the hydrogen storage system. It represents a future situation without a hydrogen system, with electricity generated solely from solar energy and natural gas. In this scenario, natural gas accounts for 63% of the electricity production, while solar panels contribute 37%.

The second scenario, "Hydrogen implementation," investigates the optimization of the operating strategy with the addition of a hydrogen storage system to the first scenario. The results show that electricity generated from hydrogen represents approximately 0.04% of the total electricity production, with natural gas accounting for 62.9% and solar electricity for 37.1%. The reason behind this is that electricity from hydrogen is much costlier than that from natural gas. Consequently, the optimal strategy avoids using hydrogen electricity, even though it is sourced from a sustainable resource.

The third scenario, "Sustainable priority," examines the optimal operating strategy when hydrogen electricity incurs no costs. This scenario explores a future situation where hydrogen electricity becomes significantly cheaper than natural gas, aiming to determine if the electricity system can consistently provide sustainable electricity to the grid. The results indicate that electricity from hydrogen accounts for approximately 57.4% of the total production, while natural gas contributes only 9.2%, and solar electricity represents 33.5%. This scenario yields the most favorable outcomes so far, as the optimal operating strategy generates a substantial portion of electricity from sustainable resources, resulting in the lowest objective function.

Lastly, the study investigates a future scenario where hydrogen electricity competes with natural gas electricity. To achieve this scenario, significant cost reductions for hydrogen electricity and a substantial increase in carbon credit prices are implemented. The results reveal that electricity from hydrogen accounts for approximately 49.3% of the total production, while natural gas contributes merely 13.6%, and solar electricity represents 37.1%. In terms of costs, solar panels account for only 0.01% of the total, while hydrogen accounts for 99.4%, and natural gas for 0.6%. Therefore, it can be concluded that in this scenario, hydrogen is utilized, and solar energy performs better compared to the previous "Competitive hydrogen" scenario.

#### 8.2.6. Main conclusion

In conclusion, the main research question has been addressed by examining the sub-questions and synthesizing the findings. By quantifying CO2 emissions in terms of financial costs, the key performance indicators of reliability, sustainability, and financial costs have been compared, leading to the formulation of a multi-objective function. Particle swarm optimization has been identified as a suitable method for minimizing this function and determining the optimal operating strategy.

Through the use of a simulation model, various scenarios were examined to investigate the feasibility of implementing a hydrogen storage system. The results indicate that the implementation of a hydrogen system, with the present circumstances is not economically viable. The sustainability benefits of hydrogen electricity do not outweigh the financial advantages of non-sustainable electricity. Even when considering the costs of carbon credits, the operating costs of a hydrogen system are significantly higher than those of natural gas. The main cost drivers are the high costs associated with electrolysis, followed by the costs of fuel cells.

To achieve a future scenario in which hydrogen electricity competes with natural gas, substantial cost reductions in the components of the hydrogen system are required, along with an increased price of carbon credits. These measures would enhance the economic viability of hydrogen electricity generation and create a more competitive landscape in comparison to natural gas.

#### 8.2.7. Sub-question 6:

## What are the recommendations that can be formulated for the problem's stakeholders based on the results of sub-question 5?

Based on the main conclusion, it can be determined that generating electricity from hydrogen is not currently profitable, as indicated by the model results. Assuming that the stakeholders, including energy companies and the Dutch government, desire the implementation of a hydrogen system in the electricity grid that is competitive with natural gas, two recommendations can be made.

As mentioned earlier in the section 7.2 Result interpretation, one recommendation is for the government to increase the price of carbon credits. However, a drawback of this strategy is that it would raise the overall costs and could potentially make electricity unaffordable. On the positive side, it could serve as an incentive to focus on alternative sustainable energy sources, leading us to the second recommendation.

The second recommendation is to invest more in research and development of hydrogen systems. This investment can be made by both energy companies and governments. A combination of both recommendations appears to be the best option for expediting the transition to a future where hydrogen electricity competes with natural gas electricity.

By taking these measures, stakeholders can work towards a future where hydrogen electricity becomes a viable and competitive option, contributing to a more sustainable electricity system.

## 8.3. Suggestions for further research

As previously mentioned in the knowledge gap section (see 1.2), there is a knowledge gap regarding efficiently operating hydrogen systems [118]. Furthermore, the literature review conducted in Chapter 3 identified a scarcity of research that considers both financial costs and environmental costs when exploring the optimal operating strategy. Although this thesis aims to address this knowledge gap, there is still ample room for further contributions on this topic. For instance, exploring electricity systems with different components, such as batteries, wind energy, etc., could be investigated.

Moreover, this research has its limitations, which can serve as inputs for further research. As discussed in the limitations of the research section (see 7.3), the model used in this study is limited to using only solar electricity. However, when supplying a national electricity grid, it is likely that other sustainable resources, such as wind turbines, would also contribute to electricity generation. Therefore, expanding the model of this research to include other sustainable resources would add value. Additionally, the strategy employed in this research is rule-based and consists of seven rules. It would be interesting to explore alternative rules or expand the existing rules to formulate a more sophisticated operating strategy.

Furthermore, as previously mentioned in the recommendations, further research is needed on the components within a hydrogen system, such as electrolyzers and fuel cells. Therefore, I would also emphasize the importance of conducting research on these components to make research on the optimal operating strategy for a hydrogen system more relevant.

# References

- Zainul Abdin, Kaveh Khalilpour, and Kylie Catchpole. "Projecting the levelized cost of large scale hydrogen storage for stationary applications". In: *Energy Conversion and Management* 270 (2022), p. 116241.
- U.S. Environmental Protection Agency. Sources of Greenhouse Gas Emissions. https://www.epa.gov/ghgemissions/sources-greenhouse-gas-emissions. [Accessed: May 11, 2023]. 2021.
- [3] Nabil A Ahmed, Masafumi Miyatake, and AK Al-Othman. "Power fluctuations suppression of stand-alone hybrid generation combining solar photovoltaic/wind turbine and fuel cell systems". In: *Energy Conversion and management* 49.10 (2008), pp. 2711–2719.
- [4] Joakim Andersson and Stefan Grönkvist. "Large-scale storage of hydrogen". In: *International journal of hydrogen energy* 44.23 (2019), pp. 11901–11919.
- [5] Amirmohammad Behzadi et al. "An efficient renewable hybridization based on hydrogen storage for peak demand reduction: A rule-based energy control and optimization using machine learning techniques". In: *Journal of Energy Storage* 57 (2023), p. 106168.
- [6] Hans Böhm, Sebastian Goers, and Andreas Zauner. "Estimating future costs of power-to-gasa component-based approach for technological learning". In: *International Journal of Hydrogen Energy* 44.59 (2019), pp. 30789–30805.
- [7] Ulf Bossel and Baldur Eliasson. *Energy and the Hydrogen Economy*. Tech. rep. 2003. URL: https://afdc.energy.gov/files/pdfs/hyd\_economy\_bossel\_eliasson.pdf.
- [8] Jörn Brauns and Thomas Turek. "Alkaline water electrolysis powered by renewable energy: A review". In: *Processes* 8.2 (2020), p. 248.
- [9] Cory Budischak et al. "Cost-minimized combinations of wind power, solar power and electrochemical storage, powering the grid up to 99.9% of the time". In: *journal of power sources* 225 (2013), pp. 60–74.
- [10] Abba Lawan Bukar et al. "A rule-based energy management scheme for long-term optimal capacity planning of grid-independent microgrid optimized by multi-objective grasshopper optimization algorithm". In: *Energy conversion and management* 221 (2020), p. 113161.
- [11] Ulrich Bünger et al. "Large-scale underground storage of hydrogen for the grid integration of renewable energy and other applications". In: *Compendium of hydrogen energy*. Elsevier, 2016, pp. 133–163.
- [12] Bruno Burger and Dirk Kranzer. "Extreme high efficiency PV-power converters". In: 2009 13th European Conference on Power Electronics and Applications. IEEE. 2009, pp. 1–13.
- [13] CAS. Lithium Batteries Hydrogen Fuel Cells. Website. Year. URL: https://www.cas.org/ resources/cas-insights/sustainability/lithium-batteries-hydrogen-fuel-cells.
- [14] CBS. "Aandeel hernieuwbare elektriciteit met 20 procent gestegen in 2022". In: (Mar. 2023). URL: https://www.cbs.nl/nl-nl/nieuws/2023/10/aandeel-hernieuwbare-elektriciteit -met-20-procent-gestegen-in-2022.
- [15] CBS. "Meer elektriciteit uit kolen en hernieuwbare bronnen, minder uit gas". In: CBS (Sept. 2022). URL: https://www.cbs.nl/nl-nl/nieuws/2022/36/meer-elektriciteit-uit-kolen-en-hernieuwbare-bronnen-minder-uit-gas.
- [16] Centraal Bureau voor de Statistiek (CBS). Hernieuwbare energie in Nederland 2020: Zonneenergie. https://www.cbs.nl/nl-nl/longread/aanvullende-statistische-diensten/ 2021/hernieuwbare-energie-in-nederland-2020/5-zonne-energie. [Accessed on: April 30, 2023]. 2021.

- [17] Xiao Cheng et al. "Single crystal perovskite solar cells: development and perspectives". In: *Advanced Functional Materials* 30.4 (2020), p. 1905021.
- [18] Chevron. Chevron Supports Innovation to Scale Climate Solutions. https://www.chevron. com/-/media/chevron/sustainability/documents/2021-chevron-supports-innovationto-scale-climate-solutions.pdf. Accessed: April 29, 2023. 2021.
- [19] Juan-Carlos Ciscar et al. "Physical and economic consequences of climate change in Europe". In: *Proceedings of the National Academy of Sciences* 108.7 (2011), pp. 2678–2683.
- [20] David Connolly et al. "The technical and economic implications of integrating fluctuating renewable energy using energy storage". In: *Renewable energy* 43 (2012), pp. 47–60.
- [21] Mugdha V Dambhare, Bhavana Butey, and SV Moharil. "Solar photovoltaic technology: A review of different types of solar cells and its future trends". In: *Journal of Physics: Conference Series*. Vol. 1913. 1. IOP Publishing. 2021, p. 012053.
- [22] TU Delft. Step 1: Start your Master Thesis Project. URL: https://www.tudelft.nl/studenten/ faculteiten/tbm-studentenportal/onderwijs/master/graduation-portal/step-1start-your-master-thesis-project.
- [23] Bharat H Desai. "14. United Nations Environment Program (UNEP)". In: Yearbook of International Environmental Law 28 (2017), pp. 498–505.
- [24] Rodolfo Dufo-Lopez, José L Bernal-Agustín, and Javier Contreras. "Optimization of control strategies for stand-alone renewable energy systems with hydrogen storage". In: *Renewable* energy 32.7 (2007), pp. 1102–1126.
- [25] EIA. "US residential electricity prices increase as summer cooling demand rises". In: U.S. Energy Information Administration (Aug. 2021). URL: https://www.eia.gov/todayinenergy/detail. php?id=48296.
- [26] EIA. What are the greenhouse gas and air pollutant emissions factors for fuels and electricity? U.S. Energy Information Administration. 2021. URL: https://www.eia.gov/tools/faqs/faq. php?id=667&t=8.
- [27] Omar Ellabban, Haitham Abu-Rub, and Frede Blaabjerg. "Renewable energy resources: Current status, future prospects and their enabling technology". In: *Renewable and sustainable energy reviews* 39 (2014), pp. 748–764.
- [28] Harikishan R Ellamla et al. "Current status of fuel cell based combined heat and power systems for residential sector". In: *Journal of Power Sources* 293 (2015), pp. 312–328.
- [29] U.S. Department of Energy. Cost of Electrolytic Hydrogen Production. https://www.hydrogen. energy.gov/pdfs/20004-cost-electrolytic-hydrogen-production.pdf. Accessed on May 24, 2023. 2004.
- [30] Energy Department of America. Science : DOE Hydrogen Program. URL: https://www.hydro gen.energy.gov/science.html.
- [31] ENTSOE. Mission Statement. URL: https://www.entsoe.eu/about/inside-entsoe/object ives/.
- [32] ENTSOE. Monthly hourly load values 2006 2015. 2017. URL: https://www.entsoe.eu/ data/power-stats/.
- [33] EPA. Greenhouse Gases Equivalencies Calculator Calculations and References. U.S. Environmental Protection Agency. 2022. URL: https://www.epa.gov/energy/greenhouse-gasesequivalencies-calculator-calculations-and-references.
- [34] Inc Epsilon Associates. Project description and emissions. Tech. rep. URL: https://www.epa. gov/sites/default/files/2015-08/documents/beld-section2.pdf.
- [35] European Commission. JRC photovoltaic geographical information system (PVGIS). Jan. 2016. URL: https://re.jrc.ec.europa.eu/pvg\_tools/en/.
- [36] Ross Ferguson, William Wilkinson, and Robert Hill. "Electricity use and economic development". In: *Energy policy* 28.13 (2000), pp. 923–934.

- [37] Rafael Serra i Florensa and Rogelio Leal Cueva. "Architectural Integration of Solar Cells". In: *Practical Handbook of Photovoltaics*. Elsevier, 2012, pp. 917–941.
- [38] John Grin, Jan Rotmans, and Johan Schot. "On patterns and agency in transition dynamics: Some key insights from the KSI programme". In: *Environmental innovation and societal transitions* 1.1 (2011), pp. 76–81.
- [39] Giulio Guandalini and Stefano Campanari. "Wind Power Plant and Power-to-Gas System Coupled With Natural Gas Grid Infrastructure: Techno-Economic Optimization of Operation". In: June 2015. DOI: 10.1115/GT2015-42229.
- [40] Steven Hegedus and Antonio Luque. *Handbook of photovoltaic science and engineering*. John Wiley & Sons, 2011.
- [41] HyStock. Storage in salt caverns. URL: https://www.hystock.nl/en/hydrogen/storage-in-salt-caverns.
- [42] IEA. Global reported corporate energy R&D spending in selected sectors, 2010-2019. Licence: CC BY 4.0; Accessed: April 29, 2023. IEA. 2021. URL: https://www.iea.org/data-and-sta tistics/charts/global-reported-corporate-energy-r-and-d-spending-in-selectedsectors-2010-2019.
- [43] International Council on Clean Transportation (ICCT). Assessment of Hydrogen Production Costs. https://theicct.org/wp-content/uploads/2021/06/final\_icct2020\_assessment\_ of-\_hydrogen\_production\_costs-v2.pdf. Accessed on May 21, 2023.
- [44] International Energy Agency. Empowering Electricity Consumers to Lower Their Carbon Footprint. IEA Commentaries. 2023. URL: https://www.iea.org/commentaries/empoweringelectricity-consumers-to-lower-their-carbon-footprint.
- [45] International Energy Agency. Energy security. URL: https://www.iea.org/topics/energysecurity (visited on 12/11/2022).
- [46] International Energy Agency. Global Average Levelised Cost of Hydrogen Production by Energy Source and Technology (2019 and 2050). Website. 2019. URL: https://www.iea.org/dataand-statistics/charts/global-average-levelised-cost-of-hydrogen-production-byenergy-source-and-technology-2019-and-2050.
- [47] IRENA. Green Hydrogen Cost Reduction. Tech. rep. International Renewable Energy Agency, 2020. URL: https://www.irena.org/publications/2020/Sep/Green-hydrogen-costreduction.
- [48] Ehsanul Kabir et al. "Solar energy: Potential and future prospects". In: Renewable and Sustainable Energy Reviews 82 (2018), pp. 894–900.
- [49] James Kennedy and Russell Eberhart. "Particle swarm optimization". In: Proceedings of ICNN'95international conference on neural networks. Vol. 4. IEEE. 1995, pp. 1942–1948.
- [50] Nipon Ketjoy, Wisut Chamsa-ard, and Pornthip Mensin. "Analysis of factors affecting efficiency of inverters: Case study grid-connected PV systems in lower northern region of Thailand". In: *Energy Reports* 7 (2021), pp. 3857–3868.
- [51] M Jahangir Khan and Mohammad Tariq Iqbal. "Pre-feasibility study of stand-alone hybrid energy systems for applications in Newfoundland". In: *Renewable energy* 30.6 (2005), pp. 835–854.
- [52] Khalid Al-Khori, Yusuf Bicer, and Muammer Koç. "Comparative techno-economic assessment of integrated PV-SOFC and PV-Battery hybrid system for natural gas processing plants". In: *Energy* 222 (2021), p. 119923.
- [53] Seoyong Kim, Jae Eun Lee, and Donggeun Kim. "Searching for the next new energy in energy transition: Comparing the impacts of economic incentives on local acceptance of fossil fuels, renewable, and nuclear energies". In: *Sustainability* 11.7 (2019), p. 2037.
- [54] Mohan Kolhe et al. "Analytical model for predicting the performance of photovoltaic array coupled with a wind turbine in a stand-alone renewable energy system based on hydrogen". In: *Renewable energy* 28.5 (2003), pp. 727–742.
- [55] S Shiva Kumar and Hankwon Lim. "An overview of water electrolysis technologies for green hydrogen production". In: *Energy Reports* 8 (2022), pp. 13793–13813.

- [56] Averill M Law. Simulation modeling and analysis. Vol. 3. Mcgraw-hill New York, 2007.
- [57] Nouhaila Lazaar et al. "A genetic algorithm based optimal sizing strategy for PV/battery/hydrogen hybrid system". In: Artificial Intelligence and Industrial Applications: Artificial Intelligence Techniques for Cyber-Physical, Digital Twin Systems and Engineering Applications. Springer. 2021, pp. 247–259.
- [58] Alain Le Duigou et al. "Relevance and costs of large scale underground hydrogen storage in France". In: *International Journal of Hydrogen Energy* 42.36 (2017), pp. 22987–23003.
- [59] RL LeRoy. "Industrial water electrolysis: present and future". In: International Journal of Hydrogen Energy 8.6 (1983), pp. 401–417.
- [60] Juan Lu, Minghai Li, and Qinyang Li. "Modeling and optimal design of a grid-independent solutions based on solar-hydrogen storage feeding green building by optimization algorithm". In: *Journal of Energy Storage* 62 (2023), p. 106844.
- [61] Fang Lin Luo and Hong Ye. Advanced DC/AC inverters: applications in renewable energy. Crc Press, 2017.
- [62] Konrad Mertens. *Photovoltaics: fundamentals, technology, and practice*. John Wiley & Sons, 2018.
- [63] Bin Miao and Siew Hwa Chan. "The economic feasibility study of a 100-MW Power-to-Gas plant". In: *International Journal of Hydrogen Energy* 44.38 (2019), pp. 20978–20986.
- [64] Amaresh Mishra and Peter Bäuerle. "Small molecule organic semiconductors on the move: promises for future solar energy technology". In: Angewandte Chemie International Edition 51.9 (2012), pp. 2020–2067.
- [65] Mohammad Jafar Hadidian Moghaddam et al. "Optimal sizing and energy management of stand-alone hybrid photovoltaic/wind system based on hydrogen storage considering LOEE and LOLE reliability indices using flower pollination algorithm". In: *Renewable energy* 135 (2019), pp. 1412–1434.
- [66] Eric R Morgan, James F Manwell, and Jon G McGowan. "Opportunities for economies of scale with alkaline electrolyzers". In: *international journal of hydrogen energy* 38.36 (2013), pp. 15903– 15909.
- [67] National Energy Education Development Project (NEED). *Hydrogen fuel cell*. Nov. 2021. URL: https://ie.unc.edu/clean-tech-post/hello-green-hydrogen/.
- [68] NOS. Nederland Europees koploper zonnepanelen, maar hoe kan het beter? https://nos. nl/artikel/2439211-nederland-europees-koploper-zonnepanelen-maar-hoe-kan-hetbeter. [Accessed on: April 30, 2023]. 2021.
- [69] NOS. "Netbeheerder: stroomnet te vol, stop met gunstige regeling zonnepanelen". In: (May 2021). URL: https://nos.nl/artikel/2413476-netbeheerder-stroomnet-te-vol-stop-met-gunstige-regeling-zonnepanelen.
- [70] NSO et al. Zonneparken in Nederland. Jan. 2022.
- [71] Ryan O'Hayre. "Fuel cells for electrochemical energy conversion". In: *EPJ Web of Conferences* 148 (Jan. 2017), p. 00013. DOI: 10.1051/epjconf/201714800013.
- [72] Office of energy efficiency renewable energy. Crystalline Silicon Photovoltaics Research. URL: https://www.energy.gov/eere/solar/crystalline-silicon-photovoltaics-research.
- [73] Dong-Hoon Oh, Chang-Ha Lee, and Jae-Cheol Lee. "Performance and cost analysis of natural gas combined cycle plants with chemical looping combustion". In: ACS omega 6.32 (2021), pp. 21043–21058.
- [74] MS Okundamiya. "Size optimization of a hybrid photovoltaic/fuel cell grid connected power system including hydrogen storage". In: *International Journal of Hydrogen Energy* 46.59 (2021), pp. 30539–30546.
- [75] AG Olabi et al. "Large-vscale hydrogen production and storage technologies: Current status and future directions". In: *International Journal of Hydrogen Energy* 46.45 (2021), pp. 23498– 23528.

- [76] Rijksdienst voor Ondernemend Nederland. *Netcapaciteit: Zonne-energie*. https://www.rvo. nl/onderwerpen/zonne-energie/netcapaciteit. Accessed 2023.
- [77] Amani Al-Othman et al. "Artificial intelligence and numerical models in hybrid renewable energy systems with fuel cells: Advances and prospects". In: *Energy Conversion and Management* 253 (2022), p. 115154.
- [78] Mark Palmer. "Special Interest Group: When geoscientists meet social scientists in grand challenges: Net zero, energy transition policy and the economy". In: *18th Corporate Responsibility Research Conference*. 2022.
- [79] Paris Agreement. UN Treaty. United Nations, Dec. 2015. URL: https://treaties.un.org/ pages/ViewDetails.aspx?src=TREATY&mtdsg\_no=XXVII-7-d&chapter=27&clang=\_en (visited on 03/28/2019).
- [80] Planbureau voor de Leefomgeving (PBL). Het potentieel van zonnestroom in de gebouwde omgeving van Nederland. https://www.pbl.nl/sites/default/files/downloads/pbl-2014-dnv-gl-het-potentieel-van-zonnestroom-in-de-gebouwde-omgeving-van-nederla nd\_01400.pdf. [Accessed on: April 30, 2023]. 2014.
- [81] Power Outages. [Online; accessed 14-May-2023]. URL: https://www.ready.gov/poweroutages#:~:text=A%20power%20outage%20may%3A,food%20spoilage%20and%20water% 20contamination..
- [82] A Rajendra Prasad and E Natarajan. "Optimization of integrated photovoltaic–wind power generation systems with battery storage". In: *Energy* 31.12 (2006), pp. 1943–1954.
- [83] Robert Rapier. "Wind And Solar Provided A Record 10% Of The World's Power Generation In 2021". In: (July 4, 2022). URL: https://www.forbes.com/sites/rrapier/2022/07/04/windand-solar-provided-a-record-10-of-the-worlds-power-in-2021/?sh=3f6b72bd14aa (visited on 11/08/2022).
- [84] Wetenschappelijke Raad voor het Regeringsbeleid (WRR). Infrastructures: Time to Invest. Amsterdam University Press, 2008. URL: https://www.wrr.nl/publicaties/rapporten/2008/ 06/11/infrastructures-time-to-invest.
- [85] Anita H Reksten et al. "Projecting the future cost of PEM and alkaline water electrolysers; a CAPEX model including electrolyser plant size and technology development". In: International Journal of Hydrogen Energy 47.90 (2022), pp. 38106–38113.
- [86] Hongbo Ren et al. "Multi-objective optimization for the operation of distributed energy systems considering economic and environmental aspects". In: *Applied Energy* 87.12 (2010), pp. 3642– 3651.
- [87] N Rispoli et al. "Constrained optimal design of a reversible solid oxide cell-based multiple load renewable microgrid". In: *Journal of Energy Storage* 31 (2020), p. 101570.
- [88] S Safari, MM Ardehali, and MJ Sirizi. "Particle swarm optimization based fuzzy logic controller for autonomous green power energy system with hydrogen storage". In: *Energy conversion and management* 65 (2013), pp. 41–49.
- [89] Johns Hopkins University SAIS. "Renewable Energy vs. Sustainable Energy". In: The Energy Journal (2019). URL: https://energy.sais-jhu.edu/articles/renewable-energy-vssustainable-energy/.
- [90] Albert H Schrotenboer et al. "A Green Hydrogen Energy System: Optimal control strategies for integrated hydrogen storage and power generation with wind energy". In: *Renewable and Sustainable Energy Reviews* 168 (2022), p. 112744.
- [91] National Academy of Sciences. *Natural Gas.* National Academy of Sciences. 2021. URL: http: //needtoknow.nas.edu/energy/energy-sources/fossil-fuels/natural-gas/.
- [92] Tareq M. Shami et al. "Particle Swarm Optimization: A Comprehensive Survey". In: *IEEE Access* 10 (2022), pp. 10031–10061. DOI: 10.1109/ACCESS.2022.3142859.
- [93] Masoud Sharafi and Tarek Y ELMekkawy. "Multi-objective optimal design of hybrid renewable energy systems using PSO-simulation based approach". In: *Renewable energy* 68 (2014), pp. 67– 79.

- [94] Shell. Sectoral Decarbonisation Approach. https://reports.shell.com/energy-transi tion-progress-report/2021/\_assets/downloads/sectoral-decarbonisation-shelletr21.pdf. Accessed: April 29, 2023. 2021.
- [95] Shell and Wuppertal Institut. Energy of the future? Hamburg, de, 2017. URL: https://s06. static-shell.com/content/dam/royaldutchshell/documents/shell-h2-study-new.pdf (visited on 11/04/2022).
- [96] Yuhui Shi and Russell Eberhart. "A modified particle swarm optimizer". In: 1998 IEEE international conference on evolutionary computation proceedings. IEEE world congress on computational intelligence (Cat. No. 98TH8360). IEEE. 1998, pp. 69–73.
- [97] Mandeep Singh, Dario Zappa, and Elisabetta Comini. "Solid oxide fuel cell: Decade of progress, future perspectives and challenges". In: *International Journal of Hydrogen Energy* 46.54 (2021), pp. 27643–27674.
- [98] Yujia Song et al. "Multi-objective optimization of large-scale grid-connected photovoltaic-hydrogennatural gas integrated energy power station based on carbon emission priority". In: International Journal of Hydrogen Energy 48.10 (2023), pp. 4087–4103.
- [99] Joanneke Spruijt. *Wat levert een Zonneweide per ha op?* https://edepot.wur.nl/521731. Accessed: Apr. 30, 2023. ACRRES - Wageningen UR, 2015.
- [100] Albert Sweetser. "A comparison of system dynamics (SD) and discrete event simulation (DES)". In: 17th International Conference of the System Dynamics Society. 1999, pp. 20–23.
- [101] Axel Thevenot. Particle Swarm Optimization. Dec. 202. URL: https://www.google.com/ url?sa=i&url=https%5C%3A%5C%2F%5C%2Ftowardsdatascience.com%5C%2Fparticleswarm-optimization-visually-explained-46289eeb2e14&psig=AOvVawOom4NN8VbvfdAND da6p2Fb&ust=1680029440252000&source=images&cd=vfe&ved=OCBAQjRxqFwoTCOjF5cHj\_ POCFQAAAAAdAAAABAj.
- [102] TU Delft. *Processen in de energiesector*. Nov. 2022. URL: https://eduweb.eeni.tbm.tudelft. nl/TB242E/?elektriciteitscentrales-soorten.
- [103] TU Delft. Step 1: Start your Master Thesis Project. URL: https://www.tudelft.nl/studenten/ faculteiten/tbm-studentenportal/onderwijs/master/graduation-portal/step-1start-your-master-thesis-project.
- [104] John Twidell. *Renewable energy resources*. Routledge, 2021.
- [105] VV Tyagi et al. "Progress in solar PV technology: Research and achievement". In: *Renewable and sustainable energy reviews* 20 (2013), pp. 443–461.
- [106] U.S. Department of Energy. End-of-Life Management for Solar Photovoltaics. https://www. energy.gov/eere/solar/end-life-management-solar-photovoltaics. Accessed: Apr. 30, 2023. 2021.
- [107] U.S. Department of energy. *Electrolysis*. URL: https://www.energy.gov/eere/fuelcells/ hydrogen-production-electrolysis.
- [108] U.S. Department of energy. "Fuel cell technologies office". In: (Nov. 2015). URL: https://www. energy.gov/eere/fuelcells/articles/fuel-cells-fact-sheet.
- [109] Raza Umar. "Hybrid Cooperative Energy Detection Techniques in Cognitive Radio Networks". In: Jan. 2014.
- [110] United Nations. The Sustainable Development Goals Report 2022. 2022. URL: https://unst ats.un.org/sdgs/report/2022/ (visited on 11/17/2022).
- [111] SF Van Gessel et al. "Underground hydrogen storage in salt caverns in the Netherlands–Storage performance and implications for geomechanical stability". In: *The Mechanical Behavior of Salt X*. CRC Press, 2022, pp. 607–615.
- [112] Gert PJ Verbong and Frank W Geels. "Exploring sustainability transitions in the electricity sector with socio-technical pathways". In: *Technological Forecasting and Social Change* 77.8 (2010), pp. 1214–1221.

- [113] SR Vosen and JO Keller. "Hybrid energy storage systems for stand-alone electric power systems: optimization of system performance and cost through control strategies". In: *International journal of hydrogen energy* 24.12 (1999), pp. 1139–1156.
- [114] J Wahono et al. "Green prosperity: a natural-based solution for rural electrification in Indonesia". In: *IOP Conference Series: Earth and Environmental Science*. Vol. 749. 1. IOP Publishing. 2021, p. 012023.
- [115] Richard L Welch and Ganesh K Venayagamoorthy. "Comparison of two optimal control strategies for a grid independent photovoltaic system". In: *Conference Record of the 2006 IEEE Industry Applications Conference Forty-First IAS Annual Meeting*. Vol. 3. IEEE. 2006, pp. 1120– 1127.
- [116] Cameron Wells et al. "Strategies for the Adoption of Hydrogen-Based Energy Storage Systems: An Exploratory Study in Australia". In: *Energies* 15.16 (2022), p. 6015.
- [117] Fei Wu et al. "A comprehensive evaluation of wind-PV-salt cavern-hydrogen energy storage and utilization system: A case study in Qianjiang salt cavern, China". In: *Energy Conversion and Management* 277 (2023), p. 116633.
- [118] Chuanbo Xu et al. "Data-driven configuration optimization of an off-grid wind/PV/hydrogen system based on modified NSGA-II and CRITIC-TOPSIS". In: *Energy Conversion and Management* 215 (2020), p. 112892.
- [119] Fardila Mohd Zaihidee et al. "Dust as an unalterable deteriorative factor affecting PV panel's efficiency: Why and how". In: *Renewable and Sustainable Energy Reviews* 65 (2016), pp. 1267– 1278.
- [120] Kai Zeng and Dongke Zhang. "Recent progress in alkaline water electrolysis for hydrogen production and applications". In: *Progress in energy and combustion science* 36.3 (2010), pp. 307– 326.
- [121] Weiping Zhang et al. "An artificial intelligence approach to optimization of an off-grid hybrid wind/hydrogen system". In: *International Journal of Hydrogen Energy* 46.24 (2021), pp. 12725– 12738.



# Dataset transformation coding

Coding for the transforming the dataset on solar irradiance

```
1 from google.colab import drive
2 import pandas as pd
3 import numpy as np
4 drive.mount('/content/gdrive')
6 #This line of code imports the dataset and skips the first 10 rows since this contains other
      information than the hourly data
7 df_pv=pd.read_csv('gdrive/My Drive/Thesis Stijn Knoop/Datasets/Timeseries_52.092_5.104
      _SA2_1kWp_crystSi_14_39deg_-1deg_2005_2020.csv', skiprows=10)
9 #This line of code truncates the last values of the table because these cells contain
      explainations on the column names and are not intended for the dataset
10 df_pv = df_pv[:-7]
11
12 #This line of code visualizes the first 20 rows of the dataset
13 df_pv.head(20)
14
15 #This line of code takes the 'Power column and creates a dataset of only the values of this
     column
16 df_time = df_pv[['time']]
17
18 #These line of codes split the column 'time' into multiple columns:Year, Month, Day, Hour.
      Also it removes the initial column time in the second line of code
df_time['time'].str[-2:]
20 df_time = df_time[['Year', 'Month', 'Day', 'Hour']]
21
22 #This line of code takes the 'Power column and creates a dataset of only the values of this
     column
23 df_P = df_pv[['P']]
24
25 #This line of code merges the two datasets (the 'time dataset' and the 'power dataset')
26 df_pv = pd.concat([df_time, df_P], axis=1)
27
28 #This line of code visualizes the first 20 rows of the dataset
29 df_pv.head(20)
30
31 #This line of code creates a csv-file and stores the data in the dataframe above in the
      folder 'Modified datasets' which can be found in the 'Coding' folder
32 with open('gdrive/My Drive/Thesis Stijn Knoop/Coding/Modified datasets/
      Electricity_solar_dataset.csv', 'w', encoding = 'utf-8-sig') as f:
   df_pv.to_csv(f)
33
```

Coding for the transforming the dataset on electricity demand

```
1 #This line of code imports the dataset and skips the first 3 rows since this contains other information than the hourly data
```

```
2 df_demand = pd.read_excel('gdrive/My Drive/Thesis Stijn Knoop/Datasets/Monthly-hourly-load-
      values_2006-2015.xlsx', skiprows=3)
4 #This line of code filters the Dataset on the data regarding the Netherlands. All the data
      about other countries are dropped
5 df_demand = df_demand.loc[df_demand['Country'] == 'NL']
7 #This line of code resets the index of the table so it starts from 0 again
8 df_demand = df_demand.reset_index(drop = True)
10 #This line of code visualizes the first 20 rows of the dataset
11 df_demand.head(20)
12
13 #In this line of code a dataframe with empty columns is created. This dataframe will be
      filled with the data from the dataset 'df_demand' which is created in previous cell
14 df_demand_2 = pd.DataFrame(columns = ['Year', 'Month', 'Day', 'Hour', 'Demand'])
15
16 #This line is setting the variable hours_in_a_day to 24. The code below will therefore run
      from 0 to 23 which is needed to read the values in the columns of 'df_demand' and put
      them in df_demand_2
17 hours_in_a_day = 24
18
19 #This line of code is creating an empty list. This is needed before running the for-loop.
      This list will be filled with the corresponding data and will be added to 'df_demand_2'
20 list_to_append_to_df_demand_2 = []
21
22 #This for-loop will go through df_demand row-for-row. 'i' will be incremented for every row
      it reads
23 for i in range(len(df_demand)):
24
25
    #This line of code is keeping track of the progress how many rows have been gone through
        and prints every 100 rows its status. The person running this cell is shown the
        progress
    if i/100 == i//100:
26
      print(i, ' of ',len(df_demand), ' lines have been run')
27
28
    #this for loop will produce a value for 'j' from 0 up to 23. This is needed to read the
29
        columns 0,1,3,..., 23 of the dataset df_demand
    for j in range(hours_in_a_day):
30
31
      #In the following lines of codes the Year, Month, Day, Hour and Demand are determined.
32
      Year = df_demand['Year'][i]
33
      Month = df_demand['Month'][i]
34
35
      Day = df_demand['Day'][i]
36
      #This if-else statement is needed to produce the correct syntax for Hour. This code makes
37
           sure the value of Hour will be (for example): '01:00' instead of '1:00'.
      if len(str(j)+':00')<5:</pre>
38
        Hour = '0' + str(j) + ':00'
39
40
      else:
        Hour = str(j)+':00'
41
42
      Demand = df_demand[j][i]
43
44
      #This line of code puts all the variables that have been created in the for-loop in a
45
          list
      list_to_append_to_df_demand_2 = [Year, Month, Day, Hour, Demand]
46
47
48
      #This line of code appends the list to the dataframe df_demand_2.
      df_demand_2.loc[len(df_demand_2)] = list_to_append_to_df_demand_2
49
50
51 #This line of code will let the person who runs this code know when this cell is done running
52 print('Finished')
53
54 #This line of code visualizes the first 20 rows of the dataset
55 df_demand_2.head(20)
56
57 #This line of code creates a csv-file and stores the data in the dataframe above in the
      folder 'Modified datasets' which can be found in the 'Coding' folder
58 with open('gdrive/My Drive/Thesis Stijn Knoop/Coding/Modified datasets/Demand_dataset.csv',
  w', encoding = 'utf-8-sig') as f:
```

59 df\_demand\_2.to\_csv(f)

# В

# Model coding

Adding source code to your report/thesis is supported with the package listings. An example can be found below. Files can be added using \lstinputlisting[language=<language>]{<filename>}.

```
1 def shortage_off():
   global hydrogen_storage
2
    global energy_match
3
   global i
4
   global solar
5
6
    global hydrogen
    global gas
7
8
    global start_up_ticker
    global A
9
    #global B
10
    global C
11
    global D
12
    global E
13
14
    global F
15
    global switch_on_sources
16
17
    energy_match = abs(energy_match)
18
   # if hydrogen_storage *100/max_hydrogen_storage < A or hydrogen_storage == 0: # below</pre>
19
       threshold A so keep fuel cell off and use gas turbine
    if hydrogen_storage == 0 :
20
21
      gas = energy_match
22
      energy_match = 0
      sources['gas_turbine_active'] = 'true'
23
24
    elif hydrogen_storage*100/max_hydrogen_storage >= A:
                                                             #above threshold A so start fuel
25
        cell
      if hydrogen_storage*100/max_hydrogen_storage < D:</pre>
                                                             # treshold D to determine how much
26
          Watt produced by fuel cells
        Watt_produced_fuel_cell = C
27
      else:
28
        Watt_produced_fuel_cell = E
29
30
      sources['fuel_cell_active'] = 'true'
31
      start_up_ticker = start_up_hours_fuel_cells #
32
      switch_on_sources['fuel_cell'] = switch_on_sources['fuel_cell'] +1
33
      #print(start_up_ticker)
34
      Watt_in_storage = (hydrogen_storage * (kWh_per_kg_hydrogen*1000))*(
35
          efficiency_hydrogen_tank/100)*(inverter_hydrogen_efficiency/100) #efficiencies of
          fuel cell/storage/inverter is already calculated here
36
      Max_watt_fuel_cell = (number_of_fuel_cells * Watt_produced_fuel_cell)
37
      if start_up_ticker > 0:
38
        start_up_ticker = start_up_ticker -1
39
        sources['gas_turbine_active'] = 'true'
40
        sources['fuel_cell_active'] = 'true'
41
42
      gas = energy_match
```

```
43
        energy_match = 0
44
45
      elif start_up_ticker <=0: #this does not happen (if start up time of fuel cells is more</pre>
46
           than 0 hour) because ticker is just activated. However, rules made when experimenting
            with quick start up time
        if Max_watt_fuel_cell <= Watt_in_storage:</pre>
47
          if Max_watt_fuel_cell >= energy_match:
48
49
            hydrogen = energy_match
            hydrogen_storage = (Watt_in_storage -energy_match)/(kWh_per_kg_hydrogen*1000)/(
50
                 efficiency_hydrogen_tank/100)/(inverter_hydrogen_efficiency/100)
51
             energy_match = 0
             sources['gas_turbine_active'] = 'false'
52
             sources['fuel_cell_active'] = 'true'
53
54
          elif Max_watt_fuel_cell < energy_match:</pre>
55
            hydrogen = Max_watt_fuel_cell
56
57
             energy_match = energy_match - Max_watt_fuel_cell
58
             gas = energy_match
59
             energy_match = 0
            hydrogen_storage = (Watt_in_storage - Max_watt_fuel_cell)/(kWh_per_kg_hydrogen
60
                 *1000)/(efficiency_hydrogen_tank/100)/(inverter_hydrogen_efficiency/100)
             sources['gas_turbine_active'] = 'true'
61
             sources['fuel_cell_active'] = 'false'
62
63
64
65
        elif Max_watt_fuel_cell > Watt_in_storage:
          if Watt_in_storage >= energy_match:
66
            hydrogen = energy_match
67
            hydrogen_storage = (Watt_in_storage - energy_match)/(kWh_per_kg_hydrogen*1000)/(
68
                 efficiency_hydrogen_tank/100)/(inverter_hydrogen_efficiency/100)
             energy match = 0
69
70
             sources['gas_turbine_active'] = 'false'
71
           elif Watt_in_storage < energy_match:</pre>
72
            hydrogen = Watt_in_storage
73
             energy_match = energy_match - Watt_in_storage
74
75
             gas = energy_match
             energy_match = 0
76
             hydrogen_storage = 0
77
             sources['gas_turbine_active'] = 'true'
78
            sources['fuel_cell_active'] = 'false'
79
1 def surplus_off(): #solar panels generate enough electricity to fulfill the demand --> fuel
       cells are turned off and stay turned off. if gasturbine is turned on --> shut down
2
    global hydrogen_storage
    global energy_match
3
    global i
4
    global solar
5
    global hydrogen
6
7
    global gas
    global q
8
    #global B
9
    global w
10
    global e
11
    global r
12
13
    global t
    global electrolysis_list
14
15
    sources['gas_turbine_active'] = 'false'
16
    sources['fuel_cell_active'] = 'false'
17
18
19
    if energy_match > number_of_electrolysers* max_W_electrolyser:
      energy_match = number_of_electrolysers* max_W_electrolyser #only consider the
20
           electricity that can be handled by the electrolysers. The electricity which cant be
           handled is wasted
21
    kg_produced = energy_match * 1/(kWh_needed_for_1_kg_hydrogen*1000)
22
    hydrogen_storage = hydrogen_storage + kg_produced
23
24
25 if hydrogen_storage > max_hydrogen_storage:
```

```
electrolysis_list.append(electrolysis_list[-1]+hydrogen_storage - max_hydrogen_storage)
26
      hydrogen_storage = max_hydrogen_storage
27
28
    else:
      electrolysis_list.append(electrolysis_list[-1]+kg_produced)
29
30
31
    solar = (demand_dataset['Demand'][i])*1000000
32
33
34
    #if hydrogen_storage > max_hydrogen_storage:
35 # hydrogen_storage = max_hydrogen_storage
1 def surplus_on(): #keep fuel cells running, even though solar panels generate enough to
      fulfill the electricity demand. if gasturbine is turned on --> shut down
2
    global hydrogen_storage
    global energy_match
3
    global i
4
    global solar
5
    global hydrogen
6
7
    global gas
    global A
8
    #global B
9
    global C
10
    global D
11
    global E
12
13
    global F
    global electrolysis_list
14
15
    sources['gas_turbine_active'] = 'false'
16
    sources['fuel_cell_active'] = 'true'
17
18
    Watt_in_storage = (hydrogen_storage * (kWh_per_kg_hydrogen*1000))*(efficiency_hydrogen_tank
19
         /100)*(inverter_hydrogen_efficiency/100) #efficiencies of fuel cell/storage/inverter is
         already calculated here
    Max_watt_fuel_cell = (number_of_fuel_cells * F)
20
21
    if Watt_in_storage >= Max_watt_fuel_cell: #if enough in hydrogen storage
22
      hydrogen_storage = (Watt_in_storage - Max_watt_fuel_cell)/(kWh_per_kg_hydrogen*1000)/(
23
          efficiency_hydrogen_tank/100)/(inverter_hydrogen_efficiency/100)
      energy_match = solar_dataset['P'][i]*ha_solar_park*10000 + Max_watt_fuel_cell - (
24
          demand_dataset['Demand'][i])*1000000
                                                    # 10000 meter in 1 ha. # 1000000 to
          translate MW to W
25
26
      if energy_match > number_of_electrolysers* max_W_electrolyser:
27
        energy_match = number_of_electrolysers* max_W_electrolyser #only consider the
28
             electricity that can be handled by the electrolysers. The electricity which cant be
             handled is wasted
29
      kg_produced = energy_match * 1/(kWh_needed_for_1_kg_hydrogen*1000)
30
      hydrogen_storage = hydrogen_storage + kg_produced
31
32
      electrolysis_list.append(electrolysis_list[-1]+kg_produced)
33
34
      hydrogen = Max_watt_fuel_cell
35
      solar = (demand_dataset['Demand'][i])*1000000 - hydrogen
36
37
38
    elif Watt_in_storage < Max_watt_fuel_cell: #if not enough in hydrogen storage</pre>
39
      hydrogen_storage = 0
40
      energy_match = solar_dataset['P'][i]*ha_solar_park*10000 + Watt_in_storage - (
41
          demand_dataset['Demand'][i])*1000000
                                                   # 10000 meter in 1 ha.
                                                                              # 1000000 to
          translate MW to W
42
43
      if energy_match > number_of_electrolysers* max_W_electrolyser:
44
        energy_match = number_of_electrolysers* max_W_electrolyser #only consider the
45
             electricity that can be handled by the electrolysers. The electricity which cant be
             handled is wasted
46
      kg_produced = energy_match * 1/(kWh_needed_for_1_kg_hydrogen*1000)
47
      hydrogen_storage = hydrogen_storage + kg_produced
48
```

```
electrolysis_list.append(electrolysis_list[-1]+kg_produced)
49
50
51
      hydrogen = Watt_in_storage
52
      solar = (demand_dataset['Demand'][i])*1000000 - hydrogen
53
54
      sources['gas_turbine_active'] = 'false'
55
      sources['fuel_cell_active'] = 'false'
56
57
58
    if hydrogen_storage > max_hydrogen_storage:
59
60
      hydrogen_storage = max_hydrogen_storage
      electrolysis_list.append(electrolysis_list[-2]+hydrogen_storage - max_hydrogen_storage)
61
1 def shortage_on():
    global hydrogen_storage
2
    global energy_match
3
    global i
4
5
    global solar
    global hydrogen
6
7
    global gas
8
    global start_up_ticker
    global A
9
    #global B
10
    global C
11
    global D
12
    global E
13
    global F
14
15
    energy_match = abs(energy_match)
16
17
18
    if hydrogen_storage*100/max_hydrogen_storage >= D:
                                                          #above threshold D so fuel cell
19
        maximum power
      Watt_produced_fuel_cell = C
20
    if hydrogen_storage*100/max_hydrogen_storage < D: # treshold D to determine how much</pre>
21
        Watt produced by fuel cells
22
        Watt_produced_fuel_cell = E
23
24
    #print(start_up_ticker)
    Watt_in_storage = (hydrogen_storage * (kWh_per_kg_hydrogen*1000))*(efficiency_hydrogen_tank
25
         /100)*(inverter_hydrogen_efficiency/100) #efficiencies of fuel cell/storage/inverter is
          already calculated here
    Max_watt_fuel_cell = (number_of_fuel_cells * Watt_produced_fuel_cell)
26
27
28
    if start_up_ticker > 0:
      start_up_ticker = start_up_ticker -1
29
30
      sources['gas_turbine_active'] = 'true'
     gas = energy_match
31
      energy_match = 0
32
33
    elif start_up_ticker <=0: #this does not happen (if start up time of fuel cells is more
34
        than 0 hour) because ticker is just activated. However, rules made when experimenting
         with quick start up time
      if Max_watt_fuel_cell <= Watt_in_storage:</pre>
35
36
        if Max_watt_fuel_cell >= energy_match:
37
          hydrogen = energy_match
          hydrogen_storage = (Watt_in_storage -energy_match)/(kWh_per_kg_hydrogen*1000)/(
38
               efficiency_hydrogen_tank/100)/(inverter_hydrogen_efficiency/100)
          energy_match = 0
39
          sources['gas_turbine_active'] = 'false'
40
41
42
        elif Max_watt_fuel_cell < energy_match:</pre>
          hydrogen = Max_watt_fuel_cell
43
          energy_match = energy_match - Max_watt_fuel_cell
44
45
          gas = energy_match
          energy_match = 0
46
          hydrogen_storage = (Watt_in_storage - Max_watt_fuel_cell)/(kWh_per_kg_hydrogen*1000)
47
               /(efficiency_hydrogen_tank/100)/(inverter_hydrogen_efficiency/100)
           sources['gas_turbine_active'] = 'true'
48
```

```
49 sources['fuel_cell_active'] = 'true'
```

```
50
51
      elif Max_watt_fuel_cell > Watt_in_storage:
52
53
        if Watt_in_storage >= energy_match:
          hydrogen = energy_match
54
          hydrogen_storage = (Watt_in_storage - energy_match)/(kWh_per_kg_hydrogen*1000)/(
55
              efficiency_hydrogen_tank/100)/(inverter_hydrogen_efficiency/100)
          energy_match = 0
56
          sources['gas_turbine_active'] = 'false'
57
58
        elif Watt_in_storage < energy_match:</pre>
59
60
          hydrogen = Watt_in_storage
          energy_match = energy_match - Watt_in_storage
61
          gas = energy_match
62
63
          energy_match = 0
          hydrogen_storage = 0
64
          sources['gas_turbine_active'] = 'true'
65
66
          sources['fuel_cell_active'] = 'false'
```

# $\bigcirc$

# Particle swarm optimization coding

Coding for initialising the particles.

```
1 def setting_particles(par):
    global global_best
2
3
    for n in range(n_particles):
4
      dictionary = {}
5
      for i in range(len(par)):
6
7
      while dictionary['position_par0'] < dictionary['position_par2']:</pre>
8
        dictionary['position_par0'] = random.random()*100
9
10
      model(dictionary['position_par0'], dictionary['position_par1'],dictionary['position_par2'
11
          ],dictionary['position_par3'],dictionary['position_par4'])
      dictionary['personal_best_score'] = z
12
13
      if z < global_best[-1]:</pre>
14
        global_best = [dictionary['position_par0'], dictionary['position_par1'],dictionary['
15
            position_par2'],dictionary['position_par3'],dictionary['position_par4'],z]
16
17
18
      globals()['particle_%s' % n] = dictionary
      list_of_particles.append(globals()['particle_%s' % n])
19
```

Code for initialising a new position of the particles

```
1 def new_position():
      global iterations
2
3
      global p
4
      global list_of_particles
5
      for y in(range(len(par))):
6
         inertia = (0.4*((p-iterations)/iterations**2)+ 0.4 ) * (list_of_particles[j]['
7
             position_par{}'.format(y)] -list_of_particles[j]['position_previous_par{}'.format(
             y)] )
8
        personal = 0*(list_of_particles[j]['position_best_par{}'.format(y)] - list_of_particles
9
            [j]['position_par{}'.format(y)] )* random.random()* ((-3*p/iterations) +3.5)
        globaal = (global_best[y] - list_of_particles[j]['position_par{}'.format(y)] )* random.
10
            random()*((3*p/iterations) +0.5)
11
        new = list_of_particles[j]['position_par{}'.format(y)] + round(inertia + personal +
12
            globaal,2)
        if new < par[y][0]:</pre>
13
          new = par[y][0]
14
        elif new > par[y][1]:
15
          new = par[y][1]
16
17
18
        list_of_particles[j]['position_previous_par{}'.format(y)] = list_of_particles[j]['
            position_par{}'.format(y)]
19
        list_of_particles[j]['position_par{}'.format(y)] = new
```

Optimizing the model with the particle swarm optimization

20

21

22

```
1 list_of_particles = []
2 global_best_list = []
3 global_best = [0,0,0,0,0,100**10000] # to make sure global best is very high starting the
       simulation
4 n_particles = 75
5 iterations = 10
8 if len(list_of_particles) == 0:
9
       setting_particles(par)
10
11 for p in tqdm(range(1, iterations +1), desc = 'Progress', leave = True):
12 for j in tqdm(range(n_particles), desc = ' Progress of iteration {} '.format(p), leave =
         True):
13
       model(list_of_particles[j]['position_par0'], list_of_particles[j]['position_par1'],
14
           list_of_particles[j]['position_par2'],list_of_particles[j]['position_par3'],
            list_of_particles[j]['position_par4'])
15
       if z < list_of_particles[j]['personal_best_score']:</pre>
16
         list_of_particles[j]['personal_best_score'] = z
17
         for m in range(len(par)):
18
           list_of_particles[j]['position_best_par{}'.format(m)] = list_of_particles[j]['
19
                position_par{}'.format(m)]
20
21
       if z < global_best[-1]:</pre>
         print(global_best , z)
22
         global_best[-1] = z
23
24
         global_best[0] = list_of_particles[j]['position_par0']
         global_best[1] = list_of_particles[j]['position_par1']
25
26
         global_best[2] = list_of_particles[j]['position_par2']
         global_best[3] = list_of_particles[j]['position_par3']
global_best[4] = list_of_particles[j]['position_par4']
27
28
29
30
       new_position()
     global_best_list.append(global_best[-1])
31
32 print('iteration:', p, 'global best: ',global_best)
```